# Request for Qualification

for

Rashtriya Adarsh Vidyalaya

through

Public Private Partnership

Ministry of Human Resource Development Government of India

# **TABLE OF CONTENTS**

Sl. No.		Contents	
			No.
	Gloss	ary	iii
	Discla	imer	V
1	Intro	luction	1
	1.1	Background	1
	1.2	Brief description of Bidding Process	2
	1.3	Schedule of Bidding Process	4
2	Instru	ictions to Applicants	6
	2A	General	6
	2.1	Scope of Application	6
	2.2	Eligibility of Applicants	6
	2.3	Number of Applications and costs thereof	11
	2.4	Site visit and verification of information	11
	2.5	Acknowledgement by Applicant	11
	2.6	Right to accept or reject any or all Applications/ Bids	12
	2B	Documents	13
	2.7	Contents of the RFQ	13
	2.8	Clarifications	13
	2.9	Amendment of RFQ	14
	<b>2</b> C	Preparation and Submission of Application	14
	2.10	Language	14
	2.11	Format and signing of Application	15
	2.12	Sealing and marking of Applications	15
	2.13	Application Due Date	16
	2.14	Late Applications	17
	2.15	Modifications/ substitution/withdrawal of Applications	17
	2D	Evaluation Process	17
	2.16	Opening and Evaluation of Applications	17
	2.17	Confidentiality	18
	2.18	Tests of responsiveness	18
	2.19	Clarifications	19
	<b>2</b> E	Qualification and Bidding	19
	2.20	Pre-qualification and notification	19

	2.21 Submission of Bids	20
	2.22 Proprietary data	20
	2.23 Correspondence with the Applicant	20
	2.24 Validity of Application	20
3	Criteria for Evaluation	21
	3.1 Evaluation of Applications	21
	3.2 Evaluation of Bidders	21
4	Fraud and Corrupt Practices	23
5	Pre-Application Conference	25
6	Miscellaneous	26
	Appendices	
Ι	Format for Letter Comprising the Application for Pre-	27
	Qualification	
	Annex – I Details of Applicant	30
	Annex – II Experience of the Applicant	32
	Annex – III Financial Capacity of the Applicant	33
	Annex – IV Details of Existing Schools	35
	Annex – V Details of Educational Institutions	38
	Annex – VI Preference for location of Schools	40
	Annex – VII Statement of Legal Capacity	41
II	Power of Attorney for signing of Application	42
III	Guidelines of the Department of Disinvestment	44
IV	Information Memorandum for 500 model schools through PPP	46
V	List of 3203 Blocks identified for Rashtriya Adarsh Vidyalayas	58

# GLOSSARY

Applicant	As defined in Clause 1.2.1
Application	As defined in the Disclaimer
Application Due Date	As defined in Clause 1.1.5
Associate	As defined in Clause 2.2.7
Authority	As defined in Clause 1.1.1
Bids	As defined in Clause 1.2.3
Bid Due Date	As defined in Clause 1.2.3
Bid Security	As defined in Clause 1.2.4
Bidders	As defined in Clause 1.1.2
<b>Bidding Documents</b>	As defined in Clause 1.2.3
Bidding Process	As defined in Clause 1.2.1
Bid Stage	As defined in Clause 1.2.2
Blocks	As defined in Clause 1.1.1
Concessionaire	As defined in Clause 1.1.3
<b>Concession Agreement</b>	As defined in Clause 1.1.3
<b>Conflict of Interest</b>	As defined in Clause 2.2.1(c)
DBFO	As defined in Clause 1.1.1
<b>Educational Institution</b>	As defined in Clause 2.2.2(c)
Government	Government of India
Highest Bidder	As defined in Clause 1.2.5
Information Memorandum	As defined in Clause 1.1.1
LOA	Letter of Award
Net Worth	As defined in Clause 2.2.3 (iii)
PPP	Public Private Partnership
Project	As defined in Clause 1.1.1
Qualification	As defined in Clause 1.2.1
Qualification Stage	As defined in Clause 1.2.1
Re. or Rs. or INR	Indian Rupee
<b>RFP or Request for Proposals</b>	As defined in Clause 1.2.2
RFQ	As defined in the Disclaimer
School	As defined in Clause 1.1.1

The words and expressions beginning with capital letters and defined in this document shall, unless repugnant to the context, have the meaning ascribed thereto herein.

#### DISCLAIMER

The information contained in this Request for Qualification document (the "**RFQ**") or subsequently provided to Applicant(s), whether verbally or in documentary or any other form, by or on behalf of the Authority or any of its employees or advisors, is provided to Applicant(s) on the terms and conditions set out in this RFQ and such other terms and conditions subject to which such information is provided.

This RFQ is not an agreement and is neither an offer nor invitation by the Authority to the prospective Applicants or any other person. The purpose of this RFO is to provide interested parties with information that may be useful to them in the formulation of their application for qualification pursuant to this RFQ (the "Application"). This RFQ includes statements, which reflect various assumptions and assessments arrived at by the Authority in relation to the Project. Such assumptions, assessments and statements do not purport to contain all the information that each Applicant may require. This RFQ may not be appropriate for all persons, and it is not possible for the Authority, its employees or advisors to consider the educational or investment objectives, financial situation and particular needs of each party who reads or uses this RFQ. The assumptions, assessments, statements and information contained in this RFQ may not be complete, accurate, adequate or correct. Each Applicant should therefore, conduct its own investigations and analysis and should check the accuracy, adequacy, correctness, reliability and completeness of the assumptions, assessments, statements and information contained in this RFQ and obtain independent advice from appropriate sources.

Information provided in this RFQ to the Applicant(s) is on a wide range of matters, some of which may depend upon interpretation of law. The information given is not intended to be an exhaustive account of statutory requirements and should not be regarded as a complete or authoritative statement of law. The Authority accepts no responsibility for the accuracy or otherwise for any interpretation or opinion on law expressed herein.

The Authority, its employees and advisors make no representation or warranty and shall have no liability to any person, including any Applicant or Bidder, under any law, statute, rules or regulations or tort, principles of restitution or unjust enrichment or otherwise for any loss, damages, cost or expense which may arise from or be incurred or suffered on account of anything contained in this RFQ or otherwise, including the accuracy, adequacy, correctness, completeness or reliability of the RFQ and any assessment, assumption, statement or information contained therein or deemed to form part of this RFQ or arising in any way with pre-qualification of Applicants for participation in the Bidding Process. The Authority also accepts no liability of any nature whether resulting from negligence or otherwise howsoever caused arising from reliance of any Applicant upon the statements contained in this RFQ.

The Authority may, in its absolute discretion but without being under any obligation to do so, update, amend or supplement the information, assessment or assumptions contained in this RFQ.

The issue of this RFQ does not imply that the Authority is bound to select and short-list pre-qualified Applications for Bid Stage or to appoint the selected Bidder or Concessionaire, as the case may be, for the Project and the Authority reserves the right to reject all or any of the Applications or Bids without assigning any reasons whatsoever.

The Applicant shall bear all its costs associated with or relating to the preparation and submission of its Application including but not limited to preparation, copying, postage, delivery fees, expenses associated with any demonstrations or presentations which may be required by the Authority or any other costs incurred in connection with or relating to its Application. All such costs and expenses will remain with the Applicant and the Authority shall not be liable in any manner whatsoever for the same or for any other costs or other expenses incurred by an Applicant in preparation or submission of the Application, regardless of the conduct or outcome of the Bidding Process.

#### Department of School Education & Literacy Ministry of Human Resource Development Government of India

#### **1. INTRODUCTION**

#### 1.1 Background

- 1.1.1 The Department of School Education & Literacy, Ministry of Human Resource Development, Government of India (the "Authority") is engaged provision of education, especially in the for the under privileged population, and as part of this endeavour, the Authority proposes to undertake development, operation and management of 2,500 Rashtriya Adarsh Vidyalaya through Public-Private Partnership (the "PPP") on Design, Build, Finance, and Operate (the "DBFO") basis. The school shall be located in (preferably block headquarters) villages/towns which are situated in any of the 3,203 non-educationally backward blocks listed in Appendix-V (the "Blocks"), but not more than one School shall be allocated for any Block. In the first phase of this scheme, it is proposed to award concessions for 500 (five hundred) higher secondary schools. In the pilot stage of this first phase, the Authority intends to award concessions for 50 schools to be located in any of the Blocks selected by it (the "Project") and concessions for the balance 450 schools shall be awarded subsequently. As a part of this Project, it has decided to carry out the process for selection of private entities to whom concessions for a school in each of the selected Blocks (the "School") may be awarded. The information memorandum for 500 model schools through PPP may be seen at Appendix-IV (the "Information Memorandum"). The locations for these 50 schools shall be chosen from the list of 3,203 locations in Appendix-V. The Applicants are required to indicate their preferred locations to enable the Authority to identify 50 locations based on Applicants' responses. {Note: The list of non-EBBs will be subject to amendment/change based on the Census 2011in second phase (when made available)}.
- 1.1.2 In the first stage, the Authority intends to pre-qualify suitable Applicants, who will be eligible for participation in the Bid Stage (the "**Bidders**"), for award of Schools through a fair and transparent selection process in accordance with the procedure set out herein.
- 1.1.3 The selected Bidder, who would be an entity that is either (i) a trust registered under the Indian Trusts Act, 1882 or the Bombay Public Trusts Act, 1950 (or other applicable laws) or (ii) a society registered under the Societies Registration Act, 1860 (or other applicable laws) or (iii) a notfor-profit company incorporated under Section 25 of the Companies Act, 1956, or (iv) an Associate that is , an entity specified hereinabove, (the "Concessionaire") shall be responsible for designing, financing, construction, operation and management of the respective Schools under and in accordance with the provisions of a long MHRD Education rfg/06.07.2012

term concession agreement (the "Concession Agreement") to be entered into between the Concessionaire and the Authority in the form provided by the Authority as part of the Bidding Documents pursuant hereto. The Concessionaire hereunder shall also include a trust, society or company referred to hereinabove which has been incorporated as such by a corporate entity that has been selected on the basis of its Net Worth.

- 1.1.4 The scope of work will broadly include procurement of land, construction, establishment, operation and management of a Higher Secondary School affiliated to the Central Board for Secondary Education (CBSE) in accordance with the provisions of the Concession Agreement. In exceptional circumstances, the Authority may allow affiliation with other National Boards.
- 1.1.5 The Authority shall receive Applications pursuant to this RFQ in accordance with the terms set forth herein as modified, altered, amended and clarified from time to time by the Authority, and all Applications shall be prepared and submitted in accordance with such terms on or before the date specified in Clause 1.3 for submission of Applications (the "Application Due Date").

## **1.2 Brief description of Bidding Process**

- 1.2.1. The Authority has adopted a two-stage process (collectively referred to as the "**Bidding Process**") for selection of the bidder for award of the Schools. The first stage (the "**Qualification Stage**") of the process involves qualification (the "**Qualification**") of interested parties who make an Application in accordance with the provisions of this RFQ (the "**Applicant**").
- 1.2.2. Prior to making an Application, the Applicant shall pay to the Authority a sum of Rs. 10,000 (Rupees ten thousand) as the cost of the RFQ process. At the end of this stage, the Authority expects to announce a list of suitable pre-qualified Applicants who shall be eligible for participation in the second stage of the Bidding Process to be conducted separately for each School (the "Bid Stage") comprising Request for Proposals (the "Request for Proposals" or "RFP"). In the Qualification Stage, Applicants would be required to furnish all the information specified in this RFQ. Only those Applicants that are pre-qualified by the Authority shall be invited to submit their Bids for the Project.
- 1.2.3. In the Bid Stage, the Bidders will be called upon to submit their offers (the "Bids") in accordance with the RFP and other documents to be provided by the Authority (collectively the "Bidding Documents"). The Bidding Documents for the Project will be provided to every Bidder on payment of Rs. 5,000 (Rs. five thousand only) per School applied for. The Bid shall be

valid for a period of not less than 120 days from the date specified in Clause 1.3 for submission of bids (the "**Bid Due Date**").

Government of India has issued guidelines (see Appendix-III) for qualification of bidders seeking to acquire stakes in any public sector enterprise through the process of disinvestment. These guidelines shall apply *mutatis mutandis* to this Bidding Process. The Authority shall be entitled to disqualify an Applicant in accordance with the aforesaid guidelines at any stage of the Bidding Process. Applicants must satisfy themselves that they are qualified to bid, and should give an undertaking to this effect in the form at Appendix-I.

- 1.2.4. In terms of the RFP, a Bidder will be required to deposit, along with its Bid, a bid security of Rs. 5 lakh (Rupees five lakh) (the "**Bid Security**"), refundable no later than 60 (sixty) days from the Bid Due Date, except in the case of the selected Bidder whose Bid Security shall be retained till it has provided a Performance Security under the Concession Agreement. The Bidders will have an option to provide Bid Security in the form of a demand draft or a bank guarantee acceptable to the Authority<sup>\$</sup> and in such event, the validity period of the demand draft or bank guarantee, as the case may be, shall not be less than 180 (one hundred and eighty) days from the Bid Due Date, inclusive of a claim period of 60 (sixty) days, and may be extended as may be mutually agreed between the Authority and the Bidder from time to time. The Bid shall be summarily rejected if it is not accompanied by the Bid Security.
- 1.2.5. In this RFQ, the term "**Highest Bidder**" shall mean the Bidder who secures the highest technical score for a School in the Bid Stage.
- 1.2.6. Generally, the Highest Bidder shall be the selected Bidder. The remaining Bidders for the School shall be kept in reserve and may, in accordance with the process specified in the RFP, be awarded the concession in case the Highest Bidder for such School withdraws or is not selected for any reason. However, in such an event the Authority may, in its discretion, annul the Bidding Process for the School.
- 1.2.7. During the Bid Stage, Bidders are invited to examine the Project in greater detail, and to carry out, at their cost, such studies as may be required for submitting their respective Bids for award of the concession, including construction and operation of the School.
- 1.2.8. As part of the Bidding Documents, the Authority will provide a draft of the Concession Agreement prepared by the Authority and other information pertaining/ relevant to the Project available with it.

<sup>&</sup>lt;sup>\$</sup> The format for the bank guarantee will be provided as part of the RFP document.

- 1.2.9. Bids will be invited for each School and will be awarded to the bidder with the highest technical score for the respective location. The concession period shall be pre-determined, and will be indicated in the draft Concession Agreement forming part of the Bidding Documents.
- 1.2.10. The details of the process to be followed at the Bid Stage and the terms thereof will be spelt out in the Bidding Documents.
- 1.2.11. Any queries or request for additional information concerning this RFQ shall be submitted in writing or by fax and e-mail to the officer designated in Clause 2.12.3 below. The envelopes/ communications shall clearly bear the following identification/ title:

"Queries/ Request for Additional Information: RFQ for Rashtriya Adarsh Vidyalayas".

## 1.3 Schedule of Bidding Process

The Authority shall endeavour to adhere to the following schedule:

	<b>Event Description</b>	Date
	Qualification Stage	
1.	Last date for receiving queries	August 16, 2012
2.	Pre-Application Conference	August 23, 2012
3.	Authority response to queries by	August 30, 2012
4.	Application Due Date	October 01, 2012
5.	Announcement of pre-qualified bidders	October 26, 2012
	<u>Bid Stage</u>	Estimated Date
1.	Sale of Bid Documents	November 07, 2012
2.	Last date for receiving queries	December 07, 2012
3.	Pre-Bid meeting – 1	December 14, 2012
4.	Authority response to queries by	December 21, 2012
5.	Pre-bid meeting -2	December 28, 2012
5.	Bid Due Date	January 15, 2013
6.	Opening of Bids	On Bid Due Date
7.	Letter of Award (LOA)	Within 30 days of Bid Due

Date

8.	Validity of Bids	120 days of Bid Due Date
----	------------------	--------------------------

9. Signing of Concession Agreement Within 30 days of award of LOA

# 2. INSTRUCTIONS TO APPLICANTS

## A. GENERAL

#### 2.1 Scope of Application

- 2.1.1 The Authority wishes to receive Applications for Qualification in order to pre-qualify eligible Applicants for the Bid Stage.
- 2.1.2 Pre-qualified Applicants may be subsequently invited to submit the Bids for setting up the Schools forming part of the Project.

#### 2.2 Eligibility of Applicants

- 2.2.1 For determining the eligibility of Applicants for their pre-qualification hereunder, each of the following shall apply:
  - (a) The Applicant for pre-qualification shall be a single entity.
  - (b) The Applicant should either be a (i) trust registered under the Indian Trusts Act, 1882 or the Bombay Public Trusts Act, 1950 (or other applicable laws); or (ii) a society registered under the Societies Registration Act, 1860 (or other applicable laws) or (iii) a not-forprofit company, incorporated under Section 25 of the Companies Act, 1956. Corporate entities are also eligible to apply on the basis of their Net Worth, provided they undertake to set up the School through an eligible trust, society or not-for-profit company.
  - (c) An Applicant shall not have a conflict of interest (the "**Conflict of Interest**") that affects the Bidding Process. Any Applicant found to have a Conflict of Interest shall be disqualified. An Applicant shall be deemed to have a Conflict of Interest affecting the Bidding Process, if:
    - (i) Any applicant is an Associate of any other Applicant;
    - (ii) such Applicant, or any Associate thereof receives or has received any direct or indirect subsidy, grant, concessional loan or subordinated debt from any other Applicant, or any Associate thereof or has provided any such subsidy, grant, concessional loan or subordinated debt to any other Applicant, or any Associate thereof; or
    - (iii) such Applicant has the same legal representative for purposes of this Application as any other Applicant; or

- (iv) such Applicant, or any Associate thereof has participated as a consultant to the Authority in the preparation of any documents, design or technical specifications of the Project.
- (d) An Applicant shall be liable for disqualification if any legal, financial or technical adviser of the Authority in relation to the Project is engaged by the Applicant, or any Associate thereof, as the case may be, in any manner for matters related to or incidental to the Project. For the avoidance of doubt, this disqualification shall not apply where such adviser was engaged by the Applicant or it's Associate in the past but its assignment expired or was terminated 6 (six) months prior to the date of issue of this RFQ. Nor will this disqualification apply where such adviser is engaged after a period of 3 (three) years from the date of award of the School.
- (e) The Applicant shall furnish the details of eligibility for the last 2 (two) financial years immediately preceding the Application Due Date.
- 2.2.2 An Applicant shall be eligible for pre-qualification in the following categories:

(a) An Applicant having experience of running a CBSE-affiliated school(s) where at least 2 (two) batches have passed out from Class X would be eligible for award of 3 (three) Schools against each such existing school;

(b) an Applicant who is running a CBSE-affiliated school(s) from where 2 (two) batches of Class X have not yet passed out, shall be eligible for award of one School against each such existing school;

(c) An Applicant would qualify for award of 3 (three) Schools as against each such existing educational institution provided it has:

- (i) experience of at least 5 (five) years in running a higher secondary school, affiliated with any Board of Secondary Education or Intermediate Education, or
- (ii) experience of at least 5 (five) years in running an Educational Institution for imparting education in academic, technical or vocational courses.

Provided that a selected Bidder who is eligible for award of Schools pursuant to this sub-clause (c) shall undertake to make an interest-bearing

deposit of Rs. 25 lakh for each School awarded, which would be released in three equal annual instalments after the School is commissioned;

For the purpose of this RFQ:

"Educational Institutions" shall mean institutions recognized/approved by a competent authority e.g., Central/State Government or other authorized agencies, e.g., Boards of Secondary Education, UGC, AICTE, MCI, ICAR, Directorate of Education or other duly empowered authority".

(d) any company (including a company incorporated under section 25 of the Companies Act ,1956) ,), trust or society having a Net-Worth of Rs. 25 crore or more during each of the preceding 2 (two) financial years and intending to set up a PPP school(s) through a not-for-profit entity specified in Clause 2.2.1(b) above would be eligible for allocation of one School for every Rs. 25 crore of Net-Worth, provided it shall undertake to make an interest-bearing deposit of Rs. 50 lakh for each School awarded to it, which would be released in three equal annual instalments after the School is commissioned in accordance with the Concession Agreement. In case more than 3 (three)Schools are allocated to a single entity under this clause 2.2.2(d), it shall make an interest-bearing deposit of Rs. 25 lakh for each additional school awarded to it.

- 2.2.3 The Applicant shall enclose with its application, to be submitted in the format at Appendix-I, complete with its Annexes, including the following:
  - Copy of the certificate of registration and other requisite documents under the Indian Companies Act, 1956, the Societies Registration Act, 1860 (or other applicable laws), Indian Trusts Act, 1882 or the Bombay Public Trusts Act, 1950 (or other applicable laws), as applicable, along with copies of the documents of incorporation, Trust Deed, Memorandum of Association or Articles of Association, as applicable;
  - (ii) Certificate(s) from the affiliating educational board, indicating the validity of the affiliation specified in paragraph 2.2.2 above;
  - (iii) certificate(s) from its auditors specifying the Net Worth of the Applicant, as at the close of 2 (two) preceding financial years, and also specifying that the methodology adopted for calculating such Net Worth conforms to the provisions of this Clause 2.2.3 (iii) in respect of the eligibility specified in paragraph 2.2.2 (d) above. For the purposes of this RFQ, Net Worth (the "Net Worth") in the case of Companies shall mean the sum of subscribed and paid up equity and/or reserves from which shall be deducted the sum of revaluation reserves, miscellaneous expenditure not written off and reserves not

available for distribution to equity share holders and in the case of Trusts/Societies shall mean the sum of total contribution held by Trust/Society, Share Capital, Capital Grant, Total Corpus, Reserves & Surplus from which shall be deducted the sum of revaluation reserves, miscellaneous expenditure not written off and accumulated losses;

- (iv) in case of an Applicant eligible under the provisions of Clause 2.2.2
   (c) (i) or 2.2.2 (c) (ii), certificate(s) from the relevant District Education Officer or equivalent officer of the Education Department or certificate by an auditor and relevant evidence from the affiliating Board or other relevant entity;
- (v) statements in respect of the experience of the Applicant as per formats at Annex-II, Annex-IV and Annex-V of Appendix I;
- (vi) its preference, in the format at Annex-VI of Appendix-I, of the Blocks where it is willing to set up the School(s) if it is selected in the Bidding Process. The Applicant shall specify the Blocks in the order of their preference. An Applicant may apply for up to 20 (twenty) Schools for every 1 (one) School that it is eligible to set up under Clause 2.2.2 of this RFQ;
- (vii) audited annual reports of the Applicant for the last 2 (two) financial years preceding the year in which the Application is made, and in case the annual accounts for the latest financial year are not audited, the statutory auditor shall certify the same and the Applicant shall provide the audited annual reports for 2 (two) years preceding the year for which the audited annual report is not being provided; and
- (viii) where applicable, provide details of Net worth in the format at Annex-III of Appendix-I.
- 2.2.4 The Applicant should submit a Power of Attorney as per the format at Appendix-II, authorising the signatory of the Application to commit the Applicant.
- 2.2.5 Any entity which has been barred by the Central/ State Government, or any entity controlled by it, from participating in any project (BOT or otherwise), and the bar subsists as on the date of Application, would not be eligible to submit an Application.
- 2.2.6 An Applicant including its Associate should, in the last 3 (three) years, have neither failed to perform on any contract, as evidenced by imposition of a penalty by an arbitral or judicial authority or a judicial pronouncement or arbitration award against the Applicant or Associate, as the case may be, nor has been expelled from any project or contract by any public entity nor

have had any contract terminated by any public entity for breach by such Applicant or Associate.

2.2.7 In computing the eligibility of the Applicant under Clause 2.2.2, eligibility of their respective Associates would also be reckoned hereunder.

For purposes of this RFQ, Associate means, in relation to the Applicant, a person who controls, is controlled by, or is under the common control with such Applicant (the "Associate"). As used in this definition, the expression "control" means, with respect to a person (i) which is a company, the ownership, directly or indirectly, of more than 50% (fifty per cent) of the voting shares of such company and (ii) with respect to a person which is not a company, the power to direct the management and policies of such person by its constituent documents or by operation of law or the power to nominate a majority of the trustees and/or members of the governing council or has contributed more than 50% (fifty per cent) of the total resources of the trust or society during the past 3 (three) years.

2.2.8 The following conditions shall be adhered to while submitting an application:

- (a) Applicants should attach clearly marked and referenced continuation sheets in the event that the space provided in the prescribed forms in the Annexes is insufficient. Alternatively, Applicants may format the prescribed forms making due provision for incorporation of the requested information;
- (b) information supplied by an Applicant must apply to the Applicant or Associate named in the Application and not, unless specifically requested, to other associated companies or firms or persons. Invitation to submit Bids will be issued only to Applicants whose identity and/ or constitution is identical to that at pre-qualification; and
- (c) in responding to the pre-qualification submissions, Applicants should demonstrate their capabilities in accordance with Clause 2.2.2.
- 2.2.9 Notwithstanding anything to the contrary contained herein, in the event that the Application Due Date falls within 3 (three) months of the closing of the latest financial year of an Applicant, it shall ignore such financial year for the purposes of its Application and furnish all its information and certification with reference to 2 (two) years preceding its latest financial year. For the avoidance of doubt, financial year shall, for the purposes of an Application hereunder, mean the accounting year followed by the Applicant in the course of its normal business or activity.

## 2.3 Number of Applications and costs thereof

- 2.3.1 The Applicant may make a single Application for the Qualification Stage in response to this RFQ. It may determine its eligibility for award of the maximum number of Schools in accordance with the provisions of Clause 2.2.2 and apply hereunder for upto 20 (twenty) times such maximum number. Upon Qualification of the Applicant, it may be asked to submit its Bid in response to the RFP issued by the Authority for the respective Schools for which it has indicated its preference in its Application.
- 2.3.2 No entity would be awarded more than 20 Schools in a State and 50 Schools in the country under the Scheme of Rashtriya Adarsh Vidyalaya.
- 2.3.3 The Applicants shall be responsible for all of the costs associated with the preparation of their Applications and their participation in the Bid Process. The Authority will not be responsible or in any way liable for such costs, regardless of the conduct or outcome of the Bidding Process.

## 2.4 Site visit and verification of information

Applicants are encouraged to submit their respective Applications after ascertaining for themselves the site conditions, location, surroundings, climate, availability of power, water and other utilities for construction, access to site, handling and storage of materials, weather data, applicable laws and regulations, availability of students, educational levels, other schools in adjacent areas, means of transport for children and any other matter considered relevant by them.

## 2.5 Acknowledgement by Applicant

- 2.5.1 It shall be deemed that by submitting the Application, the Applicant has:
  - (a) made a complete and careful examination of the RFQ;
  - (b) received all relevant information requested from the Authority;
  - (c) accepted the risk of inadequacy, error or mistake in the information provided in the RFQ or furnished by or on behalf of the Authority relating to any of the matters referred to in Clause 2.4 above; and
  - (d) agreed to be bound by the undertakings provided by it under and in terms hereof.
- 2.5.2 The Authority shall not be liable for any omission, mistake or error in respect of any of the above or on account of any matter or thing arising out of or concerning or relating to the RFQ or the Bidding Process, including

any error or mistake therein or in any information or data given by the Authority.

#### 2.6 Right to accept or reject any or all Applications/ Bids

- 2.6.1 Notwithstanding anything contained in this RFQ, the Authority reserves the right to accept or reject any Application or Bid and to annul the Bidding Process and reject all Applications or Bids, at any time without any liability or any obligation for such acceptance, rejection or annulment, and without assigning any reasons therefor. In the event that the Authority rejects or annuls all the Bids, it may, in its discretion, invite all eligible Bidders to submit fresh Bids hereunder.
- 2.6.2 The Authority reserves the right to reject any Application and/ or Bid if:
  - (a) at any time, a material misrepresentation is made or uncovered, or
  - (b) the Applicant does not provide, within the time specified by the Authority, the supplemental information sought by the Authority for evaluation of the Application or Bid.

If such disqualification/ rejection occurs after the Bids have been opened and the Highest Bidder gets disqualified/ rejected, then the Authority reserves the right to:

- (i) invite the Bidder with the next highest score; or
- (ii) take any such measure as may be deemed fit in the sole discretion of the Authority, including annulment of the Bidding Process.
- 2.6.3 In case it is found during the evaluation or at any time before signing of the Concession Agreement, or after its execution and during the period of subsistence thereof, that one or more of the pre-qualification conditions contained in this RFQ have not been met by the Applicant, or the Applicant has made material misrepresentation or has given any materially incorrect or false information, the Applicant shall be disgualified forthwith if not yet appointed as the Concessionaire either by issuance of the LOA or entering into of the Concession Agreement, and if the Applicant has already been issued the LOA or has entered into the Concession Agreement, as the case may be, the same shall, notwithstanding anything to the contrary contained therein or in this RFQ, be liable to be terminated, by a communication in writing by the Authority to the Applicant, without the Authority being liable in any manner whatsoever to the Applicant and without prejudice to any other right or remedy which the Authority may have under this RFQ, the Bidding Documents, the Concession Agreement or under applicable law.

2.6.4 The Authority reserves the right to verify all statements, information and documents submitted by the Applicant in response to the RFQ. Any such verification or lack of such verification by the Authority shall not relieve the Applicant of its obligations or liabilities hereunder nor will it affect any rights of the Authority thereunder.

# **B. DOCUMENTS**

## 2.7 Contents of the RFQ

This RFQ comprises the disclaimer set forth hereinabove, the contents as listed below, and will additionally include any Addenda issued in accordance with Clause 2.9.

## **Invitation for Qualification**

Section 1.	Introduction
Section 2.	Instructions to Applicants
Section 3.	Criteria for Evaluation
Section 4.	Fraud & Corrupt Practices
Section 5.	Pre Application Conference
Section 6.	Miscellaneous

## Appendices

- I. Letter comprising the Application
- II. Power of Attorney for signing of Application
- III. Guidelines of the Department of Disinvestment
- IV. Information Memorandum for 500 model schools through PPP
- V. List of 3,203 Blocks identified for Rashtriya Adarsh Vidyalayas

## 2.8 Clarifications

2.8.1 Applicants requiring any clarification on the RFQ may notify the Authority in writing or by fax and e-mail in accordance with Clause 1.2.11. They should send in their queries before the date specified in the schedule of Bidding Process contained in Clause 1.3. The Authority shall endeavour to respond to the queries within the period specified therein, but no later than 10 (ten) days prior to the Application Due Date. The responses will be sent by fax and/or e-mail. The Authority will forward all the queries and its responses thereto, to all purchasers of the RFQ without identifying the source of queries.

- 2.8.2 The Authority shall endeavour to respond to the questions raised or clarifications sought by the Applicants. However, the Authority reserves the right not to respond to any question or provide any clarification, in its sole discretion, and nothing in this Clause shall be taken or read as compelling or requiring the Authority to respond to any question or to provide any clarification.
- 2.8.3 The Authority may also on its own motion, if deemed necessary, issue interpretations and clarifications to all Applicants. All clarifications and interpretations issued by the Authority shall be deemed to be part of the RFQ. Verbal clarifications and information given by Authority or its employees or representatives shall not in any way or manner be binding on the Authority.

## 2.9 Amendment of RFQ

- 2.9.1 At any time prior to the deadline for submission of Application, the Authority may, for any reason, whether at its own initiative or in response to clarifications requested by an Applicant, modify the RFQ by the issuance of Addenda.
- 2.9.2 Any Addendum thus issued will be sent in writing to all those who have purchased the RFQ.
- 2.9.3 In order to afford the Applicants a reasonable time for taking an Addendum into account, or for any other reason, the Authority may, in its sole discretion, extend the Application Due Date.<sup>\$</sup>

## C. PREPARATION AND SUBMISSION OF APPLICATION

## 2.10 Language

The Application and all related correspondence and documents in relation to the Bidding Process shall be in English language. Supporting documents and printed literature furnished by the Applicant with the Application may be in any other language provided that they are accompanied by translations of all the pertinent passages in the English language, duly authenticated and certified by the Applicant. Supporting materials, which are not translated

<sup>&</sup>lt;sup>§</sup> While extending the Application Due Date on account of an addendum, the Authority shall have due regard for the time required by bidders to address the amendments specified therein. In the case of significant amendments, at least 15 (fifteen) days shall be provided between the date of amendment and the Application Due Date, and in the case of minor amendments, at least 7 (seven) days shall be provided.

into English, may not be considered. For the purpose of interpretation and evaluation of the Application, the English language translation shall prevail.

# 2.11 Format and signing of Application

- 2.11.1 The Applicant shall provide all the information sought under this RFQ. The Authority will evaluate only those Applications that are received in the required formats and complete in all respects. Incomplete and /or conditional Applications shall be liable to rejection.
- 2.11.2 The Applicant shall prepare 1 (one) original set of the Application (together with originals/ copies of documents required to be submitted along therewith pursuant to this RFQ) and clearly marked "ORIGINAL". In addition, the Applicant shall submit 1 (one) copy of the Application, along with documents required to be submitted along therewith pursuant to this RFQ, marked "COPY". The Applicant shall also provide 2 (two) soft copies on Compact Disc (CD). In the event of any discrepancy between the original and the copy, the original shall prevail.
- 2.11.3 The Application and its copy shall be typed or written in indelible ink and signed by the authorised signatory of the Applicant who shall also initial each page in blue ink. In case of printed and published documents, only the cover shall be initialled. All the alterations, omissions, additions or any other amendments made to the Application shall be initialled by the person(s) signing the Application. The Application shall contain page numbers and shall be bound together in hard cover.

# 2.12 Sealing and Marking of Applications

- 2.12.1 The Applicant shall submit the Application in the format specified at Appendix-I, together with the documents specified in Clause 2.12.2, and seal it in an envelope and mark the envelope as "APPLICATION". The Applicant shall seal the original and the copy of the Application, together with their respective enclosures, in separate envelopes duly marking the envelopes as "ORIGINAL" and "COPY". The envelopes shall then be sealed in an outer envelope which shall also be marked in accordance with Clauses 2.12.2.
- 2.12.2 Each envelope shall contain:
  - (i) Application in the prescribed format (Appendix-I) along with Annexes and supporting documents;
  - (ii) Power of Attorney for signing the Application as per the format at Appendix-II;
  - (iii) copy of the certificate of registration and other requisite documents under the Companies Act, 1956, the Societies Registration Act,

1860, Indian Trusts Act, 1882 or the Bombay Public Trusts Act, 1950 (or other applicable laws), as applicable; along with copies of the documents of incorporation, Trust Deed, Memorandum of Association or Articles of Association, as applicable; and

(iv) copies of Applicant's duly audited balance sheet and profit and loss account for the preceding 3 (three) years.

Each of the envelopes shall clearly bear the following identification:

"Application for Selection: Rashtriya Adarsh Vidyalayas"

and shall clearly indicate the name and address of the Applicant. In addition, the Application Due Date should be indicated on the right hand corner of each of the envelopes.

2.12.3 Each of the envelopes shall be addressed to:

Smt. Radha S. Chauhan, Joint Secretary (SE.I), Department of School Education, Ministry of Human Resource Development, Government of India, Room No.107-D, Shastri Bhawan, New Delhi-110001 Tel: 23388632 Fax:23384306 Email. radha.edu@nic.in

- 2.12.4 If the envelopes are not sealed and marked as instructed above, the Authority assumes no responsibility for the misplacement or premature opening of the contents of the Application and consequent losses, if any, suffered by the Applicant.
- 2.12.5 Applications submitted by fax, telex, telegram or e-mail shall not be entertained and shall be rejected.

## 2.13 Application Due Date

- 2.13.1 Applications should be submitted before 1100 hours IST on the Application Due Date, at the address provided in Clause 2.12.3 in the manner and form as detailed in this RFQ. A receipt thereof should be obtained from the person specified in Clause 2.12.3.
- 2.13.2 The Authority may, in its sole discretion, extend the Application Due Date by issuing an Addendum in accordance with Clause 2.9 uniformly for all Applicants.

# 2.14 Late Applications

Applications received by the Authority after the specified time on the Application Due Date shall not be eligible for consideration and shall be summarily rejected.

#### 2.15 Modifications/ substitution/ withdrawal of Applications

- 2.15.1 The Applicant may modify, substitute or withdraw its Application after submission, provided that written notice of the modification, substitution or withdrawal is received by the Authority prior to the Application Due Date. No Application shall be modified, substituted or withdrawn by the Applicant on or after the Application Due Date.
- 2.15.2 The modification, substitution or withdrawal notice shall be prepared, sealed, marked, and delivered in accordance with Clause 2.12, with the envelopes being additionally marked "MODIFICATION", "SUBSTITUTION" or "WITHDRAWAL", as appropriate.
- 2.15.3 Any alteration/ modification in the Application or additional information supplied subsequent to the Application Due Date, unless the same has been expressly sought for by the Authority, shall be disregarded.

## D. EVALUATION PROCESS

#### 2.16 Opening and Evaluation of Applications

- 2.16.1 The Authority shall open the Applications at 1130 hours IST on the Application Due Date, at the place specified in Clause 2.12.3 and in the presence of the Applicants who choose to attend.
- 2.16.2 Applications for which a notice of withdrawal has been submitted in accordance with Clause 2.15 shall not be opened.
- 2.16.3 The Authority will subsequently examine and evaluate Applications in accordance with the provisions set out in this Section 2 and in Section 3.
- 2.16.4 Applicants are advised that pre-qualification of Applicants will be entirely at the discretion of the Authority. Applicants will be deemed to have understood and agreed that no explanation or justification on any aspect of the Bidding Process or selection will be given.
- 2.16.5 Any information contained in the Application shall not in any way be construed as binding on the Authority, its agents, successors or assigns, but shall be binding against the Applicant if the Project is subsequently awarded to it on the basis of such information.

- 2.16.6 The Authority reserves the right not to proceed with the Bidding Process at any time without notice or liability and to reject any or all Application(s) without assigning any reasons.
- 2.16.7 If any information furnished by the Applicant is found to be incomplete, or contained in formats other than those specified herein, the Authority may, in its sole discretion, exclude the relevant information of the Applicant from consideration.
- 2.16.8 In the event that an Applicant's claim is determined by the Authority as incorrect or erroneous, the Authority shall reject such claim and exclude the same from consideration. Where any information is found to be patently false or amounting to a material misrepresentation, the Authority reserves the right to reject the Application and/ or Bid in accordance with the provisions of Clauses 2.6.2 and 2.6.3.

## 2.17 Confidentiality

Information relating to the examination, clarification, evaluation, and recommendation for the pre-qualified Applicants shall not be disclosed to any person who is not officially concerned with the process or is not a retained professional advisor advising the Authority in relation to, or matters arising out of, or concerning the Bidding Process. The Authority will treat all information, submitted as part of Application, in confidence and will require all those who have access to such material to treat the same in confidence. The Authority may not divulge any such information unless it is directed to do so by any statutory entity that has the power under law to require its disclosure or is to enforce or assert any right or privilege of the statutory entity and/ or the Authority or as may be required by law or in connection with any legal process.

## 2.18 Tests of responsiveness

- 2.18.1 Prior to evaluation of Applications, the Authority shall determine whether each Application is responsive to the requirements of the RFQ. An Application shall be considered responsive only if:
  - (a) it is received as per format at Appendix-I.
  - (b) it is received by the Application Due Date including any extension thereof pursuant to Clause 2.13.2;
  - (c) it is signed, sealed, bound together and marked as stipulated in Clauses 2.11 and 2.12;
  - (d) it is accompanied by the Power of Attorney as specified in Clause 2.2.4;

- (e) it contains all the information and documents (complete in all respects) as requested in this RFQ;
- (f) it contains information in formats same as those specified in this RFQ;
- (g) it contains certificates from the Auditors<sup>¥</sup>, affiliating bodies, and District Education Officer<sup>\$</sup> as specified in the formats at Appendix-I of the RFQ;
- (h) it contains an attested copy of the receipt for payment of Rs. 10,000 (Rupees ten thousand only) to Authority towards the cost of the RFQ document;
- (i) it does not contain any condition or qualification; and
- (j) it is not non-responsive in terms hereof.
- 2.18.2 The Authority reserves the right to reject any Application which is nonresponsive and no request for alteration, modification, substitution or withdrawal shall be entertained by the Authority in respect of such Application.

#### 2.19 Clarifications

- 2.19.1 To facilitate evaluation of Applications, the Authority may, at its sole discretion, seek clarifications from any Applicant regarding its Application. Such clarification(s) shall be provided within the time specified by the Authority for this purpose. Any request for clarification(s) and all clarification(s) in response thereto shall be in writing.
- 2.19.2 If an Applicant does not provide clarifications sought under Clause 2.19.1 above within the prescribed time, its Application shall be liable to be rejected. In case the Application is not rejected, the Authority may proceed to evaluate the Application by construing the particulars requiring clarification to the best of its understanding, and the Applicant shall be barred from subsequently questioning such interpretation of the Authority.

# E. QUALIFICATION AND BIDDING

## 2.20 Pre-qualification and notification

<sup>&</sup>lt;sup>¥</sup> In case duly certified audited annual financial statements containing the requisite details are provided, a separate certification by auditors would not be necessary in respect of Clause 2.18.1 (g).

<sup>&</sup>lt;sup>\$</sup> This certificate may be given by any officer of the Education Department of the Central Government or the relevant State Government, but not below the rank of a District Education Officer in the case of aided schools or certificate by an auditor and relevant evidence from the affiliating Board in case of a non-aided school.

- 2.20.1 After the evaluation of Applications, the Authority would announce a list of pre-qualified Applicants (Bidders) who will be eligible for participation in the Bid Stage. At the same time, the Authority would notify the other Applicants that they have not been short-listed. The Authority will not entertain any query or clarification from Applicants who fail to qualify.
- 2.20.2 In Phase-I of this Scheme for award of 500 (five hundred) Schools, the pilot stage will comprise the Bidding Process for award of 50 (fifty) Schools. The Bidders who have applied for any of the 50 (fifty) Schools selected by the Authority shall be invited to submit their Bids in the Bid Stage only for the schools in the blocks for which they have indicated their preference in their application.

#### 2.21 Submission of Bids

The Bidders will be requested to submit a Bid in the form and manner to be set out in the Bidding Documents.

Only pre-qualified Applicants shall be invited by the Authority to submit their Bids for the Project. The Authority is likely to provide a comparatively short time span for submission of the Bids for the Project. The Applicants are therefore advised to familiarise themselves with the Project by the time of submission of the Application. No extension of time is likely to be considered for submission of Bids pursuant to invitation that may be issued by the Authority.

## 2.22 Proprietary data

All documents and other information supplied by the Authority or submitted by an Applicant to the Authority shall remain or become the property of the Authority. Applicants are to treat all information as strictly confidential and shall not use it for any purpose other than for preparation and submission of their Application. The Authority will not return any Application or any information provided along therewith.

#### 2.23 Correspondence with the Applicant

Save and except as provided in this RFQ, the Authority shall not entertain any correspondence with any Applicant in relation to the acceptance or rejection of any Application.

#### 2.24 Validity of Application

The Applications shall remain valid for the bid process in respect of 500 (five hundred) schools to be awarded in Phase I of the Scheme. The pilot stage will comprise the Bidding Process for award of 50 (fifty) schools. Thereafter, the Applicants may be given an opportunity to update their Applications or provide additional information in respect of the Bid Process for the remaining 450 (four hundred and fifty) schools.

# 3. CRITERIA FOR EVALUATION

#### **3.1** Evaluation of Applications

- 3.1.1 The Applicant's eligibility for participation in the Bid Stage shall be evaluated and established by the eligibility conditions laid down in Clause 2.2 and in accordance with the provisions of Section 2 above.
- 3.1.2 The list of Blocks provided by the Applicants in the format at Annex-VI of Appendix-I shall be used for selecting the 50 (fifty) Blocks for which Bids would be invited in the pilot phase for award of 50 (fifty) Schools. Applicants meeting the eligibility conditions shall be pre-qualified and invited to Bid for such of the 50 Blocks for which they have indicated their preferences. The remaining Blocks may be taken up for bidding in a subsequent round.

#### **3.2** Evaluation of Bidders

- 3.2.1 The eligible Applicants will be entitled to submit their Bids in response to the RFP for the respective Schools. However, pre-qualification should only be regarded as a necessary condition for entry into the Bid Stage. The evaluation of Bidders in the Bid Stage will involve an objective assessment of their credentials and track record to determine their suitability and/ or financial capacity for setting up and running the School(s) as centres of excellence. The evaluation will be undertaken through a transparent and objective scoring methodology that would be indicated in the RFP. Only those Bidders who achieve the minimum technical score specified in the RFP will be considered for award of schools. For the purposes of ranking among competing Bidders applying for the same Block, the Bidder who achieves the highest technical score shall be the highest Bidder.
- 3.2.2 For purposes of evaluation and ranking among Bidders at the RFP stage, scores will be assigned, *inter alia*, on the following parameters:
  - (a) Nature and extent of experience in running schools affiliated with CBSE; schools affiliated with other Boards; and other Educational Institutions;
  - (b) qualitative outcomes of the aforesaid schools and Educational Institutions measured in terms of the pass percentage, number of students securing distinction etc.;
  - (c) the length of experience in operation of the aforesaid schools and Educational Institutions; and
  - (d) Net Worth of the Applicant where applicable.

3.2.3 Applicants are advised to assess their own track record with reference to the aforesaid parameters before making an Application. Among those eligible Applicants who achieve the minimum score specified in RFP, the award of a School will depend on the comparative scores of the respective Bidders who submit Bids for and in respect of the School for which an RFP is issued. The Bidder with the highest technical score shall be awarded the School. There shall be no financial bids.

# 4. Fraud and corrupt practices

- 4.1 The Applicants and their respective officers, employees, agents and advisers shall observe the highest standard of ethics during the Bidding Process. Notwithstanding anything to the contrary contained herein, the Authority may reject an Application without being liable in any manner whatsoever to the Applicant if it determines that the Applicant has, directly or indirectly or through an agent, engaged in corrupt practice, fraudulent practice, coercive practice, undesirable practice or restrictive practice in the Bidding Process.
- 4.2 Without prejudice to the rights of the Authority under Clause 4.1 hereinabove, if an Applicant is found by the Authority to have directly or indirectly or through an agent, engaged or indulged in any corrupt practice, fraudulent practice, coercive practice, undesirable practice or restrictive practice during the Bidding Process, such Applicant shall not be eligible to participate in any tender or RFQ issued by the Authority during a period of 2 (two) years from the date such Applicant is found by the Authority to have directly or indirectly or through an agent, engaged or indulged in any corrupt practice, fraudulent practice, coercive practice, undesirable practice or restrictive or restrictive practice, fraudulent practice, coercive practice, undesirable practice or restrictive practice, as the case may be.
- 4.3 For the purposes of this Clause 4, the following terms shall have the meaning hereinafter respectively assigned to them:
  - "corrupt practice" means (i) the offering, giving, receiving, or (a) soliciting, directly or indirectly, of anything of value to influence the actions of any person connected with the Bidding Process (for avoidance of doubt, offering of employment to, or employing, or engaging in any manner whatsoever, directly or indirectly, any official of the Authority who is or has been associated in any manner, directly or indirectly, with the Bidding Process or the LOA or has dealt with matters concerning the Concession Agreement or arising therefrom, before or after the execution thereof, at any time prior to the expiry of one year from the date such official resigns or retires from or otherwise ceases to be in the service of the Authority, shall be deemed to constitute influencing the actions of a person connected with the Bidding Process); or (ii) save and except as permitted under sub clause (c) of Clause 2.2.1, engaging in any manner whatsoever, whether during the Bidding Process or after the issue of the LOA or after the execution of the Concession Agreement, as the case may be, any person in respect of any matter relating to the Project or the LOA or the Concession Agreement, who at any time has been or is a legal, financial or technical adviser of the Authority in relation to any matter concerning the Project;

- (b) **"fraudulent practice"** means a misrepresentation or omission of facts or suppression of facts or disclosure of incomplete facts, in order to influence the Bidding Process;
- (c) **"coercive practice"** means impairing or harming or threatening to impair or harm, directly or indirectly, any person or property to influence any person's participation or action in the Bidding Process;
- (d) **"undesirable practice"** means (i) establishing contact with any person connected with or employed or engaged by the Authority with the objective of canvassing, lobbying or in any manner influencing or attempting to influence the Bidding Process; or (ii) having a Conflict of Interest; and
- (e) **"restrictive practice"** means forming a cartel or arriving at any understanding or arrangement among Applicants with the objective of restricting or manipulating a full and fair competition in the Bidding Process.

# 5. PRE-APPLICATION CONFERENCE

- 5.1 A Pre-Application conference of the interested parties shall be convened at the designated date, time and place. Only those persons who have purchased the RFQ document shall be allowed to participate in the Pre-Application conference. Applicants who have downloaded the RFQ document from the Authority's website (<u>www.mhrd.gov.in</u>) should submit a demand draft of Rs. 10,000 (Rupees ten thousand only) towards the cost of document, through their representative attending the conference. A maximum of three representatives of each Applicant shall be allowed to participate on production of authority letter from the Applicant.
- 5.2 During the course of Pre-Application conference, the Applicants will be free to seek clarifications and make suggestions for consideration of the Authority. The Authority shall endeavour to provide clarifications and such further information as it may, in its sole discretion, consider appropriate for facilitating a fair, transparent and competitive Bidding Process.

#### 6. MISCELLANEOUS

- 6.1 The Bidding Process shall be governed by, and construed in accordance with, the laws of India and the Courts at New Delhi shall have exclusive jurisdiction over all disputes arising under, pursuant to and/ or in connection with the Bidding Process.
- 6.2 The Authority, in its sole discretion and without incurring any obligation or liability, reserves the right, at any time, to;
  - (a) suspend and/ or cancel the Bidding Process and/ or amend and/ or supplement the Bidding Process or modify the dates or other terms and conditions relating thereto;
  - (b) consult with any Applicant in order to receive clarification or further information;
  - (c) pre-qualify or not to pre-qualify any Applicant and/ or to consult with any Applicant in order to receive clarification or further information;
  - (d) retain any information and/ or evidence submitted to the Authority by, on behalf of, and/ or in relation to any Applicant; and/ or
  - (e) independently verify, disqualify, reject and/ or accept any and all submissions or other information and/ or evidence submitted by or on behalf of any Applicant.
- 6.3 It shall be deemed that by submitting the Application, the Applicant agrees and releases the Authority, its employees, agents and advisers, irrevocably, unconditionally, fully and finally from any and all liability for claims, losses, damages, costs, expenses or liabilities in any way related to or arising from the exercise of any rights and/ or performance of any obligations hereunder and the Bidding Documents, pursuant hereto, and/ or in connection with the Bidding Process, to the fullest extent permitted by applicable law, and waives any and all rights and/ or claims it may have in this respect, whether actual or contingent, whether present or in future.

## APPENDIX I

# Letter Comprising the Application for Pre-Qualification

(Refer Clause 2.12.1)

Dated:

To,

## Smt. Radha S. Chauhan, Joint Secretary (SE.I), Department of School Education, Ministry of Human Resource Development, Government of India, Room No.107-D, Shastri Bhawan, New Delhi-110001

Sub: Application for pre-qualification for 50 Rashtriya Adarsh Vidyalayas.

Dear Madam,

With reference to your RFQ document dated ....., I/we, having examined the RFQ document and understood its contents, hereby submit my/our Application for Pre-Qualification for the aforesaid project. The Application is unconditional and unqualified.

2. I/ We acknowledge that the Authority will be relying on the information provided in the Application and the documents accompanying such Application for pre-qualification of the Applicants for the aforesaid project, and we certify that all information provided in the Application and in Annexes I to VII is true and correct; nothing has been omitted which renders such information misleading; and all documents accompanying such Application are true copies of their respective originals.

3. This statement is made for the express purpose of qualifying as an Applicant for the development, operation and management of the aforesaid Schools.

4. I shall make available to the Authority any additional information it may find necessary or require to supplement or authenticate the Qualification statement.

5. I acknowledge the right of the Authority to reject our Application without assigning any reason or otherwise and hereby waive, to the fullest extent permitted by applicable law, our right to challenge the same on any account whatsoever.

6. I certify that in the last three years, we or our Associates have neither failed to perform on any contract, as evidenced by imposition of a penalty by an arbitral

or judicial authority or a judicial pronouncement or arbitration award, nor been expelled from any project or contract by any public authority nor have had any contract terminated by any public authority for breach on our part.

- 7. I/We declare that:
  - (a) I have examined and have no reservations to the RFQ document, including any Addendum issued by the Authority;
  - (b) I do not have any conflict of interest in accordance with Clauses 2.2.1(c) and 2.2.1(d) of the RFQ document;
  - (c) I have not directly or indirectly or through an agent engaged or indulged in any corrupt practice, fraudulent practice, coercive practice, undesirable practice or restrictive practice, as defined in Clause 4.3 of the RFQ document, in respect of any tender or request for proposal issued by or any agreement entered into with the Authority or any other public sector enterprise or any government, Central or State; and
  - (d) I hereby certify that we have taken steps to ensure that in conformity with the provisions of Section 4 of the RFQ document, no person acting for us or on our behalf has engaged or will engage in any corrupt practice, fraudulent practice, coercive practice, undesirable practice or restrictive practice.

8. I understand that you may cancel the Bidding Process at any time and that you are neither bound to accept any Application that you may receive nor to invite the Applicants to Bid for the School, without incurring any liability to the Applicants, in accordance with Clause 2.16.6 of the RFQ document.

9. I believe that we satisfy the eligibility criteria and meet all the requirements as specified in the RFQ document and are qualified to submit a Bid.

10. I certify that in regard to matters other than security and integrity of the country, we or any of our Associates have not been convicted by a Court of Law or indicted or adverse orders passed by a regulatory authority which could cast a doubt on our ability to undertake the School or which relates to a grave offence that outrages the moral sense of the community.

11. I further certify that in regard to matters relating to security and integrity of the country, we or any of our Associates have not been charge-sheeted by any agency of the Government or convicted by a Court of Law.

12. I further certify that no investigation by a regulatory authority is pending either against us or against our Associates or against our CEO or any of our directors/ trustees/managers/members/employees.

13. I undertake that in case due to any change in facts or circumstances during the Bidding Process, we are attracted by the provisions of disqualification in terms of the provisions of this RFQ, we shall intimate the Authority of the same immediately.

14. The Statement of Legal Capacity as per format provided at Annex-VII in Appendix-I of the RFQ document, and duly signed, is enclosed. The power of attorney for signing of application as per format provided at Appendix II of the RFQ, is also enclosed.

15. I understand that the selected Bidder shall set up a School(s) either as a trust registered under the Indian Trusts Act, 1882 or the Bombay Public Trusts Act, 1950, or a society registered under the Societies Registration Act, 1860 (or other applicable laws) or a not-for-profit company incorporated under Section 25 of the Companies Act, 1956.

16. I hereby irrevocably waive any right or remedy which we may have at any stage at law or howsoever otherwise arising to challenge or question any decision taken by the Authority in connection with the selection of Applicants, selection of the Bidder, or in connection with the selection/ Bidding Process itself, in respect of the above mentioned Schools and the terms and implementation thereof.

18. I agree and undertake to abide by and fulfill all the terms and conditions of the RFQ document. {In particular, I/ we undertake to make a deposit of Rs. 25 lakh/Rs. 50 lakh per school in terms of Clause 2.2.2 of the RFQ.}

In witness thereof, I submit this application under and in accordance with the terms of the RFQ document.

Yours faithfully,

Date: (Signature, name and designation of the Authorised Signatory)

Place:

Name and seal of the Applicant

# Appendix I Annex-I

#### ANNEX-I

# **Details of Applicant**

- 1. (a) Name:
  - (b) Place of incorporation/registration:
  - (c) Address of the headquarters and its branch office(s), if any:

(d) Date of incorporation or registration and commencement of its activities:

- 2. Brief description of the Trust/ Society/ Company including details of its main lines of activities and proposed role and responsibilities in this Project:
- 3. Details of individual(s) who will serve as the point of contact/ communication for the Authority:
  - (a) Name:
  - (b) Designation:
  - (c) Company/Trust/Society:
  - (d) Address:
  - (e) Telephone Number:
  - (f) E-Mail Address:
  - (g) Fax Number:
- 4. Particulars of the Authorised Signatory of the Applicant:
  - (a) Name:
  - (b) Designation:
  - (c) Address:
  - (d) Phone Number:
  - (e) Fax Number:

#### 5. The following information shall be provided for the Applicant:

# Name of Applicant:

No.	Criteria	Yes	No
1.	Has the Applicant been barred by the Central/ State Government, or any entity controlled by it, from participating in any project (BOT or otherwise)?		
2.	If the answer to 1 is yes, does the bar subsist as on the date of Application?		
3.	Has the Applicant paid liquidated damages of more than 5% of the contract value in a contract due to delay or has been penalised due to any other reason in relation to execution of a contract, in the last three years?		

6. A statement by the Applicant or any of its Associates disclosing material non-performance or contractual non-compliance in past projects, contractual disputes and litigation/ arbitration in the recent past is given below (Attach extra sheets, if necessary):

Appendix I Annex-II

#### ANNEX-II

# **Experience of the Applicant**<sup>@</sup> (*Refer Clause 2.2.2*)

Name of the Applicant/ Associate	Name of school/ Educational Institution	Board of affiliation/ accreditation	Experience		Number of maximum Schools that can be awarded against each school/
			Number of years in operation	No. of batches which passed qualifying exam as specified in Cl. 2.2.2 of RFQ	
(1)	(2)	(3)	(4)	(5)	(8)
	a				
	b				
	c				
	d				
	a				
	b				
	c				
	d				
	a				
	b				
	c				
	d				
				Total No. of schools that can be awarded	

<sup>(a)</sup> Provide details of only those schools that are being run by the Applicant under its own name and/ or by an Associate specified in Clause 2.2.7.

 $^{\$}$  The Applicant shall determine its eligibility for award of Schools in accordance with the provisions of Clause 2.2.2 (a), (b), (c) & (d), and specify the number of Schools that can be awarded against each school/Educational Institution for which information has been provided in Col.(2).

Appendix I Annex-III

#### ANNEX-III

# **Financial Capacity of the Applicant**

(Refer to Clauses 2.2.2 and, 2.2.3 of the RFQ)

#### (In Rs. crore)

Name of the Applicant/ Associate <sup>\$</sup>	Net Worth	Net Worth
	(2010-11)	(2011-12)
(1)	(2)	(3)

#### Name & address of Applicant's Bankers:

<sup>\$</sup>The Applicant should provide details of its own financial capacity or of an Associate specified in Clause 2.2.7.

### **Instructions:**

- 1. The Applicant shall attach copies of the balance sheets, financial statements and Annual Reports for 2 (two) years preceding the Application Due Date. The financial statements shall:
  - (a) reflect the financial situation of the Applicant or their Associates where the Applicant is relying on its Associate's financials;

- (b) be audited by an auditor;
- (c) be complete, including all notes to the financial statements; and
- (d) correspond to accounting periods already completed and audited (no statements for partial periods shall be requested or accepted).
- 2. Net Worth shall mean (Subscribed and Paid-up Equity + Reserves) less (Revaluation reserves + miscellaneous expenditure not written off + reserves not available for distribution to equity shareholders) and in the case of Trusts/Societies shall mean the sum of total contribution held by Trust/Society, Share Capital, Capital Grant, Total Corpus, Reserves & Surplus from which shall be deducted the sum of revaluation reserves, miscellaneous expenditure not written off and accumulated losses.
- 3. Information will be provided for the latest completed financial year, preceding the bidding. In case the Application Due Date falls within 3 (three) months of the close of the latest financial year, refer to Clause 2.2.9.
- 4. The Applicant shall also provide the name and address of the bankers to the Applicant.
- 5. The Applicant shall provide an Auditor's certificate specifying the Net Worth of the Applicant and also specifying the methodology adopted for calculating such Net Worth in accordance with Clause 2.2.3 (iii), of the RFQ document.

#### ANNEX-IV

# **Details of Existing School**

(Refer to Clauses 2.2.2 of the RFQ)

Item	Particulars
(1)	(2)
Name of the School and address	
Date of establishment	
Name of the affiliating Board	
CBSE/ICSE/Other Board's registration no. with date	
Location	
Details of batches that has passed out	
Whether credit is being taken for the experience of an Associate (Yes/ No)	
Number of Schools that can be awarded against the existing school/s	

# **Instructions:**

1. Applicants are expected to provide information in respect of each School mentioned in Annex II. The schools cited must comply with the relevant eligibility criteria specified in Clause 2.2.2 of the RFQ. Applicable certificates for each existing School must be furnished as per formats below.

#### **Certificate from the Competent Authority**

I further certify that . ...... (number) of batches comprising students of Class(es) ......have passed out from the School since ------ (year) as indicated below:

Year	No. of batches (class -wise)		
1			
2			
3			

Name and Designation of the Competent Authority <sup>@</sup>: Seal of the officer:

(Signature, name and designation of the authorized signatory)

Date:

<sup>®</sup> This certificate may be given by any officer of the Education Department of the Central Government or the relevant State Government, but not below the rank of a District Education Officer in case of aided schools or by an auditor with relevant evidence from the affiliating Board in case of a non-aided school.

2. In the event that credit is being taken for experience of an Associate, as defined in Clause 2.2.7, the Applicant should, in addition to a certificate in the above format, also provide a certificate in the format below:

#### Certificate from Auditor/Company Secretary regarding Associate<sup>\$</sup>

A brief description of the said equity held, directly or indirectly, is given below:

{Describe the share-holding of the Applicant and the Associate. In the event the Associate is under common control with the Applicant, the relationship may be suitably described and similarly certified herein}

Name of the audit firm:

Date:

(Signature, name and designation of the authorised signatory).

<sup>\$</sup> In the event that the Applicant exercises control over an Associate by virtue of its constituent documents or by operation of law, or in any other manner specified in Clause 2.2.7 of the RFQ, this certificate may be suitably modified and copies of the constituent documents and/or relevant law may be enclosed and referred to. In case of a Trust or Society, this Certificate must be obtained from the Auditor based on the authenticated record of the relevant Board of Trust or Registrar of Societies, as the case may be.

3. It may be noted that in the absence of any detail in the above certificates, the information would be considered inadequate and could lead to exclusion of the relevant school or Educational Institution in computation of experience of the Applicant.

# ANNEX-V

# **Details of Educational Institution**

(Refer to Clauses 2.2.2(c)(ii) and 2.2.3 of the RFQ)

Item	Particulars
(1)	(2)
Name of the Educational Institution and address	
Date of establishment	
Name of the affiliating/accrediting organisation	
Affiliating/accrediting organisation's registration no. with date	
Details of classes being run (degree/diploma)	
Location	
No. of batches of degree/diploma holders passed out	
Whether credit is being taken for the experience of an Associate (Yes/No)	
Number of Schools that can be awarded against the Educational Institution	

#### **Instructions:**

1. Applicants are expected to provide information in respect of each Educational Institution mentioned in Annex II. The institutions cited must comply with the relevant eligibility criteria specified in Clause 2.2.2 of the RFQ. Applicable certificates for each Educational Institution must be furnished as per formats below.

#### Certificate from the Affiliating/accrediting organisation

Name and Designation of the certifying authority<sup>@</sup>: Seal of the officer:

(Signature, name and designation of the authorized signatory)

Date:

 $^{\textcircled{0}}$  This certificate may be given by the authorised signatory of the affiliating/accrediting organisation.

2. In the event that credit is being taken for experience of an Associate, as defined in Clause 2.2.7, the Applicant should, in addition to a certificate in the above format, also provide a certificate in the format specified in Paragraph 2 of the Instructions in Annex IV of this Appendix-I.

Appendix I Annex-VI

#### ANNEX-VI

# **Preference for location of Schools**

(Refer to Clauses 2.3.1 & 3.1.2 of the RFQ)

Sl. No. <sup>£</sup>	S. No. in list of MHRD <sup>\$</sup>	Name of Block	Name of District	Name of State
1.				
2.				
3.				
4.				
5.				
6.				
7.				
8.				
9.				
10.				

In terms of Clause 2.2.2 (a)/(b)/(c)/(d) of the RFQ, I/we are eligible for award of a total Number of ..... Schools under this Project. In accordance with Clause 3.1.5 of the RFQ, I/we are eligible for pre-qualification in respect of Schools upto 20 (twenty) times the aforesaid number. Accordingly, I/we have specified the aforesaid Schools for which we intend to submit Bids for setting up school(s), if pre-qualified in the Qualification Process.

Signature, Name and Designation of the Authorised Signatory

 $<sup>^{\</sup>pounds}$  The serial numbers indicated in column 1 shall be treated as the order of preference for the respective Schools.

<sup>&</sup>lt;sup>\$</sup> Refer to Clauses 2.3.1 and 3.1.2 of RFQ. The location of Schools may be selected from the list of 3,203 Blocks at Appendix-V and the serial number specified in this column should be the same as specified at Appendix-V.

Appendix I Annex-VII

#### ANNEX-VII

# **Statement of Legal Capacity**

(To be forwarded on the letterhead of the Applicant)

Ref. Date:

To,

Smt. Radha S. Chauhan, Joint Secretary (SE.I), Department of School Education, Ministry of Human Resource Development, Government of India, Room No.107-D, Shastri Bhawan, New Delhi-110001

Dear Madam,

We hereby confirm that we satisfy the terms and conditions laid out in the RFQ document.

We have agreed that ..... (insert individual's name) will act as our representative on its behalf and has been duly authorized to submit the RFQ. Further, the authorised signatory is vested with requisite powers to furnish such letter and authenticate the same.

Thanking you,

Yours faithfully,

(Signature, name and designation of the authorised signatory)

For and on behalf of.....

#### APPENDIX II

# **Power of Attorney for signing of Application**

(Refer Clause 2.2.4)

Know all men by these presents, We..... (name of the trust/society/company incorporated under Section 25 of the Companies Act/ corporate and address of the registered office) do hereby irrevocably constitute, nominate, appoint and authorise Mr/ Ms (name), ..... son/daughter/wife of ..... and presently residing at ....., who is presently employed with us and holding the position of ....., as our true and lawful attorney (hereinafter referred to as the "Attorney") to do in our name and on our behalf, all such acts, deeds and things as are necessary or required in connection with or incidental to submission of our application for pre-qualification and submission of our bid for the Rashtriya Adarsh Vidyalayas Project proposed or being developed by the Department of School Education & Literacy, Ministry of Human Resource Development, Government of India (the "Authority") including but not limited to signing and submission of all applications, bids and other documents and writings, participate in Pre-Applications and other conferences and providing information/ responses to the Authority, representing us in all matters before the Authority, signing and execution of all contracts including the Concession Agreement and undertakings consequent to acceptance of our bid, and generally dealing with the Authority in all matters in connection with or relating to or arising out of our bid for the said Project and/ or upon award thereof to us and/or till the entering into of the Concession Agreement with the Authority.

AND we hereby agree to ratify and confirm and do hereby ratify and confirm all acts, deeds and things done or caused to be done by our said Attorney pursuant to and in exercise of the powers conferred by this Power of Attorney and that all acts, deeds and things done by our said Attorney in exercise of the powers hereby conferred shall and shall always be deemed to have been done by us.

For

.....

(Signature, name, designation and address)

Witnesses:

1.

(Notarised)

Appendix II Page 2

2.

Accepted

.....

(Signature)

(Name, Title and Address of the Attorney)

Notes:

- The mode of execution of the Power of Attorney should be in accordance with the procedure, if any, laid down by the applicable law and the charter documents of the executant(s) and when it is so required, the same should be under common seal affixed in accordance with the required procedure.
- Wherever required, the Applicant should submit for verification the extract of the charter documents and documents such as a board or shareholders' resolution/ power of attorney in favour of the person executing this Power of Attorney for the delegation of power hereunder on behalf of the Applicant.
- For a Power of Attorney executed and issued overseas, the document will also have to be legalised by the Indian Embassy and notarised in the jurisdiction where the Power of Attorney is being issued. However, the Power of Attorney provided by Applicants from countries that have signed the Hague Legislation Convention 1961 are not required to be legalised by the Indian Embassy if it carries a conforming Appostille certificate.

#### APPENDIX III

# **Guidelines of the Department of Disinvestment**

(Refer Clause 1.2.3)

No. 6/4/2001-DD-II Government of India Department of Disinvestment

> Block 14, CGO Complex New Delhi. Dated 13<sup>th</sup> July, 2001.

#### OFFICE MEMORANDUM

Sub: Guidelines for qualification of Bidders seeking to acquire stakes in Public Sector Enterprises through the process of disinvestment

Government has examined the issue of framing comprehensive and transparent guidelines defining the criteria for bidders interested in PSE-disinvestment so that the parties selected through competitive bidding could inspire public confidence. Earlier, criteria like Net Worth, experience etc. used to be prescribed. Based on experience and in consultation with concerned departments, Government has decided to prescribe the following additional criteria for the qualification/ disqualification of the parties seeking to acquire stakes in public sector enterprises through disinvestment:

- (a) In regard to matters other than the security and integrity of the country, any conviction by a Court of Law or indictment/ adverse order by a regulatory authority that casts a doubt on the ability of the bidder to manage the public sector unit when it is disinvested, or which relates to a grave offence would constitute disqualification. Grave offence is defined to be of such a nature that it outrages the moral sense of the community. The decision in regard to the nature of the offence would be taken on case to case basis after considering the facts of the case and relevant legal principles, by the Government of India.
- (b) In regard to matters relating to the security and integrity of the country, any charge-sheet by an agency of the Government/ conviction by a Court of Law for an offence committed by the bidding party or by any sister concern of the bidding party would result in disqualification. The decision in regard to the relationship between the sister concerns would be taken, based on the relevant facts and after examining whether the two concerns are substantially controlled by the same person/ persons.

- (c) In both (a) and (b), disqualification shall continue for a period that Government deems appropriate.
- (d) Any entity, which is disqualified from participating in the disinvestment process, would not be allowed to remain associated with it or get associated merely because it has preferred an appeal against the order based on which it has been disqualified. The mere pendency of appeal will have no effect on the disqualification.
- (e) The disqualification criteria would come into effect immediately and would apply to all bidders for various disinvestment transactions, which have not been completed as yet.
- (f) Before disqualifying a concern, a Show Cause Notice why it should not be disqualified would be issued to it and it would be given an opportunity to explain its position.
- (g) Henceforth, these criteria will be prescribed in the advertisements seeking Expression of Interest (EOI) from the interested parties. The interested parties would be required to provide the information on the above criteria, along with their Expressions of Interest (EOI). The bidders shall be required to provide with their EOI an undertaking to the effect that no investigation by a regulatory authority is pending against them. In case any investigation is pending against the concern or its sister concern or against its CEO or any of its Directors/ Managers/ employees, full details of such investigation including the name of the investigating agency, the charge/ offence for which the investigation has been launched, name and designation of persons against whom the investigation has been launched and other relevant information should be disclosed, to the satisfaction of the Government. For other criteria also, a similar undertaking shall be obtained along with EOI.

sd/-

(A.K. Tewari)

Under Secretary to the Government of India

# APPENDIX-IV

# Information Memorandum for 500 Model Schools through Public Private Partnership

The information contained herein is tentative and indicative. The precise terms of the scheme will be spelt out in the Concession Agreement which will form part of the bid documents at the RFP stage.

# 1. Preamble

- 1.1 This scheme for setting up of model school through Public Private Partnership (PPP) has been formulated in pursuance of the Government decision to set up 2,500 model schools through PPP in blocks to be selected out of 3,203 blocks which have been identified as Non-Educationally Backward Blocks. The principal objective of this scheme is to set up senior secondary schools, for providing access to quality education to the rural children.
- 1.2 It is well recognized that education is a pre-requisite for the growth and development of a modern and equitable socio-economic order. It is essential for empowering and enabling the economically weaker sections of the society to participate more fully in a democratic polity that identifies equality of opportunity as one of its cherished goals. For these reasons, education, especially school education, is regarded as the responsibility of the Government and the recent enactment of the Right to Education Act is a step in that direction.
- 1.3 Over the last 60 years, successive Governments have been committed to achieving the national goal of universal education and have steadily increased the budgetary allocation for education. At present, the expenditure on education in India is about 3.7% of the GDP and this needs to be increased to at least 6% of the GDP, especially for addressing unequal access to education in rural areas and among the urban poor.
- 1.4 The provision of universal school education is a complex task of gigantic proportions in a country like India. It is, therefore, essential to explore diverse options for providing school education and financing the same in pursuance of the education goals set by the Government. The scheme for setting up 6,000 new schools as model pace-setting institutions to set benchmarks across the country is an important initiative of the present Government and the decision to provide 2,500 schools out of 6,000 through Public Private Partnership (PPP) constitutes recognition of the need for plurality in the system of providing education services. The objective of all model schools, including those in the PPP mode, would be to enhance access to quality education for all, especially those who live in rural areas or under-served urban areas and belong to low-income families.

1.5 The scheme for 2,500 model schools under PPP mode should be viewed as an opportunity to evolve innovative ways to empower and enable non-government players to engage in providing quality education to children especially from low-income rural families. The objective should be to combine the respective strengths of the public and private sectors to complement each other in pursuit of the shared goal of good education for all. In particular, adoption of the PPP mode would lead to rapid expansion of access to quality education by low-income families.

# 2 Objective of the Scheme

#### 2.1 Creating access to quality education

Children from low-income families in rural as well as urban areas are unable to access high quality school education as they cannot afford to pay the fees normally charged by good private schools while good schools that can provide such access in the public sector are limited. To ensure inclusive growth and an equitable socio-economic order, affirmative action is necessary to accelerate the provision of quality education to those who have hitherto lacked access. The objective of this scheme is to set up 2,500 schools through Public Private Partnership (PPP) for providing quality education to about 40 lakh children, of which 25 lakh would be from socially and economically disadvantaged categories (the "Select Students"). These schools will reach out to the underprivileged children and educate them with a view to enabling them to avail of the opportunities offered by growth and development.

# 2.2 Centers of Excellence

The objective of this scheme is to create centers of excellence in school education by combining the respective strengths of the public and private sectors in a manner that would create models in the respective Blocks and set benchmarks for others to emulate. By focusing on low income families, these schools will help build a cohesive, secular and merit-based environment that would promote inclusive growth and train future leaders.

# **3 PPP-the concept**

3.1 PPP in school education is essentially an arrangement where the private sector partner participates in the provision of services traditionally provided by the Government. It is usually characterized by an agreement between the government and the private sector, with the latter undertaking to deliver an agreed service on the payment of a unitary charge by the Government. The need for PPP in school education primarily arises out of the Government's commitment to provide quality education to under-privileged children who cannot afford the tuition fee that a private school would normally charge. The justification for PPP schools arises primarily from the need to accelerate the expansion of education, supplement investment and enable different models for improving the quality of education.

- 3.2 PPPs are sometimes mistaken for privatization which normally involves withdrawal of the Government in favour of market based operations. In fact, a sound PPP contract enables the Government to retain oversight and control over all critical elements of performance and outcomes while payments are typically made against services delivered. Through this mode of procurement, the government hopes to achieve a comparatively better outcome by combining the respective strengths of the public and the private sector in the form of a true partnership that would ensure a better outcome as compared to public or private schools.
- 3.3 PPP schools will not be private enterprises acting in pursuit of profits. Nor will their management have the freedom that a private school normally enjoys in matters such as admissions and fees. These schools will essentially be public schools run by non-profit private entities which are to be guided by the public objectives set by the Government and enshrined in an enforceable contract. The success of PPPs as an efficient mode of implementation of public projects has already been demonstrated in several sectors such as highways, airports, power, railways, etc.

# 4 Benefits of the PPP Approach

Some good reasons for adopting the PPP approach for scaling up capacity in school education are as follows:

- (i) The investments required may be too large compared to available budgetary resources, and private capital could, therefore, enable an accelerated roll out of schools;
- (ii) functional efficiency of private entities would enable early delivery of quality education;
- (iii) risk of project completion and delivery of agreed outputs would be transferred to the private entity;
- (iv) public funds would be expended only upon delivery of agreed outcomes; and
- (v) private sector efficiency in the context of a long-term agreement is expected to optimise on life-cycle costs and improve on quality of education, including the school infrastructure.

# 5 Proposed framework for PPP schools

The proposed framework for setting up and managing 2,500 schools under this scheme would be as follows:

# 5.1 Schools to adhere to CBSE guidelines

All schools under this scheme would have to adhere to CBSE norms, standards and guidelines relating to affiliation, character of management, syllabus, teacher/student ratio, infrastructure, land, class rooms, labs, library, extracurricular activities, playgrounds, quality of education, teacher qualifications, service conditions, teachers' salary, examination, reserve fund, accounting and audit, pedagogy etc. Since these schools are expected to be Model Schools, the Concession Agreement would provide for additional norms which would be in accordance with Good Education Practice. In exceptional circumstances, affiliation with National Boards other than CBSE may also be considered.

#### 5.2 Location of PPP schools

The location of the 2,500 model schools under PPP mode would be headquarters of the blocks which are not educationally backward. In Blocks where a model school has already been set up or is being planned under Corporate Social Responsibility (CSR), no model school shall be set up.

#### 5.3 Infrastructure, teaching and management by private entities

Selected private entities, such as trusts, societies and not-for-profit companies, will be responsible for providing infrastructure such as land, school buildings, etc. They will also be responsible for management of the schools including provision of teaching. This would not only include employing of school teachers and other staff but also provision of teaching aids, laboratories, library, equipment, playgrounds and other extra-curricular facilities. The maximum capacity of a school would not exceed 2,500 students. Irrespective of whatever Board the school is affiliated to, the Schools would have to adhere to all norms of CBSE or as provided in the concession agreement with reference to infrastructure, teaching resources and pedagogy.

#### 5.4 Capital costs to be borne by private entity

The estimated capital cost of setting up a senior secondary school would vary between Rs. 5 to 7 crore. It is envisaged that the entire capital expenditure for setting up schools under this scheme would be borne by private entities. This would imply sharing of the financial risk by the private entity which would receive payments from the Government only upon successful delivery of well-defined outcomes in the form of education services.

# 5.5 Land to be arranged by the private entity

The land required for establishing a school has to be procured by the private entities from the respective State Governments or local authorities at applicable rates. Alternatively, they may procure land and/or buildings from any other source either through purchase or by way of a long-term lease of at least 25 years. The Central Government will advise the respective State Governments from time to time to facilitate private entities in procuring land.

#### 5.6 Recurring/ academic support for 980 Government sponsored Students

In each school, Government support shall be provided to 140 Select Students in each class totaling up to a maximum of 980 Select Students for the School.

Within the Select Students, reservation for SC/ ST/ OBC students shall be as per reservations applicable in the respective States. Further, there will be reservation of 33 per cent for girls amongst the Select Students. In the event that the reservation for girls is increased in the 3,500 Government model schools, the same shall apply to the PPP Schools. There will also be reservation as per the respective State Government norms for the disabled children. There will be no reservation for any category under the management quota.

In case, the private entity is unable to get adequate number of sponsored students in a particular class, the ceiling per class may be relaxed within the overall limit of 980 Select Students. In addition, the school management may enroll students on payment of fee as appropriate. The total number of students in the school shall not exceed 2,500 in any case.

The recurring support in respect of the tuition fees etc. for Select Students shall be equal to the average expenditure per student incurred in the Kendriya Vidyalayas. This support shall be made available to each School on a half-yearly basis in advance against bank guarantee of equivalent amount valid for two years.

For the year 2013-14, the amount of recurring support would be fixed provisionally on the basis of the actual expenditure per student incurred in KendriyaVidyalayas during 2011-12 plus 10% thereof calculated at the rate of a provisional increase of 5% per annum. Before the close of 2014-15, the actual expenditure per student incurred in 2013-14 in the KendriyaVidyalayas shall be computed and if such expenditure is higher/lower than the amount paid on a provisional basis for the year 2013-14, the difference shall be adjusted in the forthcoming payments due to the school. This process will apply *mutatis mutandis* to all subsequent years.

In the first year, classes 6 to 8 will be started together and thereafter one higher class would be added in each year till the school reaches class 12. Government support would be provided in the first year for 420 students and this would increase every year by 140 till year  $5^{th}$  when it would reach 980

A school may be run in a temporary or partial accommodation for the first two years. From the third year onwards, the school should have the entire infrastructure specified in this Scheme.

# 5.7 Support for infrastructure

In addition to the aforesaid recurring support, a sum equal to 25% of the monthly recurring support for each Select Student shall be disbursed by way of infrastructure grant which may be used for defraying rentals, interest, debt repayment, etc. The annual infrastructure grant shall not exceed an amount equal to 10% of the capital investment in the school, as certified by an approved valuer.

# 5.8 Commencement of government support

Government support under this scheme shall commence only after CBSE affiliation is granted.

# 5.9 Medium of Instruction

The medium of instruction shall beas per the norms of the affiliating National Board.

# 5.10 Disbursements linked to output parameters

The disbursements under this scheme shall be linked to the provision of specified infrastructure and delivery of agreed output parameters

A pre-determined system of incentives and penalties will be specified based on the CBSE results and other key performance indicators such as provision of qualified teachers, student attendance and dropout rate. The output parameters would be developed in accordance with the best practices and specified clearly in the Concession Agreement.

# 5.11 Graduated support based on location

The support to be provided by the government will vary between locations having regard to the differential costs and the likely availability of students. The following graduated support would be provided for different locations:

- 75% in cities with population exceeding 30 lakh
- 90% in cities with population exceeding 10 lakh
- 100% in towns with population exceeding 1 lakh
- 105% in towns exceeding a population of 20,000
- 115% in a town/village with a population of less than 20,000
- 125% in NE, J&K & tribal areas

# 5.12 Duration of support

Government support would be provided to each school for a period of 10 years in accordance with the above scheme. From the eleventh year onwards, recurring academic support shall be provided only in respect of the Select

Students who were enrolled during or prior to the tenth year and shall continue until completion of their school education. The concession can be extended by mutual consent.

# 5.13 Concession agreement between Government and private entity

A concession agreement specifying the rights and obligations of both parties shall be signed between the Government and the selected private entity. This will enable the private entity to raise funds from the financial institutions for meeting its capital expenditure. The concession agreements will specify the over-arching principles while sufficient flexibility would be provided to private entities to manage their respective schools in conformity with the socio-economic requirements of different regions. Regular monitoring would be undertaken by the Government for enforcing the provisions of the concession agreement. The key features of the concession agreement would be:

- Application of CBSE guidelines
- Scheme of financial support
- Key Performance Indicators
- Incentives and penalties
- Monitoring & inspection mechanism; use of UID
- Use of premises for other educational purposes
- Suspension/ Termination for breach of Agreement

# 5.14 Tuition fee

The Select Students of Class IX to XII would be required to pay a tuition fee to inculcate a sense of participation. A discounted fee of Rs. 25 per month would be paid by Select Students belonging to the SC, ST, Girls and BPL categories while other Select Students would be required to pay Rs. 50 per month.

# 5.15 Market-based fee for other students

Apart from the 980 Select Students, the private entity may enrol up to1,470 additional students, who will be charged fees as determined by the private entity.

# 5.16 Use of school premises by the private entity

The private entity would be permitted to use the school premises for vocational education or any approved educational activity (excluding coaching) so as to enhance its revenue streams. This will help cross-subsidise the education for Select Students and also raise the level of other facilities. However, the additional revenue generated by the private entity will have to be ploughed back into the school and utilised towards improving the quality of the school.

#### 5.17 Selection criteria for applicants

Success of this scheme would depend on the private entities selected for this purpose. Private sponsors for these schools would, therefore, be selected on the basis of a transparent and fair selection process that would ensure selection of experienced, motivated and not-for-profit entities that are willing to invest their time and resources in creating centers of excellence in school education. The selection would be based on proven track record or capacity to establish and manage the schools rather than on financial bids based on the lowest offer. This is to ensure that the quality of education is not compromised by awarding schools to the lowest financial bidder, which in turn could result in lower salaries for teachers besides cost-cutting on other essential ingredients of good education.

An Applicant shall be eligible for pre-qualification in the following categories

(a) An Applicant having experience of running a CBSE-affiliated school(s) where at least 2 (two) batches have passed out from Class X would be eligible for award of 3 (three) Schools against each such existing school;

(b) An Applicant who is running a CBSE-affiliated school(s) from where 2 (two) batches of Class X have not yet passed out, shall be eligible for award of one School against each such existing school;

(c) An Applicant would qualify for award of 3 (three) Schools as against each such existing educational institution provided it has:

- (i) experience of at least 5 (five) years in running a higher secondary school, affiliated with any Board of Secondary Education or Intermediate Education, or
- (ii) experience of at least 5 (five) years in running an Educational Institution for imparting education in academic, technical or vocational courses.

Provided that a selected Bidder who is eligible for award of Schools pursuant to this sub-clause (c) shall undertake to make an interest-bearing deposit of Rs. 25 lakh for each School awarded, which would be released in three equal annual instalments after the School is commissioned;

For the purpose of this RFQ:

"Educational Institutions" shall mean institutions recognized/approved by a competent authority e.g., Central/State Government or other authorized agencies, e.g., Boards of Secondary Education, UGC, AICTE, MCI, ICAR, Directorate of Education or other duly empowered authority".

(d) any company (including a company incorporated under section 25 of the Companies Act ,1956) ,), trust or society having a Net-Worth of Rs. 25 crore or more during each of the preceding 2 (two) financial years and intending to set up a PPP school(s) through a not-for-profit entity specified in Clause 2.2.1(b) above would be eligible for allocation of one School for every Rs. 25 crore of Net-Worth, provided it shall undertake to make an interest-bearing deposit of Rs. 50 lakh for each School awarded to it, which would be released in three equal annual instalments after the School is commissioned in accordance with the Concession Agreement. In case more than 3 (three)Schools are allocated to a single entity under this clause 2.2.2(d), it shall make an interest-bearing deposit of Rs. 25 lakh for each additional school awarded to it.

A fair and transparent system of evaluation and scoring would be evolved and announced before inviting applications under this scheme. The selection criteria and its application would be fair and transparent so that in addition to ensuring that the best available applicants are selected, the selection process also enhances the confidence of the civil society in this initiative. For each location, the applicant who gets the highest score shall be the preferred applicant for award of the concession under this scheme.

An applicant under this scheme may either set up a new school or upgrade an existing institution to meet the requirements of this scheme.

# 5.18 Ceiling on number of schools

Not more than 50 schools may be awarded to a single entity. No single entity would be permitted to establish more than 20 schools in a single State or Union Territory. The above ceiling may be suitably relaxed if there are no eligible applicants for any particular location.

# 5.19 Selection criteria for students

Students will be selected by the management of the school on the basis of a transparent criteria and process which is consistent with the objectives of this Scheme. The selection criteria applicable to the 3,500 State sector Model Schools, to the extent applicable to this scheme, shall broadly apply to the PPP Schools.

# 5.20 Roll-out of the Scheme

The award of concessions would be spread over 3 years based on the following indicative roll-out plan:

Academic Year No. of schools

2012-13	500
2013-14	1,000
2014-15	1,000

#### 5.21 Bidding Process

Bidding would take place for each of the 2,500 Blocks. Each bidder would have to provide details about its financial and technical eligibility and capability. The criteria for evaluation and scoring of bids would be provided in the bidding documents (RFP).

#### 5.22 Managing Committee/ Governing Board

The Managing Committee/ Governing Board will have adequate representation from the parents, local bodies, State Government and educational experts. This would enhance the spirit of community participation and accountability that would help the schools to benefit from local support.

#### 5.23 Teachers' Salaries and Training

The quality of education is directly dependent on the quality of teachers and the level of their knowledge and skills. Since the selected private entity would, as a condition of this Scheme, be required to commit that it shall pay to its teaching staff salary and allowances no lower than the scales applicable for the KV schools, it should be possible for private entities to employ wellqualified teachers. Their training would be mandatory and shall conform to the standards recommended by CBSE for this purpose. It is expected that this arrangement would help create quality teachers who would set benchmarks in their respective spheres of influence.

#### 5.24 Name of Model Schools

All schools under this scheme shall be known, promoted, displayed and advertised by the name of Rashtriya Adarsh Vidyalaya. However, the respective private entities would be permitted to add two words of their choice (not being a name associated with any religion, caste or surname) before the words 'Rashtriya Adarsh Vidyalaya'. The said two words will be placed in the upper line and the name of the school i.e. Rashtriya Adarsh Vidyalaya will be prominently displayed in the main line. However, once the Government stops sponsoring the Select Students, the words "Rashtriya Adarsh Vidyalaya" will no longer be used by the private entity.

#### 5.25 Networking responsibility

These schools are expected to serve as models for all schools in their neighbourhood. They will be expected to act as mentors for other schools in the Block, organise bi-annual conferences for experience sharing and generally disseminate best practices. Their networking with other schools is expected to have a snowballing effect in improving school education in their respective Blocks.

#### 5.26 Special teaching for weaker students

In order to ensure the optimal level of learning achievement, intensive supportive teaching at the beginning of Class VI will be arranged so as to enable smooth transition of students to Class VI. Since many of the Select Students may have studied in the regional medium up to Class V, whereas the medium of instruction in the model school could be English, intensive coaching in English language will be made available by the school for students in classes VI and VII.

#### 5.27 Role of the State Government

The State Government may facilitate availability of land for the school. It may also consider providing financial support beyond the initial 10 year period of support from the Central Government. It would also be requested to provide uniforms and mid-day meals as applicable in Government schools. The State Government may also consider providing transport subsidy or free transport for students to come from different parts of the Blocks to the school and back.

# 6 Concession structure

A Design, Build, Finance and Operate (DBFO) model would be followed for this scheme. At the end of the concession period, the school would remain with the private entity. The concessionaire would be responsible for financing, constructing and maintaining the physical infrastructure of the school campus and for managing the school in accordance with laid down parameters and key performance indicators.

# 7 Enforcement and Inspections

- 7.1 The concession agreement would be enforced by regular inspections, audit and monitoring for quality assurance. There would be stiff penalties for violation of the agreement or for shortfalls in key performance indicators coupled with incentives for better performance.
- 7.2 Detailed arrangements would be spelt out in the concession agreement for regular reporting of outcomes which will be closely monitored by the Government through extensive use of IT and UID systems besides appropriate tests, inspections and surveys. Since payment to the private entity will be based on output parameters, a close monitoring thereof would be ensured. Detailed arrangements for regular monitoring would be spelt out and enforced as part of the concession framework.

#### 8 Nodal Agency

For initiating and implementing this scheme for 2,500 PPP schools, the Government will set up a nodal agency broadly on the lines of Kendriya Vidyalaya Sangthan, but with special emphasis on creating inter-disciplinary capacity to roll-out and monitor this scheme. The Nodal Agency called "Model School Organisatiion" or "Adarsh Vidyalaya Sangathan" may have regional offices, as necessary. Expert staff would be hired on contract basis, as necessary. The institutional approach and mindset required for rolling out and managing PPPs in school education is significantly different as compared to the direct control and administration of schools. Hence, a separate agency would be essential for ensuring the expected outcomes of this scheme.

# APPENDIX- V

Sl. No.	State/UT	District	Block
1	A & N ISLANDS	ANDAMANS	BRC- HUTBAY
2	A & N ISLANDS	ANDAMANS	BRC- SOUTH ANDAMAN
3	A & N ISLANDS	ANDAMANS	BRC- WIMBERLYGUNJ
4	A & N ISLANDS	MIDDLE AND NORTH ANDAMANS	BRC- DIGLIPUR
5	A & N ISLANDS	MIDDLE AND NORTH ANDAMANS	BRC- MAYABUNDER
6	A & N ISLANDS	MIDDLE AND NORTH ANDAMANS	BRC- RANGAT
7	A & N ISLANDS	NICOBARS	CAMPBELL BAY
8	A & N ISLANDS	NICOBARS	CAR NICOBAR
9	A & N ISLANDS	NICOBARS	NANCOWRY
10	ANDHRA PRADESH	ANANTAPUR	KOTHACHERUVU
11	ANDHRA PRADESH	ANANTAPUR	PENUKONDA
12	ANDHRA PRADESH	CHITTOOR	B.N.KANDRIGA
13	ANDHRA PRADESH	CHITTOOR	BANGARUPALEM
14	ANDHRA PRADESH	CHITTOOR	CHANDRAGIRI
15	ANDHRA PRADESH	CHITTOOR	CHINNAGOTTIGALLU
16	ANDHRA PRADESH	CHITTOOR	CHITTOOR
17	ANDHRA PRADESH	CHITTOOR	CHOWDEPALLE
18	ANDHRA PRADESH	CHITTOOR	GANGADHARANELLORE
19	ANDHRA PRADESH	CHITTOOR	GUDIPALA
20	ANDHRA PRADESH	CHITTOOR	GURRAMKONDA
21	ANDHRA PRADESH	CHITTOOR	IRALA
22	ANDHRA PRADESH	CHITTOOR	KALIKIRI
23	ANDHRA PRADESH	CHITTOOR	KARVETINAGAR
24	ANDHRA PRADESH	CHITTOOR	MADANAPALLE
25	ANDHRA PRADESH	CHITTOOR	NAGALAPURAM
26	ANDHRA PRADESH	CHITTOOR	NAGARI
27	ANDHRA PRADESH	CHITTOOR	NARAYANAVARAM
28	ANDHRA PRADESH	CHITTOOR	NINDRA
29	ANDHRA PRADESH	CHITTOOR	PAKALA
30	ANDHRA PRADESH	CHITTOOR	PALAMANER
31	ANDHRA PRADESH	CHITTOOR	PALASAMUDRAM
32	ANDHRA PRADESH	CHITTOOR	PEDDAPANJANI
33	ANDHRA PRADESH	CHITTOOR	PENUMUR
34	ANDHRA PRADESH	CHITTOOR	PILER
35	ANDHRA PRADESH	CHITTOOR	PITCHATUR
36	ANDHRA PRADESH	CHITTOOR	PULICHERLA
37	ANDHRA PRADESH	CHITTOOR	PUTHALAPATTU
38	ANDHRA PRADESH	CHITTOOR	PUTTUR
39	ANDHRA PRADESH	CHITTOOR	R.C.PURAM
40	ANDHRA PRADESH	CHITTOOR	RENIGUNTA
41	ANDHRA PRADESH	CHITTOOR	S.R.PURAM
42	ANDHRA PRADESH	CHITTOOR	SATHYAVEDU

# List of 3203 Blocks identified for Rashtriya Adarsh Vidyalayas

43	ANDHRA PRADESH	CHITTOOR	SODAM
44	ANDHRA PRADESH	CHITTOOR	SOMALA
45	ANDHRA PRADESH	CHITTOOR	SRIKALAHASTI
46	ANDHRA PRADESH	CHITTOOR	THAVANAMPALLE
47	ANDHRA PRADESH	CHITTOOR	THOTTAMBEDU
48	ANDHRA PRADESH	CHITTOOR	TIRUPATHI(R)
49	ANDHRA PRADESH	CHITTOOR	TIRUPATHI(U)
50	ANDHRA PRADESH	CHITTOOR	V.KOTA
51	ANDHRA PRADESH	CHITTOOR	VADAMALAPET
52	ANDHRA PRADESH	CHITTOOR	VARADAIAHPALEM
53	ANDHRA PRADESH	CHITTOOR	VAYALPAD
54	ANDHRA PRADESH	CHITTOOR	VEDURUKUPPAM
55	ANDHRA PRADESH	CHITTOOR	VIJAYAPURAM
56	ANDHRA PRADESH	CHITTOOR	YADAMARI
57	ANDHRA PRADESH	CHITTOOR	YERPEDU
58	ANDHRA PRADESH	CUDDAPAH	BADVEL
59	ANDHRA PRADESH	CUDDAPAH	C.K.DINNE
60	ANDHRA PRADESH	CUDDAPAH	CHENNUR
61	ANDHRA PRADESH	CUDDAPAH	CUDDAPAH
62	ANDHRA PRADESH	CUDDAPAH	GOPAVARAM
63	ANDHRA PRADESH	CUDDAPAH	KAMALAPURAM
64	ANDHRA PRADESH	CUDDAPAH	KODUR
65	ANDHRA PRADESH	CUDDAPAH	KONDAPURAM
66	ANDHRA PRADESH	CUDDAPAH	LINGALA
67	ANDHRA PRADESH	CUDDAPAH	MUDDANUR
68	ANDHRA PRADESH	CUDDAPAH	MYLAVARAM
69	ANDHRA PRADESH	CUDDAPAH	NANDALUR
70	ANDHRA PRADESH	CUDDAPAH	PRODDUTUR
71	ANDHRA PRADESH	CUDDAPAH	PULIVENDLA
72	ANDHRA PRADESH	CUDDAPAH	RAJAMPET
73	ANDHRA PRADESH	CUDDAPAH	RAJUPALEM
74	ANDHRA PRADESH	CUDDAPAH	SIMADHIPURAM
75	ANDHRA PRADESH	CUDDAPAH	SIDHOUT
76	ANDHRA PRADESH	CUDDAPAH	THONDUR
77	ANDHRA PRADESH	CUDDAPAH	V.N.PALLE
78	ANDHRA PRADESH	CUDDAPAH	VONTIMITTA
79	ANDHRA PRADESH	СИДДАРАН	YERRAGUNTLA
80	ANDHRA PRADESH	EAST GODAVARI	ADDATEEGALA
81	ANDHRA PRADESH	EAST GODAVARI	AINAVILLI
82	ANDHRA PRADESH	EAST GODAVARI	ALAMURU
83	ANDHRA PRADESH	EAST GODAVARI	ALLAVARAM
84	ANDHRA PRADESH	EAST GODAVARI	AMALAPURAM
85	ANDHRA PRADESH	EAST GODAVARI	AMBAJIPETA
86	ANDHRA PRADESH	EAST GODAVARI	ANAPARTHI
87	ANDHRA PRADESH	EAST GODAVARI	ATREYAPURAM
88	ANDHRA PRADESH	EAST GODAVARI	BICCAVOLU
89	ANDHRA PRADESH	EAST GODAVARI	DEVIPATNAM
90	ANDHRA PRADESH	EAST GODAVARI	GANDEPALLI
91	ANDHRA PRADESH	EAST GODAVARI	GOKAVARAM
92	ANDHRA PRADESH	EAST GODAVARI	GOLLAPROLU
93	ANDHRA PRADESH	EAST GODAVARI	I.POLAVARAM
94	ANDHRA PRADESH	EAST GODAVARI	JAGGAMPETA
95	ANDHRA PRADESH	EAST GODAVARI	K.GANGAVARAM

96	ANDHRA PRADESH	EAST GODAVARI	KADIAM
97	ANDHRA PRADESH	EAST GODAVARI	KAJULURU
98	ANDHRA PRADESH	EAST GODAVARI	KAKINDA(RURAL)
99	ANDHRA PRADESH	EAST GODAVARI	KAKINDA(URBAN)
100	ANDHRA PRADESH	EAST GODAVARI	KAPILESWARAPURAM
100	ANDHRA PRADESH	EAST GODAVARI	KARAPA
101	ANDHRA PRADESH	EAST GODAVARI	KATRENIKONA
102	ANDHRA PRADESH	EAST GODAVARI	KIRLAMPUDI
103	ANDHRA PRADESH	EAST GODAVARI	KORUKONDA
101	ANDHRA PRADESH	EAST GODAVARI	КОТНАРЕТА
105	ANDHRA PRADESH	EAST GODAVARI	MALIKIPURAM
100	ANDHRA PRADESH	EAST GODAVARI	MAMIDIKUDURU
107	ANDHRA PRADESH	EAST GODAVARI	MANDAPETA
100	ANDHRA PRADESH	EAST GODAVARI	MUMMIDIVARAM
110	ANDHRA PRADESH	EAST GODAVARI	P.GANNAVARAM
110	ANDHRA PRADESH	EAST GODAVARI	PEDAPUDI
111	ANDHRA PRADESH	EAST GODAVARI	PEDDAPURAM
112	ANDHRA PRADESH	EAST GODAVARI	PITHAPURAM
113	ANDHRA PRADESH	EAST GODAVARI	PRATIPADU
114	ANDHRA PRADESH	EAST GODAVARI	RAJAHMUNDRY(RURAL)
116	ANDHRA PRADESH	EAST GODAVARI	RAJAHMUNDRY(URBAN)
117	ANDHRA PRADESH	EAST GODAVARI	RAJANAGARAM
117	ANDHRA PRADESH	EAST GODAVARI	RAJAVOMMANGI
110	ANDHRA PRADESH	EAST GODAVARI	RAMACHANDRAPURAM
120	ANDHRA PRADESH	EAST GODAVARI	RANGAMPETA
120	ANDHRA PRADESH	EAST GODAVARI	RAVULAPALEM
121	ANDHRA PRADESH	EAST GODAVARI	RAYAVARM
123	ANDHRA PRADESH	EAST GODAVARI	RAZOLE
124	ANDHRA PRADESH	EAST GODAVARI	ROWTULAPUDI
125	ANDHRA PRADESH	EAST GODAVARI	SAKHINETIPALLI
126	ANDHRA PRADESH	EAST GODAVARI	SAMALKOTA
127	ANDHRA PRADESH	EAST GODAVARI	SEETHANAGARAM
128	ANDHRA PRADESH	EAST GODAVARI	THALLAREVU
129	ANDHRA PRADESH	EAST GODAVARI	U.KOTHAPALLI
130	ANDHRA PRADESH	EAST GODAVARI	UPPALAGUPTAM
131	ANDHRA PRADESH	EAST GODAVARI	YELESWARAM
132	ANDHRA PRADESH	GUNTUR	AMARAVATHI
133	ANDHRA PRADESH	GUNTUR	AMRUTHALURU
134	ANDHRA PRADESH	GUNTUR	BAPATLA
135	ANDHRA PRADESH	GUNTUR	BATTIPROLU
136	ANDHRA PRADESH	GUNTUR	CHEBROLU
137	ANDHRA PRADESH	GUNTUR	CHERUKUPALLE
138	ANDHRA PRADESH	GUNTUR	CHILAKALURIPETA
139	ANDHRA PRADESH	GUNTUR	DUGGIRALA
140	ANDHRA PRADESH	GUNTUR	EDLAPADU
141	ANDHRA PRADESH	GUNTUR	GUNTUR
142	ANDHRA PRADESH	GUNTUR	KAKUMANU
143	ANDHRA PRADESH	GUNTUR	KARLAPALEM
144	ANDHRA PRADESH	GUNTUR	KOLLIPARA
145	ANDHRA PRADESH	GUNTUR	KOLLURU
146	ANDHRA PRADESH	GUNTUR	MANGALAGIRI
147	ANDHRA PRADESH	GUNTUR	MEDIKONDURU
148	ANDHRA PRADESH	GUNTUR	NAGARAM 60

149	ANDHRA PRADESH	GUNTUR	NIZAMPATNAM
150	ANDHRA PRADESH	GUNTUR	PEDAKAKANI
151	ANDHRA PRADESH	GUNTUR	PEDAKURAPADU
152	ANDHRA PRADESH	GUNTUR	PEDANANDIPADU
153	ANDHRA PRADESH	GUNTUR	PHIRANGIPURAM
155	ANDHRA PRADESH	GUNTUR	PITTALAVANIPALEM
155	ANDHRA PRADESH	GUNTUR	PONNURU
155	ANDHRA PRADESH	GUNTUR	PRATHIPADU
150	ANDHRA PRADESH	GUNTUR	REPALLE
158	ANDHRA PRADESH	GUNTUR	SAVALYAPURAM
159	ANDHRA PRADESH	GUNTUR	TADIKONDA
160	ANDHRA PRADESH	GUNTUR	TENALI
161	ANDHRA PRADESH	GUNTUR	THADEPALLE
161	ANDHRA PRADESH	GUNTUR	THULLUR
163	ANDHRA PRADESH	GUNTUR	TSUNDUR
164	ANDHRA PRADESH	GUNTUR	VATTICHERUKURU
-	ANDHRA PRADESH		
165 166	ANDHRA PRADESH	GUNTUR HYDERABAD	VEMURU AMBERPET
166	ANDHRA PRADESH	HYDERABAD	AMBERPET
168	ANDHRA PRADESH ANDHRA PRADESH	HYDERABAD	ASIFNAGAR
169		HYDERABAD	BAHADURPURA
170	ANDHRA PRADESH	HYDERABAD	BANDLAGUDA
171	ANDHRA PRADESH	HYDERABAD	CHARMINAR
172	ANDHRA PRADESH	HYDERABAD	GOLCONDA
173	ANDHRA PRADESH	HYDERABAD	HIMAYATNAGAR
174	ANDHRA PRADESH	HYDERABAD	KHAIRATABAD
175	ANDHRA PRADESH	HYDERABAD	MAREDPALLY
176	ANDHRA PRADESH	HYDERABAD	MUSHEERABAD
177	ANDHRA PRADESH	HYDERABAD	NAMPALLY
178	ANDHRA PRADESH	HYDERABAD	SAIDABAD
179	ANDHRA PRADESH	HYDERABAD	SECUNDERABAD
180	ANDHRA PRADESH	HYDERABAD	SHAIKPET
181	ANDHRA PRADESH	HYDERABAD	TIRUMALAGIRI
182	ANDHRA PRADESH	KARIMNAGAR	ELGAID
183	ANDHRA PRADESH	KARIMNAGAR	HUZURABAD
184	ANDHRA PRADESH	KARIMNAGAR	KAMANPOOR
185	ANDHRA PRADESH	KARIMNAGAR	MANTHANI
186	ANDHRA PRADESH	KARIMNAGAR	METPALLI
187	ANDHRA PRADESH	KARIMNAGAR	PEDDAPALLI
188	ANDHRA PRADESH	KHAMMAM	ASWAPURAM
189	ANDHRA PRADESH	KHAMMAM	ASWARAOPETA
190	ANDHRA PRADESH	KHAMMAM	DAMMAPETA
191	ANDHRA PRADESH	KHAMMAM	KALLUR
192	ANDHRA PRADESH	KHAMMAM	KOTHAGUDEM
193	ANDHRA PRADESH	КНАММАМ	MADHIRA
194	ANDHRA PRADESH	KHAMMAM	MANUGURU
195	ANDHRA PRADESH	KHAMMAM	NELAKONDAPALLY
196	ANDHRA PRADESH	KHAMMAM	SATHUPALLY
197	ANDHRA PRADESH	KHAMMAM	THALLADA
198	ANDHRA PRADESH	KHAMMAM	VEMSOOR
199	ANDHRA PRADESH	KHAMMAM	WYRA
200	ANDHRA PRADESH	KHAMMAM	YELLANDU
201	ANDHRA PRADESH	KRISHNA	AGIRIPALLI 61

202	ANDHRA PRADESH	KRISHNA	AVANIGADDA
202	ANDHRA PRADESH	KRISHNA	BANTUMILLI
203	ANDHRA PRADESH	KRISHNA	BAPULAPADU
205	ANDHRA PRADESH	KRISHNA	CHALLAPALLI
205	ANDHRA PRADESH	KRISHNA	CHANDARLAPADU
207	ANDHRA PRADESH	KRISHNA	CHATRAI
208	ANDHRA PRADESH	KRISHNA	G.KONDURU
200	ANDHRA PRADESH	KRISHNA	GANNAVARAM
20)	ANDHRA PRADESH	KRISHNA	GHANTASALA
210	ANDHRA PRADESH	KRISHNA	GUDIVADA
211	ANDHRA PRADESH	KRISHNA	GUDLAVALLERU
212	ANDHRA PRADESH	KRISHNA	GUDURU
213	ANDHRA PRADESH	KRISHNA	IBRAHIMPATNAM
215	ANDHRA PRADESH	KRISHNA	JAGGAYYAPET
215	ANDHRA PRADESH	KRISHNA	KAIKALURU
210	ANDHRA PRADESH	KRISHNA	KALIDINDI
217	ANDHRA PRADESH	KRISHNA	KANCHIKACHERLA
218	ANDHRA PRADESH	KRISHNA	KANKIPADU
219	ANDHRA PRADESH	KRISHNA	KODURU
220	ANDHRA PRADESH	KRISHNA	KRUTHIVENNU
222	ANDHRA PRADESH	KRISHNA	MACHILIPATNAM
223	ANDHRA PRADESH	KRISHNA	MANDAVALLI
223	ANDHRA PRADESH	KRISHNA	MOPIDEVI
225	ANDHRA PRADESH	KRISHNA	MOVVA
226	ANDHRA PRADESH	KRISHNA	MUDIENEPALLI
227	ANDHRA PRADESH	KRISHNA	MUSUNURU
228	ANDHRA PRADESH	KRISHNA	MYLAVARAM
229	ANDHRA PRADESH	KRISHNA	NAGAYALANKA
230	ANDHRA PRADESH	KRISHNA	NANDIGAMA
231	ANDHRA PRADESH	KRISHNA	NANDIVADA
232	ANDHRA PRADESH	KRISHNA	NUZVID
233	ANDHRA PRADESH	KRISHNA	PAMARRU
234	ANDHRA PRADESH	KRISHNA	PAMIDIMUKKALA
235	ANDHRA PRADESH	KRISHNA	PEDANA
236	ANDHRA PRADESH	KRISHNA	PEDAPARAPUDI
237	ANDHRA PRADESH	KRISHNA	PENAMALURU
238	ANDHRA PRADESH	KRISHNA	PENUGANCHIPROLU
239	ANDHRA PRADESH	KRISHNA	THOTLAVALLURU
240	ANDHRA PRADESH	KRISHNA	TIRUVURU
241	ANDHRA PRADESH	KRISHNA	UNGUTURU
242	ANDHRA PRADESH	KRISHNA	VATSAVAI
243	ANDHRA PRADESH	KRISHNA	VERULLAPADU
244	ANDHRA PRADESH	KRISHNA	VIJAYAWADA RURAL
245	ANDHRA PRADESH	KRISHNA	VIJAYAWADA URBAN
246	ANDHRA PRADESH	KRISHNA	VISSANNAPET
247	ANDHRA PRADESH	KRISHNA	VUYYURU
248	ANDHRA PRADESH	KURNOOL	ATMAKUR
249	ANDHRA PRADESH	KURNOOL	NANDIKOTKUR
250	ANDHRA PRADESH	KURNOOL	SRISAILAM
251	ANDHRA PRADESH	MEDAK	KOHEER
252	ANDHRA PRADESH	MEDAK	PATANCHERU
253	ANDHRA PRADESH	MEDAK	THOGUTA
254	ANDHRA PRADESH	NALGONDA	ALAIR 62

255	ANDHRA PRADESH	NALGONDA	BIBINAGAR
255	ANDHRA PRADESH	NALGONDA	CHILUKUR
250	ANDHRA PRADESH	NALGONDA	CHITYALA
258	ANDHRA PRADESH	NALGONDA	HUZUR NAGAR
250	ANDHRA PRADESH	NALGONDA	KODAD
260	ANDHRA PRADESH	NALGONDA	NAKREKAL
260	ANDHRA PRADESH	NALGONDA	NALGONDA
261	ANDHRA PRADESH	NALGONDA	NARKATPALLE
262	ANDHRA PRADESH	NALGONDA	NEREDUCHERLA
264	ANDHRA PRADESH	NALGONDA	RAJAPET
265	ANDHRA PRADESH	NALGONDA	RAMANNAPET
265	ANDHRA PRADESH	NALGONDA	YADAGIRIGUTTA
267	ANDHRA PRADESH	NELLORE	ALLUR
268	ANDHRA PRADESH	NELLORE	ANANTHASAGARAM
269	ANDHRA PRADESH	NELLORE	ATHMAKUR
270	ANDHRA PRADESH	NELLORE	BALAYAPALLI
270	ANDHRA PRADESH	NELLORE	BOGOLE
271 272	ANDHRA PRADESH	NELLORE	BUCHIREDDIPALEM
272	ANDHRA PRADESH	NELLORE	CHEJERLA
273	ANDHRA PRADESH	NELLORE	CHILLAKUR
274			CHITTAMUR
275	ANDHRA PRADESH ANDHRA PRADESH	NELLORE	
		NELLORE	DAGADARTHI
277	ANDHRA PRADESH	NELLORE	DAKKILI
278	ANDHRA PRADESH	NELLORE	GUDUR
279	ANDHRA PRADESH	NELLORE	INDUKURPET
280	ANDHRA PRADESH	NELLORE	JALADANKI
281	ANDHRA PRADESH	NELLORE	KALUVOYA
282	ANDHRA PRADESH	NELLORE	KODAVALUR
283	ANDHRA PRADESH	NELLORE	KOTA
284	ANDHRA PRADESH	NELLORE	KOVUR
285	ANDHRA PRADESH	NELLORE	MANUBOLU
286	ANDHRA PRADESH	NELLORE	MUTHUKUR
287	ANDHRA PRADESH	NELLORE	NAIDUPET
288	ANDHRA PRADESH	NELLORE	NELLORE
289	ANDHRA PRADESH	NELLORE	OZILI
290	ANDHRA PRADESH	NELLORE	PELLAKUR
291	ANDHRA PRADESH	NELLORE	PODALAKUR
292	ANDHRA PRADESH	NELLORE	RAPUR
293	ANDHRA PRADESH	NELLORE	SAIDAPURAM
294	ANDHRA PRADESH	NELLORE	SANGAM
295	ANDHRA PRADESH	NELLORE	SULLURPET THOTADALLICUDUD
296	ANDHRA PRADESH	NELLORE	THOTAPALLIGUDUR
297	ANDHRA PRADESH	NELLORE	UDAYAGIRI
298	ANDHRA PRADESH	NELLORE	VAKADU
299	ANDHRA PRADESH	NELLORE	VARIKUNTAPADU
300	ANDHRA PRADESH	NELLORE	VENKATACHALAM
301	ANDHRA PRADESH	NELLORE	VIDAVALUR
302	ANDHRA PRADESH	NELLORE DRAKASAM	VINJAMUR
303	ANDHRA PRADESH	PRAKASAM	ADDANKI
304	ANDHRA PRADESH	PRAKASAM	CHIMAKURTHI
305	ANDHRA PRADESH	PRAKASAM	CUMBUM
306	ANDHRA PRADESH	PRAKASAM	GIDDALURU
307	ANDHRA PRADESH	PRAKASAM	INKOLLU 63

308	ANDHRA PRADESH	PRAKASAM	J.PANGULURU
309	ANDHRA PRADESH	PRAKASAM	KARAMCHEDU
310	ANDHRA PRADESH	PRAKASAM	KONDAPI
311	ANDHRA PRADESH	PRAKASAM	KORISAPADU
312	ANDHRA PRADESH	PRAKASAM	MADDIPADU
313	ANDHRA PRADESH	PRAKASAM	MARTURU
313	ANDHRA PRADESH	PRAKASAM	NAGULUPPALAPADU
314	ANDHRA PRADESH	PRAKASAM	ONGOLE
315	ANDHRA PRADESH	PRAKASAM	PARCHURU
310	ANDHRA PRADESH	PRAKASAM	SANTHANUTHALAPADU
317	ANDHRA PRADESH	PRAKASAM	SINGARAYAKONDA
318	ANDHRA PRADESH	PRAKASAM	TANGUTUR
319	ANDHRA PRADESH	PRAKASAM	VETAPALEM
320	ANDHRA PRADESH	PRAKASAM PRAKASAM	YEDDANAPUDI
322	ANDHRA PRADESH	RANGAREDDI	BALANAGAR
323	ANDHRA PRADESH	RANGAREDDI	BOLARAM
324	ANDHRA PRADESH	RANGAREDDI	GHATKESAR
325	ANDHRA PRADESH	RANGAREDDI	HAYATHNAGAR
326	ANDHRA PRADESH	RANGAREDDI	KEESARA
327	ANDHRA PRADESH	RANGAREDDI	MEDCHAL
328	ANDHRA PRADESH	RANGAREDDI	QUTHBULLAPUR
329	ANDHRA PRADESH	RANGAREDDI	RAJENDRANAGAR
330	ANDHRA PRADESH	RANGAREDDI	SAROORNAGAR
331	ANDHRA PRADESH	RANGAREDDI	SHAMIRPET
332	ANDHRA PRADESH	RANGAREDDI	SHRELINGAMPALLI
333	ANDHRA PRADESH	RANGAREDDI	UPPAL
334	ANDHRA PRADESH	SRIKAKULAM	NARASANNAPETA
335	ANDHRA PRADESH	SRIKAKULAM	PALAKONDA
336	ANDHRA PRADESH	SRIKAKULAM	PALASA
337	ANDHRA PRADESH	SRIKAKULAM	SARUBUJJILI
338	ANDHRA PRADESH	SRIKAKULAM	TEKKALI
339	ANDHRA PRADESH	SRIKAKULAM	VAJRAPUKOTTURU
340	ANDHRA PRADESH	VISAKHAPATNAM	ANAKAPALLI
341	ANDHRA PRADESH	VISAKHAPATNAM	CHINAGADILA
342	ANDHRA PRADESH	VISAKHAPATNAM	GAJUWAKA
343	ANDHRA PRADESH	VISAKHAPATNAM	PARAWADA
344	ANDHRA PRADESH	VISAKHAPATNAM	PAYAKARAOPETA
345	ANDHRA PRADESH	VISAKHAPATNAM	PEDAGANTAYADA
346	ANDHRA PRADESH	VISAKHAPATNAM	PENDURTHY
347	ANDHRA PRADESH	VISAKHAPATNAM	VISAKHAPATNAM URBAN
348	ANDHRA PRADESH	VISAKHAPATNAM	YELLAMANCHILI
349	ANDHRA PRADESH	VIZIANAGARAM	SRUNGAVARAPUKOTA
350	ANDHRA PRADESH	WARANGAL	HANAMKONDA
351	ANDHRA PRADESH	WARANGAL	MANGAPET
352	ANDHRA PRADESH	WARANGAL	NARSAMPET
353	ANDHRA PRADESH	WARANGAL	PARKAL
354	ANDHRA PRADESH	WARANGAL	WARANGAL
355	ANDHRA PRADESH	WEST GODAVARI	ACHANTA
356	ANDHRA PRADESH	WEST GODAVARI	AKIVEEDU
357	ANDHRA PRADESH	WEST GODAVARI	ATTILI
358	ANDHRA PRADESH	WEST GODAVARI	BHIMADOLE
359	ANDHRA PRADESH	WEST GODAVARI	BHIMAVARAM
360	ANDHRA PRADESH	WEST GODAVARI	BUTTAIGUDEM
	ducation rfa/06 07 2012	•	64

361	ANDHRA PRADESH	WEST GODAVARI	CHAGALLU
362	ANDHRA PRADESH	WEST GODAVARI	CHINTALAPUDI
363	ANDHRA PRADESH	WEST GODAVARI	DENDULURU
364	ANDHRA PRADESH	WEST GODAVARI	DEVARAPALLI
365	ANDHRA PRADESH	WEST GODAVARI	DWARAKATIRUMALA
366	ANDHRA PRADESH	WEST GODAVARI	ELAMANCHILI
367	ANDHRA PRADESH	WEST GODAVARI	ELURU
368	ANDHRA PRADESH	WEST GODAVARI	GANAPAVARAM
369	ANDHRA PRADESH	WEST GODAVARI	GOPALAPURAM
370	ANDHRA PRADESH	WEST GODAVARI	IRAGAVARAM
370	ANDHRA PRADESH	WEST GODAVARI	JANGAREDDYGUDEM
371	ANDHRA PRADESH	WEST GODAVARI	JEELUGUMILLI
372	ANDHRA PRADESH	WEST GODAVARI	KALLA
373	ANDHRA PRADESH	WEST GODAVARI	KAMAVARAPUKOTA
374	ANDHRA PRADESH	WEST GODAVARI	KOVVUR
375	ANDHRA PRADESH	WEST GODAVARI	KOYVUK KOYYALAGUDEM
		WEST GODAVARI	LINGAPALEM
377	ANDHRA PRADESH		
378 379	ANDHRA PRADESH	WEST GODAVARI WEST GODAVARI	MOGALTUR
	ANDHRA PRADESH		NALLAJERLA
380	ANDHRA PRADESH	WEST GODAVARI	NARSAPURAM
381	ANDHRA PRADESH	WEST GODAVARI	NIDADAVOLE
382	ANDHRA PRADESH	WEST GODAVARI	NIDAMARRU
383	ANDHRA PRADESH	WEST GODAVARI	PALACOLE
384	ANDHRA PRADESH	WEST GODAVARI	PALAKODERU
385	ANDHRA PRADESH	WEST GODAVARI	PEDAPADU
386	ANDHRA PRADESH	WEST GODAVARI	PEDAVEGI
387	ANDHRA PRADESH	WEST GODAVARI	PENTAPADU
388	ANDHRA PRADESH	WEST GODAVARI	PENUGONDA
389	ANDHRA PRADESH	WEST GODAVARI	PENUMANTRA
390	ANDHRA PRADESH	WEST GODAVARI	PERAVALI
391	ANDHRA PRADESH	WEST GODAVARI	PODURU
392	ANDHRA PRADESH	WEST GODAVARI	POLAVARAM
393	ANDHRA PRADESH	WEST GODAVARI	T.NARSAPURAM
394	ANDHRA PRADESH	WEST GODAVARI	TADEPALLI GUDEM
395	ANDHRA PRADESH	WEST GODAVARI	TALLAPUDI
396	ANDHRA PRADESH	WEST GODAVARI	TANUKU
397	ANDHRA PRADESH	WEST GODAVARI	UNDI
398	ANDHRA PRADESH	WEST GODAVARI	UNDRAJAVARAM
399	ANDHRA PRADESH	WEST GODAVARI	UNGUTURU
400	ANDHRA PRADESH	WEST GODAVARI	VEERAVASARAM
401	ARUNACHAL PRADESH	ANJAW	CHAGLAGAM
402	ARUNACHAL PRADESH	CHANGLANG	NAMPONG-MANMAO
403	ARUNACHAL PRADESH	DIBANG VALLEY	ANELIH-ARZU
404	ARUNACHAL PRADESH	DIBANG VALLEY	ANINI-MIPI
405	ARUNACHAL PRADESH	DIBANG VALLEY	ITALIN-MALINEY
406	ARUNACHAL PRADESH	EAST KAMENG	BAMENG II
407	ARUNACHAL PRADESH	EAST KAMENG	BANA

408	ARUNACHAL PRADESH	EAST KAMENG	PIPU
409	ARUNACHAL PRADESH	EAST KAMENG	SEIJOSA
410	ARUNACHAL PRADESH	EAST SIANG	PANGIN-BOLENG
411	ARUNACHAL PRADESH	EAST SIANG	PASIGHAT
412	ARUNACHAL PRADESH	EAST SIANG	RUKSIN
413	ARUNACHAL PRADESH	KURUNG KUMEY	SANGRAM
414	ARUNACHAL PRADESH	KURUNG KUMEY	SARLI
415	ARUNACHAL PRADESH	LOHIT	LEKANG
416	ARUNACHAL PRADESH	LOHIT	TEZU-SUNPURA
417	ARUNACHAL PRADESH	LOWER SUBANSIRI	ZIRO - I C.D. BLOCK
418	ARUNACHAL PRADESH	LOWER SUBANSIRI	ZIRO - II C.D. BLOCK
419	ARUNACHAL PRADESH	PAPUM PARE	BALIJAN
420	ARUNACHAL PRADESH	PAPUM PARE	DOIMUKH-KIMIN
421	ARUNACHAL PRADESH	PAPUM PARE	MENGIO
422	ARUNACHAL PRADESH	TIRAP	KANUBARI
423	ARUNACHAL PRADESH	TIRAP	KHONSA
424	ARUNACHAL PRADESH	TIRAP	LAZU
425	ARUNACHAL PRADESH	UPPER SIANG	GEKU
426	ARUNACHAL PRADESH	UPPER SIANG	JENGGING
427	ARUNACHAL PRADESH	UPPER SIANG	YINGKIONG
428	ARUNACHAL PRADESH	UPPER SUBANSIRI	BARIRIJO
429	ARUNACHAL PRADESH	UPPER SUBANSIRI	PUCHIGEKO
430	ARUNACHAL PRADESH	WEST KAMENG	KALAKTANG
431	ARUNACHAL PRADESH	WEST KAMENG	NAFRA
432	ARUNACHAL PRADESH	WEST KAMENG	THRIZINO-BURAGAON
433	ARUNACHAL PRADESH	WEST SIANG	ALONG EAST
434	ARUNACHAL PRADESH	WEST SIANG	ALONG WEST
435	ARUNACHAL PRADESH	WEST SIANG	BASAR
436	ARUNACHAL PRADESH	WEST SIANG	KAYING-PAYUM
437	ARUNACHAL	WEST SIANG	LIKABALI-GENSI

	PRADESH		
438	ARUNACHAL PRADESH	WEST SIANG	LIROMOBA
439	ARUNACHAL PRADESH	WEST SIANG	RUMGONG
440	ASSAM	BARPETA	BAJALI
441	ASSAM	BARPETA	BHAWANIPUR
442	ASSAM	BARPETA	JALAH
443	ASSAM	BONGAIGAON	BOITAMARI
444	ASSAM	BONGAIGAON	BONGAIGAON
445	ASSAM	BONGAIGAON	MANIKPUR
446	ASSAM	BONGAIGAON	SRIJANGRAM
447	ASSAM	CACHAR	KATIGORAH
448	ASSAM	CACHAR	LAKHIPUR
449	ASSAM	CACHAR	NARSINGHPUR
450	ASSAM	CACHAR	RAJA BAZAR
451	ASSAM	CACHAR	SALCHAPRA
452	ASSAM	CACHAR	SILCHAR URBAN
453	ASSAM	CACHAR	SONAI
454	ASSAM	CACHAR	UDHARBOND
455	ASSAM	DARRANG	DALGAON
456	ASSAM	DARRANG	KALAIGAON
457	ASSAM	DARRANG	KHOIRABARI
458	ASSAM	DARRANG	SIPAJHAR
459	ASSAM	DARRANG	UDALGURI
460	ASSAM	DHEMAJI	BORDOLONI
461	ASSAM	DHEMAJI	DHEMAJI
462	ASSAM	DHUBRI	AGOMONI
463	ASSAM	DIBRUGARH	BARBARUAH
464	ASSAM	DIBRUGARH	JOYPUR
465	ASSAM	DIBRUGARH	KHOWANG
466	ASSAM	DIBRUGARH	TENGAKHAT
467	ASSAM	GOALPARA	BALIJANA
468	ASSAM	GOALPARA	DUDHNOI
469	ASSAM	GOLAGHAT	CENTRAL BLOCK
470	ASSAM	GOLAGHAT	EAST BLOCK
471	ASSAM	GOLAGHAT	NORTH BLOCK
472	ASSAM	GOLAGHAT	SOUTH BLOCK
473	ASSAM	GOLAGHAT	WEST BLOCK
474	ASSAM	HAILAKANDI	KATLICHERRA
475	ASSAM	HAILAKANDI	LALA
476	ASSAM	JORHAT	CENTRAL JORHAT
477	ASSAM	JORHAT	EAST JORHAT
478	ASSAM	JORHAT	JORHAT
479	ASSAM	JORHAT	MAJULI
480	ASSAM	JORHAT	NORTH WEST JORHAT
481	ASSAM	JORHAT	TITABOR
482	ASSAM	KAMRUP	ВОКО
483	ASSAM	KAMRUP	CHHAYGAON
484	ASSAM	KAMRUP	DIMORIA
485	ASSAM	KAMRUP	GUWAHATI
486	ASSAM	KAMRUP	НАЈО
487	ASSAM	KAMRUP	KAMALPUR

488	ASSAM	KAMRUP	KARARA
489	ASSAM	KAMRUP	RAMPUR
489	ASSAM	KAMRUP	RANGIA
491	ASSAM	KAMRUP	RANI
491	ASSAM	KARBIANGLONG	BOKAJAN
493	ASSAM	KARBI ANGLONG	HOWRAGHAT
494	ASSAM	KARIMGANJ	BADARPUR
495	ASSAM	KARIMGANJ	NORTH KARIMGANJ
496	ASSAM	KARIMGANJ	PATHER KANDI
497	ASSAM	KARIMGANJ	R K NAGAR
498	ASSAM	KARIMGANJ	SOUTH KARIMGANJ
499	ASSAM	KOKRAJHAR	DOTMA
500	ASSAM	LAKHIMPUR	BIHPURIA
501	ASSAM	LAKHIMPUR	BORDOLONI
502	ASSAM	LAKHIMPUR	DHAKUAKHANA
503	ASSAM	LAKHIMPUR	LAKHIMPUR
504	ASSAM	LAKHIMPUR	NARAYANPUR
505	ASSAM	LAKHIMPUR	NOWBOICHA
506	ASSAM	MARIGAON	BHURBANDHA
507	ASSAM	MARIGAON	KAPILI
508	ASSAM	MARIGAON	MAYONG
509	ASSAM	NAGAON	JUGIJAN
510	ASSAM	NAGAON	KAPILI
511	ASSAM	NAGAON	KATHIATOLI
512	ASSAM	NAGAON	KHAGARIJAN
513	ASSAM	NAGAON	LANKA
514	ASSAM	NAGAON	RUPAHI
515	ASSAM	NALBARI	BASKA
516	ASSAM	NALBARI	BORIGOG BANBHAG
517	ASSAM	NALBARI	PACHIM NALBARI
518	ASSAM	NALBARI	PUB NALBARI
519	ASSAM	NALBARI	TIHU BARAMA
520	ASSAM	NORTH CACHAR HILLS	HARANGAJAO
521	ASSAM	NORTH CACHAR HILLS	MAHUR
522	ASSAM	NORTH CACHAR HILLS	MAIBANG
523	ASSAM	SIBSAGAR	AMGURI
524	ASSAM	SIBSAGAR	DEMOW
525	ASSAM	SIBSAGAR	KHELUA
526	ASSAM	SIBSAGAR	NAZIRA
527	ASSAM	SIBSAGAR	SAPEKHATI
528	ASSAM	SIBSAGAR	SONARI
529	ASSAM	SONITPUR	BALIPARA
530	ASSAM	SONITPUR	BISWANATH
531	ASSAM	SONITPUR	CHAIDUAR
532	ASSAM	SONITPUR	GABHARU
533	ASSAM	SONITPUR	NADUAR
534	ASSAM	TINSUKIA	KAKAPATHAR
535	ASSAM	TINSUKIA	MARGHERITA
536	ASSAM	TINSUKIA	SADIYA
537	BIHAR	LAKHISARAI	CHANAN
538	BIHAR	MUNGER	JAMALPUR
539	BIHAR	PATNA	PHULWARISHARIF

540	BIHAR	ROHTAS	SANJHAULI
541	CHANDIGARH	CHANDIGARH	GGMSSS-8
542	CHANDIGARH	CHANDIGARH	GGSSS-20/B
543	CHANDIGARH	CHANDIGARH	GMSSS10
544	CHANDIGARH	CHANDIGARH	GMSSS-16
545	CHANDIGARH	CHANDIGARH	GMSSS-18
546	CHANDIGARH	CHANDIGARH	GMSSS-20
547	CHANDIGARH	CHANDIGARH	GMSSS-22/A
548	CHANDIGARH	CHANDIGARH	GMSSS-23
549	CHANDIGARH	CHANDIGARH	GMSSS-33
550	CHANDIGARH	CHANDIGARH	GMSSS-35
551	CHANDIGARH	CHANDIGARH	GMSSS-37/B
552	CHANDIGARH	CHANDIGARH	GMSSS-40
553	CHANDIGARH	CHANDIGARH	GMSSS-44
554	CHANDIGARH	CHANDIGARH	GMSSS-46
555	CHANDIGARH	CHANDIGARH	GMSSS-KARSAN
556	CHANDIGARH	CHANDIGARH	GMSSS-MCM
557	CHANDIGARH	CHANDIGARH	GSSS-27
558	CHANDIGARH	CHANDIGARH	GSSS-38(W)
559	CHANDIGARH	CHANDIGARH	GSSS-45
560	CHANDIGARH	CHANDIGARH	GSSS-MMT
561	CHHATTISGARH	DHAMTARI	DHAMTARI
562	CHHATTISGARH	DHAMTARI	KURUD
563	CHHATTISGARH	DHAMTARI	MAGARLOD
564	CHHATTISGARH	DHAMTARI	NAGRI
565	CHHATTISGARH	DURG	BALOD
566	CHHATTISGARH	DURG	BERLA
567	CHHATTISGARH	DURG	DAUNDI
568	CHHATTISGARH	DURG	DAUNDI LOHARA
569	CHHATTISGARH	DURG	DHAMDHA
570	CHHATTISGARH	DURG	DURG
571	CHHATTISGARH	DURG	GUNDERDEHI
572	CHHATTISGARH	DURG	GURUR
573	CHHATTISGARH	DURG	NAWAGARH
574	CHHATTISGARH	DURG	PATAN
575	CHHATTISGARH	DURG	SAJA
576	CHHATTISGARH	JANJGIR - CHAMPA	AKALTARA
577	CHHATTISGARH	JANJGIR - CHAMPA	BALODA
578	CHHATTISGARH	JANJGIR - CHAMPA	BAMHANI DIH
579	CHHATTISGARH	JANJGIR - CHAMPA	DABHARA
580	CHHATTISGARH	JANJGIR - CHAMPA	JAIJAIPUR
580	CHHATTISGARH	JANJGIR - CHAMPA	MALKHARODA
582	CHHATTISGARH	JANJGIR - CHAMPA	NAWAGARH
583	CHHATTISGARH	JANJGIR - CHAMPA	SAKTI
584	CHHATTISGARH	JASHPUR	DULDULA
585	CHHATTISGARH	JASHPUR	FARSABAHAR
585	CHHATTISGARH	JASHPUR	JASHPUR
587	CHHATTISGARH	JASHPUR	KUNKURI
588	CHHATTISGARH	JASHPUR	MANORA
589	CHHATTISGARH	KANKER	ANTAGARH
590	CHHATTISGARH	KANKER	BHANUPRATAPPUR
1	ducation rfa/06.07.2012		69

591	CHHATTISGARH	KANKER	CHARAMA
592	CHHATTISGARH	KANKER	DURGUKONDAL
593	CHHATTISGARH	KANKER	KANKER
594	CHHATTISGARH	KANKER	KOYALIBEDA
595	CHHATTISGARH	KANKER	NARHARPUR
596	CHHATTISGARH	KAWARDHA	SAHASPUR LOHARA
597	CHHATTISGARH	KORIYA	BAIKUNTHPUR
598	CHHATTISGARH	KORIYA	SONHAT
599	CHHATTISGARH	MAHASAMUND	BAGBAHRA
600	CHHATTISGARH	MAHASAMUND	BASNA
601	CHHATTISGARH	MAHASAMUND	MHASMUND
602	CHHATTISGARH	MAHASAMUND	PITHORA
603	CHHATTISGARH	MAHASAMUND	SRIYAPALLI
604	CHHATTISGARH	RAIGARH	BARAMKELA
605	CHHATTISGARH	RAIGARH	GHARGHODA
606	CHHATTISGARH	RAIGARH	KHARSIA
607	CHHATTISGARH	RAIGARH	PUSSORE
608	CHHATTISGARH	RAIGARH	RAIGARH
609	CHHATTISGARH	RAIGARH	SARANGARH
610	CHHATTISGARH	RAIGARH	TAMNAR
611	CHHATTISGARH	RAIPUR	ABHANPUR
612	CHHATTISGARH	RAIPUR	ARANG
613	CHHATTISGARH	RAIPUR	CHHURA
614	CHHATTISGARH	RAIPUR	DHARSHIWA
615	CHHATTISGARH	RAIPUR	FINGESWAR
616	CHHATTISGARH	RAIPUR	GRIYABAND
617	CHHATTISGARH	RAIPUR	PLARI
618	CHHATTISGARH	RAIPUR	SIMGA
619	CHHATTISGARH	RAIPUR	TILDA
620	CHHATTISGARH	RAJNANDGAON	CHOWKI
621	CHHATTISGARH	RAJNANDGAON	CHUIKHADAN
622	CHHATTISGARH	RAJNANDGAON	CHURIA
623	CHHATTISGARH	RAJNANDGAON	DONGARGAON
624	CHHATTISGARH	RAJNANDGAON	DONGARGARH
625	CHHATTISGARH	RAJNANDGAON	KHAIRAGARH
626	CHHATTISGARH	RAJNANDGAON	MANPUR
627	CHHATTISGARH	RAJNANDGAON	MOHLA
628	CHHATTISGARH	RAJNANDGAON	RAJNANDGAON
629	CHHATTISGARH	SURGUJA	BALRAMPUR
630	CHHATTISGARH	SURGUJA	PREMNAGAR
631	CHHATTISGARH	SURGUJA	RAMANUJNAGAR
632	CHHATTISGARH	SURGUJA	SHANKARGARH
633	DAMAN & DIU	DAMAN	DAMAN UT
634	DAMAN & DIU	DIU	DIU
635	DELHI	CENTRAL DELHI	DARIYA GANJ
636	DELHI	CENTRAL DELHI	RANI JHANSI ROAD
637	DELHI	EAST DELHI	ANAND VIHAR
638	DELHI	EAST DELHI	CHANDRA NAGAR
639	DELHI	EAST DELHI	PREET VIHAR

640	DELHI	NEW DELHI	MALCHA MARG
641	DELHI	NORTH DELHI	KESHAV PURAM
642	DELHI	NORTH DELHI	SHYAM NATH MARG
643	DELHI	NORTH EAST DELHI	DILSHAD GARDEN
644	DELHI	NORTH EAST DELHI	SHAHDRA
645	DELHI	NORTH EAST DELHI	YAMUNA VIHAR
646	DELHI	NORTH WEST DELHI	BADLI
647	DELHI	NORTH WEST DELHI	MANGOLPURI (KAMDHENU)
648	DELHI	NORTH WEST DELHI	PITAM PURA
649	DELHI	NORTH WEST DELHI	ROHINI
650	DELHI	NORTH WEST DELHI	SHALIMAR BAGH
651	DELHI	NORTH WEST DELHI	TILAK NAGAR
652	DELHI	SOUTH DELHI	BEGAM PIUR
653	DELHI	SOUTH DELHI	DEFENCE COLONY
654	DELHI	SOUTH DELHI	SHAHPUR JAT
655	DELHI	SOUTH WEST DELHI	MOTI BAGH
656	DELHI	SOUTH WEST DELHI	NAJAFGARH
657	DELHI	SOUTH WEST DELHI	SHAHBAD MOHAMMAD PUR
658	DELHI	SOUTH WEST DELHI	VASANT VIHAR
659	DELHI	WEST DELHI	OLD RAJENDRA NAGAR
660	DELHI	WEST DELHI	PEERA GARHI
661	DELHI	WEST DELHI	PUNJABI BAGH
662	DELHI	WEST DELHI	VIKAS PURI
663	GOA	NORTH GOA	BARDEZ
664	GOA	NORTH GOA	BICHOLIM
665	GOA	NORTH GOA	PERNEM
666	GOA	NORTH GOA	PONDA
667	GOA	NORTH GOA	SATTARI
668	GOA	NORTH GOA	TISWADI
669	GOA	SOUTH GOA	CANACONA
670	GOA	SOUTH GOA	MARMUGOA
671	GOA	SOUTH GOA	QUEPEM
672	GOA	SOUTH GOA	SALCETE
673	GOA	SOUTH GOA	SANGUEM

674	GUJARAT	AHMADABAD	BARVALA
675	GUJARAT	AHMADABAD	DASCROI
676	GUJARAT	AHMADABAD	DETROJ-RAMPURA
677	GUJARAT	AHMADABAD	DHANDHUKA
678	GUJARAT	AHMADABAD	DHOLKA
679	GUJARAT	AHMADABAD	MANDAL
680	GUJARAT	AMRELI	AMRELI
681	GUJARAT	AMRELI	BABARA
682	GUJARAT	AMRELI	BAGASARA
683	GUJARAT	AMRELI	DHARI
684	GUJARAT	AMRELI	КНАМВНА
685	GUJARAT	AMRELI	KUKAVAV
686	GUJARAT	AMRELI	LATHI
687	GUJARAT	AMRELI	
688	GUJARAT	AMRELI	SAVARKUNDLA
689	GUJARAT	ANAND	ANAND
690	GUJARAT	ANAND	ANKLAV
691	GUJARAT	ANAND	BORSAD
692	GUJARAT	ANAND	KHAMBHAT
693	GUJARAT	ANAND	PETLAD
694	GUJARAT	ANAND	SOJITRA
695	GUJARAT	ANAND	TARAPUR
696	GUJARAT	ANAND	UMRETH
697	GUJARAT	BANAS KANTHA	PALANPUR
698	GUJARAT	BANAS KANTHA	VADGAM
699	GUJARAT	BHARUCH	AMOD
700	GUJARAT	BHARUCH	ANKLESHWAR
701	GUJARAT	BHARUCH	BHARUCH
702	GUJARAT	BHARUCH	HANSOT
703	GUJARAT	BHARUCH	JAMBUSAR
704	GUJARAT	BHARUCH	VAGRA
705	GUJARAT	BHARUCH	VALIA
706	GUJARAT	BHARUCH	ZAGHADIA
707	GUJARAT	BHAVNAGAR	GADHADA
708	GUJARAT	BHAVNAGAR	GARIYADHAR
709	GUJARAT	BHAVNAGAR	SHIHOR
710	GUJARAT	BHAVNAGAR	UMRALA
711	GUJARAT	BHAVNAGAR	VALLBHIPUR
712	GUJARAT	GANDHINAGAR	DEHGAM
713	GUJARAT	GANDHINAGAR	GANDHINAGAR
714	GUJARAT	GANDHINAGAR	KALOL
715	GUJARAT	GANDHINAGAR	MANSA
716	GUJARAT	JAMNAGAR	BHANVAD
717	GUJARAT	JAMNAGAR	DHROL
717	GUJARAT	JAMNAGAR	JAMJODHPUR
719	GUJARAT	JAMNAGAR	JAMJODIN OK
719	GUJARAT	JAMNAGAR	JODIYA
720	GUJARAT	JAMNAGAR	KALAVAD
721	GUJARAT	JAMNAGAR	LALPUR
722	GUJARAT	JUNAGADH	BHESAN
723	GUJARAT	JUNAGADH	JUNAGADH
725	GUJARAT	JUNAGADH	KESHOD
726	GUJARAT ducation rfa/06.07.2012	JUNAGADH	KODINAR 72

727	GUJARAT	JUNAGADH	MALIA HATINA
728	GUJARAT	JUNAGADH	MANAVADAR
729	GUJARAT	JUNAGADH	MANGROL
730	GUJARAT	JUNAGADH	MENDARDA
731	GUJARAT	JUNAGADH	TALALA
732	GUJARAT	JUNAGADH	VANTHALI
733	GUJARAT	JUNAGADH	VISAVADAR
733	GUJARAT	КАСНСНН	MANDVI
735	GUJARAT	КАСНСНИ	MUNDRA
736	GUJARAT	КАСНСНИ	NAKHATRANA
730	GUJARAT	KHEDA	KAPADVANJ
738	GUJARAT	KHEDA	KATHLAL
738	GUJARAT	KHEDA	KHEDA
739			MAHEMDAVAD
	GUJARAT	KHEDA	
741	GUJARAT	KHEDA	MAHUDHA
742	GUJARAT	KHEDA	MATAR
743	GUJARAT	KHEDA	NADIAD
744	GUJARAT	KHEDA	THASARA
745	GUJARAT	KHEDA	VIRPUR
746	GUJARAT	MAHESANA	BECHARAJI
747	GUJARAT	MAHESANA	KADI
748	GUJARAT	MAHESANA	KHERALU
749	GUJARAT	MAHESANA	MEHSANA
750	GUJARAT	MAHESANA	UNJHA
751	GUJARAT	MAHESANA	VADNAGAR
752	GUJARAT	MAHESANA	VIJAPUR
753	GUJARAT	MAHESANA	VISNAGAR
754	GUJARAT	NARMADA	NANDOD
755	GUJARAT	NARMADA	TILAKVADA
756	GUJARAT	NAVSARI	CHIKHLI
757	GUJARAT	NAVSARI	GANDEVI
758	GUJARAT	NAVSARI	JALALPOR
759	GUJARAT	NAVSARI	NAVSARI
760	GUJARAT	NAVSARI	VANSDA
761	GUJARAT	PANCH MAHALS	KALOL
762	GUJARAT	PANCH MAHALS	LUNAWADA
763	GUJARAT	PATAN	CHANSMA
764	GUJARAT	PATAN	SIDDHPUR
765	GUJARAT	PATAN	URBAN-PATAN
766	GUJARAT	PORBANDAR	KUTIYANA
767	GUJARAT	PORBANDAR	PORBANDAR
768	GUJARAT	PORBANDAR	RANAVAV
769	GUJARAT	RAJKOT	DHORAJI
770	GUJARAT	RAJKOT	GONDAL
771	GUJARAT	RAJKOT	JAMKANDORNA
772	GUJARAT	RAJKOT	JETPUR
773	GUJARAT	RAJKOT	KOTDA SANGANI
774	GUJARAT	RAJKOT	LODHIKA
775	GUJARAT	RAJKOT	MORBI
776	GUJARAT	RAJKOT	PADDHARI
777	GUJARAT	RAJKOT	RAJKOT
778	GUJARAT	RAJKOT	TANKARA
779	GUJARAT	RAJKOT	UPLETA
	ducation rfa/06.07.2012		73

780	GUJARAT	SABAR KANTHA	BAYAD
781	GUJARAT	SABAR KANTHA	BHILODA
782	GUJARAT	SABAR KANTHA	DHANSURA
783	GUJARAT	SABAR KANTHA	HIMMATNAGAR
784	GUJARAT	SABAR KANTHA	IDAR
785	GUJARAT	SABAR KANTHA	MODASA
786	GUJARAT	SABAR KANTHA	PRANTIJ
787	GUJARAT	SABAR KANTHA	TALOD
788	GUJARAT	SABAR KANTHA	VADALI
789	GUJARAT	SABAR KANTHA	VIJAYNAGAR
790	GUJARAT	SURAT	BARDOLI
790	GUJARAT	SURAT	CHORYASI
792	GUJARAT	SURAT	KAMREJ
792	GUJARAT	SURAT	MAHUVA
793	GUJARAT	SURAT	MANDAVI
794	GUJARAT	SURAT	MANGROL
			OLPAD
796	GUJARAT	SURAT	
797	GUJARAT	SURAT	PALASANA
798	GUJARAT	SURAT	VALOD
799	GUJARAT	SURAT	VYARA
800	GUJARAT	SURENDRANAGAR	WADHVAN
801	GUJARAT	THE DANGS	AHWA
802	GUJARAT	VADODARA	DABHOI
803	GUJARAT	VADODARA	KARJAN
804	GUJARAT	VADODARA	PADRA
805	GUJARAT	VADODARA	SANKHEDA
806	GUJARAT	VADODARA	SAVLI
807	GUJARAT	VADODARA	SHINOR
808	GUJARAT	VADODARA	VADODARA
809	GUJARAT	VADODARA	VAGHODIYA
810	GUJARAT	VALSAD	PARDI
811	GUJARAT	VALSAD	UMBERGAON
812	GUJARAT	VALSAD	VALSAD
813	HARYANA	AMBALA	AMBALA 1
814	HARYANA	AMBALA	AMBALA 2
815	HARYANA	AMBALA	BARARA
816	HARYANA	AMBALA	NARAINGARH
817	HARYANA	AMBALA	SAHA
818	HARYANA	AMBALA	SHAHZADPUR
819	HARYANA	BHIWANI	BADHRA
820	HARYANA	BHIWANI	BAUND KALAN
821	HARYANA	BHIWANI	BAWANI KHERA
822	HARYANA	BHIWANI	BEHAL
823	HARYANA	BHIWANI	BHIWANI
824	HARYANA	BHIWANI	DADRI
825	HARYANA	BHIWANI	KAIRU
826	HARYANA	BHIWANI	LOHARU
827	HARYANA	FARIDABAD	BALLABGARH
828	HARYANA	FARIDABAD	FARIDABAD
829	HARYANA	FATEHABAD	JAKHAL
830	HARYANA	GURGAON	F.NAGAR
831	HARYANA	GURGAON	GURGAON
832	HARYANA	GURGAON	PATAUDI
	ducation rfa/06.07.2012		74

833	HARYANA	GURGAON	SOHNA
834	HARYANA	HISAR	ADAMPUR
835	HARYANA	HISAR	HANSI-II
836	HARYANA	HISAR	HISAR-I
837	HARYANA	JHAJJAR	BAHADURGARH
838	HARYANA	JHAJJAR	BERI
839	HARYANA	JHAJJAR	JHAJJAR
840	HARYANA	JHAJJAR	MATANHAIL
841	HARYANA	JHAJJAR	SALHAWAS
842	HARYANA	JIND	JIND
843	HARYANA	JIND	JULANA
844	HARYANA	JIND	PILLUKHERA
845	HARYANA	JIND	SAFIDON
846	HARYANA	KAITHAL	GUHLA
847	HARYANA	KAITHAL	PUNDRI
848	HARYANA	KAITHAL	SIWAN
849	HARYANA	KARNAL	ASSANDH
850	HARYANA	KARNAL	GHARAUNDA
851	HARYANA	KARNAL	INDRI
852	HARYANA	KARNAL	KARNAL
853	HARYANA	KARNAL	NILOKHERI
854	HARYANA	KARNAL	NISSING
855	HARYANA	KURUKSHETRA	BABAIN
856	HARYANA	KURUKSHETRA	LADWA
857	HARYANA	KURUKSHETRA	PEHOWA
858	HARYANA	KURUKSHETRA	SHAHBAD
859	HARYANA	KURUKSHETRA	THANESAR
860	HARYANA	MAHENDRAGARH	ATELI
861	HARYANA	MAHENDRAGARH	KANINA
862	HARYANA	MAHENDRAGARH	MOHINDERGARH
863	HARYANA	MAHENDRAGARH	NARNAUL
864	HARYANA	PANCHKULA	BARWALA
865	HARYANA	PANCHKULA	MORNI
866	HARYANA	PANCHKULA	PINJORE
867	HARYANA	PANCHKULA	RAIPUR RANI
868	HARYANA	PANIPAT	ISRANA
869	HARYANA	PANIPAT	MADLAUDA
870	HARYANA	PANIPAT	PANIPAT
871	HARYANA	PANIPAT	SAMALKHA
872	HARYANA	REWARI	BAWAL
873	HARYANA	REWARI	JATUSANA
874	HARYANA	REWARI	KHOL
875	HARYANA	REWARI	NAHAR
876	HARYANA	REWARI	REWARI
877	HARYANA	ROHTAK	KALANAUR
878	HARYANA	ROHTAK	LAKHAN MAJRA
879	HARYANA	ROHTAK	MEHAM
880	HARYANA	ROHTAK	ROHTAK
881	HARYANA	ROHTAK	SAMPLA
882	HARYANA	SIRSA	SIRSA
883	HARYANA	SONIPAT	GANAUR

884	HARYANA	SONIPAT	GOHANA
885	HARYANA	SONIPAT	KATHURA
886	HARYANA	SONIPAT	KHARKHODA
887	HARYANA	SONIPAT	MUNDLANA
888	HARYANA	SONIPAT	RAI
889	HARYANA	SONIPAT	SONEPAT
890	HARYANA	YAMUNANAGAR	BILASPUR
891	HARYANA	YAMUNANAGAR	CHHACHHRAULI
892	HARYANA	YAMUNANAGAR	JAGADHRI
893	HARYANA	YAMUNANAGAR	MUSTAFABAD
894	HARYANA	YAMUNANAGAR	RADAUR
895	HARYANA	YAMUNANAGAR	SADHAURA
896	HIMACHAL PRADESH	BILASPUR	GHUMARWIN-I
897	HIMACHAL PRADESH	BILASPUR (H.P.)	GHUMARWIN-II
898	HIMACHAL PRADESH	BILASPUR (H.P.)	JHANDUTTA
899	HIMACHAL PRADESH	BILASPUR (H.P.)	SADAR
900	HIMACHAL PRADESH	BILASPUR (H.P.)	SWARGHAT
901	HIMACHAL PRADESH	СНАМВА	BANIKHET
902	HIMACHAL PRADESH	СНАМВА	BHARMOUR
903	HIMACHAL PRADESH	СНАМВА	СНАМВА
904	HIMACHAL PRADESH	СНАМВА	CHOWARI
905	HIMACHAL PRADESH	СНАМВА	GAROLA
906	HIMACHAL PRADESH	СНАМВА	HARDASPURA
907	HIMACHAL PRADESH	CHAMBA	KIANI
908	HIMACHAL PRADESH	CHAMBA	SIHUNTA
909	HIMACHAL PRADESH	СНАМВА	SUNDLA
910	HIMACHAL PRADESH	HAMIRPUR (H.P.)	BHORANJ
911	HIMACHAL PRADESH	HAMIRPUR (H.P.)	BIJHARI
912	HIMACHAL PRADESH	HAMIRPUR (H.P.)	HAMIRPUR
913	HIMACHAL PRADESH	HAMIRPUR (H.P.)	NADAUN
914	HIMACHAL PRADESH	HAMIRPUR (H.P.)	SUJANPUR
915	HIMACHAL PRADESH	KANGRA	BAIJNATH
916	HIMACHAL PRADESH	KANGRA	BHAWARNA
917	HIMACHAL PRADESH	KANGRA	CHADHIAR
918	HIMACHAL PRADESH	KANGRA	DADASIBA
919	HIMACHAL PRADESH		DEHRA
920	HIMACHAL PRADESH	KANGRA	DHARAMSALA
921	HIMACHAL PRADESH	KANGRA	FATEHPUR
922	HIMACHAL PRADESH	KANGRA	INDORA
923	HIMACHAL PRADESH	KANGRA	JAWALI
924	HIMACHAL PRADESH	KANGRA	KANGRA
925	HIMACHAL PRADESH	KANGRA	LAMBAGAON
926	HIMACHAL PRADESH	KANGRA	NAGROTA BAGWAN
927	HIMACHAL PRADESH	KANGRA	NAGROTA SURIAN
928	HIMACHAL PRADESH	KANGRA	NURPUR
929 930	HIMACHAL PRADESH HIMACHAL PRADESH	KANGRA KANGRA	PALAMPUR PANCHRUKHI
930	HIMACHAL PRADESH	KANGRA	RAIT
931	HIMACHAL PRADESH	KANGRA	RAJA KA TALAB
932	HIMACHAL PRADESH	KANGRA	RAJA KA TALAD
933	HIMACHAL PRADESH	KINNAUR	KALPA
934	HIMACHAL PRADESH	KINNAUR	NICHAR
935	HIMACHAL PRADESH	KINNAUR	РООН
	hucation rfa/06.07.2012	MIMAUN	76

937	HIMACHAL PRADESH	KULLU	ANNI
938	HIMACHAL PRADESH	KULLU	BANJAR
939	HIMACHAL PRADESH	KULLU	KULLU-1
940	HIMACHAL PRADESH	KULLU	KULLU-2
941	HIMACHAL PRADESH	KULLU	NAGGAR
942	HIMACHAL PRADESH	KULLU	NIRMAND
943	HIMACHAL PRADESH	LAHUL & SPITI	KAZA
943	HIMACHAL PRADESH	LAHUL & SPITI	KAZA KEYLONG-I
944	HIMACHAL PRADESH	LAHUL & SPITI	KEYLONG-II
945	HIMACHAL PRADESH	LAHUL & SPITI	UDAIPUR
940	HIMACHAL PRADESH	MANDI	BALH
947	HIMACHAL PRADESH	MANDI	CHACHIOT-1
948	HIMACHAL PRADESH	MANDI	CHACHIOT-2
949	HIMACHAL PRADESH	MANDI	CHAUNTRA-1
-			
951 952	HIMACHAL PRADESH HIMACHAL PRADESH	MANDI	CHAUNTRA-2
		MANDI	DHARAMPUR-1 DHARAMPUR-2
953	HIMACHAL PRADESH	MANDI	
954	HIMACHAL PRADESH	MANDI	DRANG-1
955	HIMACHAL PRADESH	MANDI	DRANG-2
956	HIMACHAL PRADESH	MANDI	GOPALPUR-1
957	HIMACHAL PRADESH	MANDI	GOPALPUR-2
958	HIMACHAL PRADESH	MANDI	KARSOG-1
959	HIMACHAL PRADESH	MANDI	KARSOG-2
960	HIMACHAL PRADESH	MANDI	SADAR-1
961	HIMACHAL PRADESH	MANDI	SADAR-2
962	HIMACHAL PRADESH	MANDI	SAIGALOO
963	HIMACHAL PRADESH	MANDI	SERAJ-1
964	HIMACHAL PRADESH	MANDI	SERAJ-2
965	HIMACHAL PRADESH	MANDI	SUNDER NAGAR-1
966	HIMACHAL PRADESH	MANDI	SUNDER NAGAR-2
967	HIMACHAL PRADESH	SHIMLA	CHAUHARA
968	HIMACHAL PRADESH	SHIMLA	CHOPAL
969	HIMACHAL PRADESH	SHIMLA	DEHA
970	HIMACHAL PRADESH	SHIMLA	DODRAKAWAR
971	HIMACHAL PRADESH	SHIMLA	JUBBAL
972	HIMACHAL PRADESH		KASUMPATI
973	HIMACHAL PRADESH	SHIMLA	KOTKHAI
974	HIMACHAL PRADESH	SHIMLA	KUMARSAIN
975	HIMACHAL PRADESH	SHIMLA	KUPVI
976	HIMACHAL PRADESH	SHIMLA	MASHOBRA
977	HIMACHAL PRADESH	SHIMLA	MATIANA
978	HIMACHAL PRADESH	SHIMLA	NANKHARI
979	HIMACHAL PRADESH	SHIMLA	NERWA
980	HIMACHAL PRADESH	SHIMLA	RAMPUR
981	HIMACHAL PRADESH	SHIMLA	RAMPUR-II AT SARAHAN
982	HIMACHAL PRADESH	SHIMLA	ROHRU
983	HIMACHAL PRADESH	SHIMLA	SHIMLA-4
984	HIMACHAL PRADESH	SHIMLA	SUNI
985	HIMACHAL PRADESH	SHIMLA	THEOG
986	HIMACHAL PRADESH	SHIMLA	TIKKAR
987	HIMACHAL PRADESH	SIRMAUR	BAKRAS
988	HIMACHAL PRADESH	SIRMAUR	DADAHU
989	HIMACHAL PRADESH	SIRMAUR	NAHAN 77

990	HIMACHAL PRADESH	SIRMAUR	NOHRADHAR
990 991	HIMACHAL PRADESH	SIRMAUR	PAONTA SAHIB
992	HIMACHAL PRADESH	SIRMAUR	RAJGARH
993	HIMACHAL PRADESH	SIRMAUR	SARAHAN
994	HIMACHAL PRADESH	SIRMAUR	SATAUN
995	HIMACHAL PRADESH	SIRMAUR	SURLA
996	HIMACHAL PRADESH	SOLAN	ARKI
997	HIMACHAL PRADESH	SOLAN	DHRAMPUR
998	HIMACHAL PRADESH	SOLAN	DHUDAN
999	HIMACHAL PRADESH	SOLAN	KANDAGHAT
1000	HIMACHAL PRADESH	SOLAN	KUTHAR
1001	HIMACHAL PRADESH	SOLAN	NALAGARH
1002	HIMACHAL PRADESH	SOLAN	RAM SHAHAR
1003	HIMACHAL PRADESH	UNA	AMB
1004	HIMACHAL PRADESH	UNA	BANGANA
1005	HIMACHAL PRADESH	UNA	GAGRET - 2
1006	HIMACHAL PRADESH	UNA	GAGRET -1
1007	HIMACHAL PRADESH	UNA	UNA - 2
1008	HIMACHAL PRADESH	UNA	UNA -1
1009	JAMMU & KASHMIR	ANANTNAG	AISHMUQAM
1010	JAMMU & KASHMIR	ANANTNAG	ANANTNAG
1011	JAMMU & KASHMIR	ANANTNAG	BIDDER
1012	JAMMU & KASHMIR	ANANTNAG	BIJBEHARA
1013	JAMMU & KASHMIR	ANANTNAG	DEVSAR
1014	JAMMU & KASHMIR	ANANTNAG	DORU
1015	JAMMU & KASHMIR	ANANTNAG	H C GAM
1016	JAMMU & KASHMIR	ANANTNAG	MATTAN
1017	JAMMU & KASHMIR	ANANTNAG	SRIGUFWARA
1018	JAMMU & KASHMIR	ANANTNAG	VAILOO
1019	JAMMU & KASHMIR	ANANTNAG	VERINAG
1020	JAMMU & KASHMIR	ANANTNAG	YARIPORA
1021	JAMMU & KASHMIR	BADGAM	CHARI-SHARIEF
1022	JAMMU & KASHMIR	BADGAM	DREYGAM
1023	JAMMU & KASHMIR	BADGAM	HARDU-PUNZOO
1024	JAMMU & KASHMIR	BADGAM	MAGAM
1025	JAMMU & KASHMIR	BADGAM	SOIBUGH
1026	JAMMU & KASHMIR	BARAMULA	CHANDANWARI
1027	JAMMU & KASHMIR	BARAMULA	CHANDOOSA
1028	JAMMU & KASHMIR	BARAMULA	DANGERPORA
1029	JAMMU & KASHMIR	BARAMULA	DANGIWACHA
1030	JAMMU & KASHMIR	BARAMULA	FATEHGARH
1031	JAMMU & KASHMIR	BARAMULA	JULLAH
1032	JAMMU & KASHMIR	BARAMULA	KUNZAR
1033	JAMMU & KASHMIR	BARAMULA	NEHALPORA
1034	JAMMU & KASHMIR	BARAMULA	QUILMUQAM
1035	JAMMU & KASHMIR	BARAMULA	SINGHPORA (KALAN)
1036	JAMMU & KASHMIR	BARAMULA	SINGHPORA (PATTAN)
1037	JAMMU & KASHMIR	DODA	BATOTE
1038	JAMMU & KASHMIR	DODA	BHADERWAH
1039	JAMMU & KASHMIR	DODA	BHALA
1040	JAMMU & KASHMIR	DODA	BHATYAS
1041	JAMMU & KASHMIR	DODA	DRABSHALA

1043       JAMMU & KASHMIR       DODA       GUNDNA         1044       JAMMU & KASHMIR       DODA       NAGSENI         1046       JAMMU & KASHMIR       DODA       NAGSENI         1046       JAMMU & KASHMIR       JAMMU       AKHNOR         1047       JAMMU & KASHMIR       JAMMU       AKHNOR         1048       JAMMU & KASHMIR       JAMMU       AKHNOR         1049       JAMMU & KASHMIR       JAMMU       AKHNOR         1051       JAMMU & KASHMIR       JAMMU       BIALWAL         1052       JAMMU & KASHMIR       JAMMU       GANDHI NAGAR         1053       JAMMU & KASHMIR       JAMMU       GANDHI NAGAR         1054       JAMMU & KASHMIR       JAMMU       GANDHI NAGAR         1055       JAMMU & KASHMIR       JAMMU       MIRAN         1056       JAMMU & KASHMIR       JAMMU       MIRAN         1057       JAMMU & KASHMIR       JAMMU       MIRAN         1058       JAMMU & KASHMIR       JAMMU       MIRAN         1059       JAMMU & KASHMIR       JAMMU       RAMGARH         1060       JAMMU & KASHMIR       JAMMU       RAMGARH         1061       JAMMU & KASHMIR       JAMMU       SATWAR	1042	JAMMU & KASHMIR	DODA	GHAT
1044         JAMMU & KASIIMIR         DODA         KIIARI           1045         JAMMU & KASIMIR         DODA         NAGSENI           1046         JAMMU & KASIMIR         JAMMU         AKIIA           1047         JAMMU & KASIMIR         JAMMU         AKINOOR           1048         JAMMU & KASIMIR         JAMMU         AKINOOR           1049         JAMMU & KASIMIR         JAMMU         BIALWAL           1051         JAMMU & KASIMIR         JAMMU         BISHNAH           1052         JAMMU & KASIMIR         JAMMU         CHOWKI CHOURA           1053         JAMMU & KASIMIR         JAMMU         GANDHI NGGAR           1054         JAMMU & KASIMIR         JAMMU         GANDHI NGGAR           1055         JAMMU & KASIMIR         JAMMU         KHOUR           1056         JAMMU & KASIMIR         JAMMU         KHOUR           1057         JAMMU & KASIMIR         JAMMU         MARH           1060         JAMMU & KASIMIR         JAMMU         RAGRAH           1061         JAMMU & KASIMIR         JAMMU         RAMGRAH           1062         JAMMU & KASIMIR         JAMMU         RAGRAH           1064         JAMMU & KASIMIR         JAMMU				
1045     JAMMU & KASHMIR     DODA     NAGSENI       1046     JAMMU & KASHMIR     DODA     UKHRAL       1047     JAMMU & KASHMIR     JAMMU     AKIINOOR       1048     JAMMU & KASHMIR     JAMMU     AKIINOOR       1049     JAMMU & KASHMIR     JAMMU     AKIINOOR       1050     JAMU & KASHMIR     JAMMU     BIIALWAL       1051     JAMMU & KASHMIR     JAMMU     BISHNAH       1052     JAMMU & KASHMIR     JAMMU     CHOWKI CHOURA       1053     JAMMU & KASHMIR     JAMMU     GANDHI NAGAR       1054     JAMMU & KASHMIR     JAMMU     JAMMU       1055     JAMMU & KASHMIR     JAMMU     JAMMU       1056     JAMMU & KASHMIR     JAMMU     MIRAN SAHB       1057     JAMMU & KASHMIR     JAMMU     MIRAN SAHB       1058     JAMMU & KASHMIR     JAMMU     MIRAN SAHB       1061     JAMMU & KASHMIR     JAMMU     RAMGARH       1061     JAMMU & KASHMIR     JAMMU     SAMBA       1062     JAMMU & KASHMIR     JAMMU     SAMBA       1064     JAMMU & KASHMIR     JAMMU     SAMBA       1065     JAMMU & KASHMIR     JAMMU     SATWARI       1064     JAMMU & KASHMIR     KATHUA     BILADU				
1046       JAMMU & KASHMIR       DODA       UKHRAL         1047       JAMMU & KASHMIR       JAMMU       AKHIOOR         1048       JAMMU & KASHMIR       JAMMU       AKNIA         1049       JAMMU & KASHMIR       JAMMU       BHALWAL         1050       JAMMU & KASHMIR       JAMMU       BHALWAL         1051       JAMMU & KASHMIR       JAMMU       CHOWKI CHOURA         1052       JAMMU & KASHMIR       JAMMU       GANDHI NAGAR         1053       JAMMU & KASHMIR       JAMMU       GANDHI NAGAR         1054       JAMMU & KASHMIR       JAMMU       GANDHI NAGAR         1055       JAMMU & KASHMIR       JAMMU       KAGAR         1056       JAMMU & KASHMIR       JAMMU       MARH         1057       JAMMU & KASHMIR       JAMMU       MARH         1058       JAMMU & KASHMIR       JAMMU       RARGAR         1060       JAMMU & KASHMIR       JAMMU       RARGAR         1061       JAMMU & KASHMIR       JAMMU       RAMGAR         1064       JAMMU & KASHMIR       JAMMU       SATWAR         1064       JAMMU & KASHMIR       KATHUA       BHADDU         1065       JAMMU & KASHMIR       KATHUA <td< td=""><td>-</td><td></td><td></td><td></td></td<>	-			
1047       JAMMU & KASHMIR       JAMMU       AKHNOOR         1048       JAMMU & KASHMIR       JAMMU       BIALWAL         1050       JAMMU & KASHMIR       JAMMU       BISHALM         1051       JAMMU & KASHMIR       JAMMU       DANSAL         1052       JAMMU & KASHMIR       JAMMU       DANSAL         1053       JAMMU & KASHMIR       JAMMU       DANSAL         1054       JAMMU & KASHMIR       JAMMU       DANSAL         1055       JAMMU & KASHMIR       JAMMU       DANSAL         1055       JAMMU & KASHMIR       JAMMU       JAMMU         1056       JAMMU & KASHMIR       JAMMU       MARH         1057       JAMMU & KASHMIR       JAMMU       KASHMIR         1058       JAMMU & KASHMIR       JAMMU       KASHMIR         1059       JAMMU & KASHMIR       JAMMU       RARH         1060       JAMMU & KASHMIR       JAMMU       RAGARH         1061       JAMMU & KASHMIR       JAMMU       SATWARI         1061       JAMMU & KASHMIR       JAMMU       SATWARI         1062       JAMMU & KASHMIR       JAMMU       SATWARI         1064       JAMMU & KASHMIR       JAMMU       SATWARI </td <td></td> <td></td> <td></td> <td></td>				
1048       JAMMU & KASHMIR       JAMMU       ARNIA         1049       JAMMU & KASHMIR       JAMMU       BHALWAL         1050       JAMMU & KASHMIR       JAMMU       BISHNAH         1051       JAMMU & KASHMIR       JAMMU       CHOWKI CHOURA         1052       JAMMU & KASHMIR       JAMMU       CHOWKI CHOURA         1053       JAMMU & KASHMIR       JAMMU       GANDHI NAGAR         1054       JAMMU & KASHMIR       JAMMU       JAMMU         1055       JAMMU & KASHMIR       JAMMU       JAMMU         1055       JAMMU & KASHMIR       JAMMU       MARI         1057       JAMMU & KASHMIR       JAMMU       MARI         1058       JAMMU & KASHMIR       JAMMU       MARI         1059       JAMMU & KASHMIR       JAMMU       PURMANDAL         1060       JAMMU & KASHMIR       JAMMU       RAGGAR         1061       JAMMU & KASHMIR       JAMMU       RAGGAR         1062       JAMMU & KASHMIR       JAMMU       SAMBA         1064       JAMMU & KASHMIR       JAMMU       SAHARGOLE         1064       JAMMU & KASHMIR       JAMMU       VJAYPUR         1065       JAMMU & KASHMIR       KARGIL       SH				
1049       JAMMU & KASHMIR       JAMMU       BHALWAL         1050       JAMMU & KASHMIR       JAMMU       BISHNAH         1051       JAMU & KASHMIR       JAMMU       CHOWKI CHOURA         1052       JAMMU & KASHMIR       JAMMU       GANDHI NAGAR         1054       JAMMU & KASHMIR       JAMMU       GANDHI NAGAR         1055       JAMMU & KASHMIR       JAMMU       GANDHI NAGAR         1056       JAMMU & KASHMIR       JAMMU       GANDHI NAGAR         1057       JAMMU & KASHMIR       JAMMU       KASHMIR         1058       JAMMU & KASHMIR       JAMMU       KIOUR         1059       JAMMU & KASHMIR       JAMMU       MIRAN SAHIB         1060       JAMMU & KASHMIR       JAMMU       RASPURA         1061       JAMMU & KASHMIR       JAMMU       RAMCARH         1062       JAMMU & KASHMIR       JAMMU       SAMBA         1063       JAMMU & KASHMIR       JAMMU       SAMBA         1064       JAMMU & KASHMIR       JAMUU       SAMBA         1065       JAMMU & KASHMIR       KATUA       BAROTI         1066       JAMMU & KASHMIR       KATUA       BAROTI         1066       JAMMU & KASHMIR       KATUA </td <td></td> <td></td> <td></td> <td></td>				
1050     JAMMU & KASHMIR     JAMMU     BISHNAH       1051     JAMMU & KASHMIR     JAMMU     CHOWKI CHOURA       1052     JAMMU & KASHMIR     JAMMU     DANSAL       1053     JAMMU & KASHMIR     JAMMU     GANDHI NAGAR       1054     JAMMU & KASHMIR     JAMMU     JAMMU       1055     JAMMU & KASHMIR     JAMMU     JOURIAN       1056     JAMMU & KASHMIR     JAMMU     KIOUR       1057     JAMMU & KASHMIR     JAMMU     MARH       1058     JAMMU & KASHMIR     JAMMU     MARH       1059     JAMMU & KASHMIR     JAMMU     MARH       1061     JAMMU & KASHMIR     JAMMU     RARA       1061     JAMMU & KASHMIR     JAMMU     RAMGARH       1062     JAMMU & KASHMIR     JAMMU     SAMBA       1064     JAMMU & KASHMIR     JAMMU     SAMBA       1065     JAMMU & KASHMIR     JAMMU     SAMBA       1066     JAMMU & KASHMIR     JAMMU     SAMBA       1066     JAMMU & KASHMIR     JAMMU     SAMBA       1066     JAMMU & KASHMIR     KATHUA     BHAROIL       1066     JAMMU & KASHMIR     KATHUA     BHAROIL       1066     JAMMU & KASHMIR     KATHUA     BHAROIL       10				
1051       JAMMU & KASHMIR       JAMMU       CHOWKI CHOURA         1052       JAMMU & KASHMIR       JAMMU       DANSAL         1053       JAMMU & KASHMIR       JAMMU       GANDHI NAGAR         1054       JAMMU & KASHMIR       JAMMU       GANDHI NAGAR         1055       JAMMU & KASHMIR       JAMMU       JAMMU         1056       JAMMU & KASHMIR       JAMMU       KHOUR         1057       JAMMU & KASHMIR       JAMMU       MARH         1058       JAMMU & KASHMIR       JAMMU       MARH         1060       JAMMU & KASHMIR       JAMMU       PURAANDAL         1060       JAMMU & KASHMIR       JAMMU       RAMGARH         1061       JAMMU & KASHMIR       JAMMU       RAMGARH         1062       JAMMU & KASHMIR       JAMMU       SATMA         1063       JAMMU & KASHMIR       JAMMU       SATMA         1064       JAMMU & KASHMIR       JAMMU       VJAYPUR         1065       JAMMU & KASHMIR       KATHUA       BARGOLE         1066       JAMMU & KASHMIR       KATHUA       BARGOLE         1066       JAMMU & KASHMIR       KATHUA       BILA WAR         1067       JAMMU & KASHMIR       KATHUA       <	-			
1052     JAMMU & KASHMIR     JAMMU     DANSAL       1053     JAMMU & KASHMIR     JAMMU     GANDHI NAGAR       1054     JAMMU & KASHMIR     JAMMU     JAMMU       1055     JAMMU & KASHMIR     JAMMU     MARNU       1056     JAMMU & KASHMIR     JAMMU     KHOUR       1057     JAMMU & KASHMIR     JAMMU     MARN       1058     JAMMU & KASHMIR     JAMMU     MARN       1059     JAMMU & KASHMIR     JAMMU     MARN       1061     JAMMU & KASHMIR     JAMMU     RAMGARH       1062     JAMMU & KASHMIR     JAMMU     RAMGARH       1064     JAMMU & KASHMIR     JAMMU     SATWAR       1065     JAMMU & KASHMIR     JAMMU     SATWARI       1066     JAMMU & KASHMIR     JAMMU     SATWARI       1066     JAMMU & KASHMIR     KARGIL     SHARGOLE       1066     JAMMU & KASHMIR     KATHUA     BHADDU       1066     JAMMU & KASHMIR     KATHUA     BHADDU <td< td=""><td>-</td><td></td><td></td><td></td></td<>	-			
1053       JAMMU & KASHMIR       JAMMU       GANDHI NAGAR         1054       JAMMU & KASHMIR       JAMMU       JAMMU       JAMMU         1055       JAMMU & KASHMIR       JAMMU       JOURIAN         1056       JAMMU & KASHMIR       JAMMU       KHOUR         1057       JAMMU & KASHMIR       JAMMU       MARH         1058       JAMMU & KASHMIR       JAMMU       MARA         1059       JAMMU & KASHMIR       JAMMU       RARASHIB         1060       JAMMU & KASHMIR       JAMMU       RAMGARH         1061       JAMMU & KASHMIR       JAMMU       RAMGARH         1062       JAMMU & KASHMIR       JAMMU       SATWARI         1064       JAMMU & KASHMIR       JAMMU       SATWARI         1065       JAMMU & KASHMIR       JAMMU       VIAYPUR         1066       JAMMU & KASHMIR       KARGIL       SHARGOLE         1066       JAMMU & KASHMIR       KATHUA       BARNOTI         1067       JAMMU & KASHMIR       KATHUA       BILAWAR         1068       JAMMU & KASHMIR       KATHUA       BILAWAR         1069       JAMMU & KASHMIR       KATHUA       BILAWAR         1070       JAMMU & KASHMIR       KAT				
1054       JAMMU & KASHMIR       JAMMU       JAMMU         1055       JAMMU & KASHMIR       JAMMU       JOURIAN         1056       JAMMU & KASHMIR       JAMMU       KHOUR         1057       JAMMU & KASHMIR       JAMMU       MARH         1058       JAMMU & KASHMIR       JAMMU       MRAN SAHIB         1059       JAMMU & KASHMIR       JAMMU       MRAN SAHIB         1060       JAMMU & KASHMIR       JAMMU       RANGARH         1061       JAMMU & KASHMIR       JAMMU       RAMGARH         1062       JAMMU & KASHMIR       JAMMU       SAMBA         1063       JAMMU & KASHMIR       JAMMU       SATWARI         1064       JAMMU & KASHMIR       JAMMU       SATWARI         1065       JAMMU & KASHMIR       KARGIL       SHARGOLE         1066       JAMMU & KASHMIR       KARGIL       SHARGOLE         1066       JAMMU & KASHMIR       KATHUA       BHADDU         1065       JAMMU & KASHMIR       KATHUA       BHAROTI         1066       JAMMU & KASHMIR       KATHUA       BHAROTI         1066       JAMMU & KASHMIR       KATHUA       BHADDU         1068       JAMMU & KASHMIR       KATHUA       B				
1055     JAMMU & KASHMIR     JAMMU     JOURIAN       1056     JAMMU & KASHMIR     JAMMU     KHOUR       1057     JAMMU & KASHMIR     JAMMU     MARH       1058     JAMMU & KASHMIR     JAMMU     MIRAN SAHIB       1059     JAMMU & KASHMIR     JAMMU     PURMANDAL       1060     JAMMU & KASHMIR     JAMMU     RAS SPURA       1061     JAMMU & KASHMIR     JAMMU     RAMGARH       1062     JAMMU & KASHMIR     JAMMU     SATWARI       1063     JAMMU & KASHMIR     JAMMU     SATWARI       1064     JAMMU & KASHMIR     JAMMU     SATWARI       1065     JAMMU & KASHMIR     KARGIL     SHARGOLE       1064     JAMMU & KASHMIR     KARGIL     SHARGOLE       1065     JAMMU & KASHMIR     KATHUA     BARNOTI       1066     JAMMU & KASHMIR     KATHUA     BHADDU       1066     JAMMU & KASHMIR     KATHUA     BHADDU       1067     JAMMU & KASHMIR     KATHUA     BHADDU       1068     JAMMU & KASHMIR     KATHUA     BHADDU       1069     JAMMU & KASHMIR     KATHUA     HIRANAGAR       1070     JAMMU & KASHMIR     KATHUA     HIRANAGAR       1071     JAMMU & KASHMIR     KATHUA     LAKHAPUR </td <td></td> <td></td> <td></td> <td></td>				
1056       JAMMU & KASHMIR       JAMMU       KHOUR         1057       JAMMU & KASHMIR       JAMMU       MARH         1058       JAMMU & KASHMIR       JAMMU       MIRAN SAHIB         1059       JAMMU & KASHMIR       JAMMU       PURMANDAL         1060       JAMMU & KASHMIR       JAMMU       R.S. PURA         1061       JAMMU & KASHMIR       JAMMU       RAMGARH         1062       JAMMU & KASHMIR       JAMMU       SAMBA         1063       JAMMU & KASHMIR       JAMMU       SAMBA         1064       JAMMU & KASHMIR       JAMMU       SATWARI         1064       JAMMU & KASHMIR       JAMMU       SATWARI         1065       JAMMU & KASHMIR       KARGIL       SHARGOLE         1066       JAMMU & KASHMIR       KATHUA       BHAROTI         1066       JAMMU & KASHMIR       KATHUA       BHAROU         1067       JAMMU & KASHMIR       KATHUA       BHAROU         1068       JAMMU & KASHMIR       KATHUA       BHAROU         1070       JAMMU & KASHMIR       KATHUA       HIRANAGAR         1071       JAMMU & KASHMIR       KATHUA       HARNAGAR         1071       JAMMU & KASHMIR       KATHUA       <				
1057       JAMMU & KASHMIR       JAMMU       MARH         1058       JAMMU & KASHMIR       JAMMU       MIRAN SAHIB         1059       JAMMU & KASHMIR       JAMMU       PURMANDAL         1060       JAMMU & KASHMIR       JAMMU       R.S. PURA         1061       JAMMU & KASHMIR       JAMMU       RAMGARH         1062       JAMMU & KASHMIR       JAMMU       SAMBA         1063       JAMMU & KASHMIR       JAMMU       SATWARI         1064       JAMMU & KASHMIR       JAMMU       VIJAYPUR         1065       JAMMU & KASHMIR       KARGIL       SHARGOLE         1066       JAMMU & KASHMIR       KATHUA       BARNOTI         1067       JAMMU & KASHMIR       KATHUA       BHADDU         1068       JAMMU & KASHMIR       KATHUA       BHADDU         1069       JAMMU & KASHMIR       KATHUA       BHADDU         1069       JAMMU & KASHMIR       KATHUA       BHADDU         1069       JAMMU & KASHMIR       KATHUA       BHADU         1069       JAMMU & KASHMIR       KATHUA       BHADU         1070       JAMMU & KASHMIR       KATHUA       HAGWAL         1071       JAMMU & KASHMIR       KATHUA <td< td=""><td>-</td><td></td><td></td><td></td></td<>	-			
1058JAMMU & KASHMIRJAMMUMIRAN SAHIB1059JAMMU & KASHMIRJAMMUPURMANDAL1060JAMMU & KASHMIRJAMMUR.S. PURA1061JAMMU & KASHMIRJAMMURAMGARH1062JAMMU & KASHMIRJAMMUSAMBA1063JAMMU & KASHMIRJAMMUSATWARI1064JAMMU & KASHMIRJAMMUSATWARI1065JAMMU & KASHMIRJAMMUVIJAYPUR1065JAMMU & KASHMIRKARGILSHARGOLE1066JAMMU & KASHMIRKATHUABARNOTI1067JAMMU & KASHMIRKATHUABHADDU1068JAMMU & KASHMIRKATHUABHADDU1069JAMMU & KASHMIRKATHUABHADDU1070JAMMU & KASHMIRKATHUAHIRANAGAR1071JAMMU & KASHMIRKATHUAHIRANAGAR1072JAMMU & KASHMIRKATHUALAKHANPUR1073JAMMU & KASHMIRKATHUAMARHEEN1074JAMMU & KASHMIRKATHUAMARHEN1075JAMMU & KASHMIRKUPWARACHAMKOOT1076JAMMU & KASHMIRKUPWARADRUGMULA1077JAMMU & KASHMIRKUPWARADRUGMULA1078JAMMU & KASHMIRKUPWARAHANDWARA1081JAMMU & KASHMIRKUPWARAMARER1081JAMMU & KASHMIRLEH (LADAKH)LEH1084JAMMU & KASHMIRLEH (LADAKH)LEH1084JAMMU & KASHMIRPULWAMAAWANTIPORA1085 <t< td=""><td></td><td></td><td></td><td></td></t<>				
1059JAMMU & KASHMIRJAMMUPURMANDAL1060JAMMU & KASHMIRJAMMUR.S.PURA1061JAMMU & KASHMIRJAMMURAMGARH1062JAMMU & KASHMIRJAMMUSAMBA1063JAMMU & KASHMIRJAMMUSATWARI1064JAMMU & KASHMIRJAMMUSATWARI1065JAMMU & KASHMIRKARGILSHARGOLE10666JAMMU & KASHMIRKARTIUABARNOTI1067JAMMU & KASHMIRKATHUABHADDU1068JAMMU & KASHMIRKATHUABHLLAWAR1069JAMMU & KASHMIRKATHUABILLAWAR1069JAMMU & KASHMIRKATHUAHIRANAGGAR1070JAMMU & KASHMIRKATHUAHIRANGGAR1071JAMMU & KASHMIRKATHUAKATHUA1072JAMMU & KASHMIRKATHUAKATHUA1073JAMMU & KASHMIRKATHUAMAHANPUR1074JAMMU & KASHMIRKATHUAMAHANPUR1075JAMMU & KASHMIRKUPWARACHAMKOOT1077JAMMU & KASHMIRKUPWARADRUGMULA1078JAMMU & KASHMIRKUPWARAMAWER1081JAMMU & KASHMIRKUPWARAMAWER1082JAMMU & KASHMIRKUPWARAMAWER1083JAMMU & KASHMIRLEH (LADAKH)LEH1084JAMMU & KASHMIRPULWAMAAWANTIPORA1085JAMMU & KASHMIRPULWAMAMAMERG1086JAMU & KASHMIRPULWAMAIMAMSAHAIB1086J				
1060JAMMU & KASHMIRJAMMUR.S. PURA1061JAMMU & KASHMIRJAMMURAMGARH1062JAMMU & KASHMIRJAMMUSAMBA1063JAMMU & KASHMIRJAMMUSATWARI1064JAMMU & KASHMIRJAMMUVIJAYPUR1065JAMMU & KASHMIRKARGILSHARGOLE1066JAMMU & KASHMIRKATHUABARNOTI1067JAMMU & KASHMIRKATHUABHADDU1068JAMMU & KASHMIRKATHUABHLLAWAR1069JAMMU & KASHMIRKATHUABILLAWAR1069JAMMU & KASHMIRKATHUAHIRANAGAR1070JAMMU & KASHMIRKATHUAGHAGWAL1071JAMMU & KASHMIRKATHUAHIRANAGAR1072JAMMU & KASHMIRKATHUAKATHUA1073JAMMU & KASHMIRKATHUAMAHANPUR1074JAMMU & KASHMIRKATHUAMAHANPUR1075JAMMU & KASHMIRKATHUAMARHEEN1076JAMMU & KASHMIRKUPWARACHAMKOOT1077JAMMU & KASHMIRKUPWARACHAMKOOT1077JAMMU & KASHMIRKUPWARAHANDWARA1080JAMMU & KASHMIRKUPWARAMAWER1081JAMMU & KASHMIRLEH (LADAKH)LEH1082JAMMU & KASHMIRLEH (LADAKH)LEH1084JAMMU & KASHMIRPULWAMAAWANTIPORA1085JAMMU & KASHMIRPULWAMAMAMTPORA1086JAMMU & KASHMIRPULWAMAIMAMSAHAIB1086 </td <td></td> <td></td> <td></td> <td></td>				
1061JAMMU & KASHMIRJAMMURAMGARH1062JAMMU & KASHMIRJAMMUSAMBA1063JAMMU & KASHMIRJAMMUSATWARI1064JAMMU & KASHMIRJAMMUVIJAYPUR1065JAMMU & KASHMIRKARGILSHARGOLE1066JAMMU & KASHMIRKATHUABARNOTI1067JAMMU & KASHMIRKATHUABHADDU1068JAMMU & KASHMIRKATHUABHADDU1069JAMMU & KASHMIRKATHUABHLAWAR1069JAMMU & KASHMIRKATHUAGHAGWAL1070JAMMU & KASHMIRKATHUAGHAGWAL1071JAMMU & KASHMIRKATHUAHIRANAGAR1072JAMMU & KASHMIRKATHUALAKHANPUR1073JAMMU & KASHMIRKATHUAMAHPUR1074JAMMU & KASHMIRKATHUAMAHPUR1075JAMMU & KASHMIRKATHUASALLAN1076JAMMU & KASHMIRKUPWARACHAMKOOT1077JAMMU & KASHMIRKUPWARADRUGMULA1078JAMMU & KASHMIRKUPWARAHANDWARA1079JAMMU & KASHMIRKUPWARAHANDWARA1080JAMMU & KASHMIRKUPWARAMAWER1081JAMMU & KASHMIRLEH (LADAKH)KHUTSE1083JAMMU & KASHMIRLEH (LADAKH)LEH1084JAMMU & KASHMIRPULWAMAAWANTIPORA1085JAMMU & KASHMIRPULWAMAMAMSAHAIB1086JAMMU & KASHMIRPULWAMAIMAMSAHAIB1086 <td></td> <td></td> <td></td> <td></td>				
1062JAMMU & KASHMIRJAMMUSAMBA1063JAMMU & KASHMIRJAMMUSATWARI1064JAMMU & KASHMIRJAMMUVIJAYPUR1065JAMMU & KASHMIRKARGILSHARGOLE1066JAMMU & KASHMIRKATHUABARNOTI1067JAMMU & KASHMIRKATHUABHADDU1068JAMMU & KASHMIRKATHUABILLAWAR1069JAMMU & KASHMIRKATHUAGHAGWAL1070JAMMU & KASHMIRKATHUAGHAGWAL1071JAMMU & KASHMIRKATHUAHRANAGAR1072JAMMU & KASHMIRKATHUAKATHUA1073JAMMU & KASHMIRKATHUAMAHANPUR1074JAMMU & KASHMIRKATHUAMAHAPUR1075JAMMU & KASHMIRKATHUASALLAN1076JAMMU & KASHMIRKUPWARACHAMKOOT1077JAMMU & KASHMIRKUPWARADRUGMULA1078JAMMU & KASHMIRKUPWARAMARGOT1079JAMMU & KASHMIRKUPWARAMAWER1080JAMMU & KASHMIRKUPWARAMAWER1081JAMMU & KASHMIRLEH (LADAKH)LEH1082JAMMU & KASHMIRLEH (LADAKH)LEH1084JAMMU & KASHMIRPULWAMAAWANTIPORA1085JAMMU & KASHMIRPULWAMAKEGAM1086JAMMU & KASHMIRPULWAMAKEGAM1087JAMMU & KASHMIRPULWAMASHADIMERG1080JAMMU & KASHMIRPULWAMASHADIMERG1084JAMM				
1063JAMMU & KASHMIRJAMMUSATWARI1064JAMMU & KASHMIRJAMMUVIJAYPUR1065JAMMU & KASHMIRKARGILSHARGOLE1066JAMMU & KASHMIRKATHUABARNOTI1067JAMMU & KASHMIRKATHUABHADDU1068JAMMU & KASHMIRKATHUABILLAWAR1069JAMMU & KASHMIRKATHUABILLAWAR1070JAMMU & KASHMIRKATHUAHIRANAGAR1071JAMMU & KASHMIRKATHUAKATHUA1072JAMMU & KASHMIRKATHUALAKHANPUR1073JAMMU & KASHMIRKATHUALAKHANPUR1074JAMMU & KASHMIRKATHUAMARHEEN1075JAMMU & KASHMIRKATHUASALLAN1076JAMMU & KASHMIRKUPWARACHAMKOOT1077JAMMU & KASHMIRKUPWARADRUGMULA1078JAMMU & KASHMIRKUPWARAMAWER1080JAMMU & KASHMIRKUPWARAMAWER1081JAMMU & KASHMIRKUPWARAMAWER1082JAMMU & KASHMIRLEH (LADAKH)KHALTSE1083JAMMU & KASHMIRPULWAMAAWANTIPORA1084JAMMU & KASHMIRPULWAMAAWANTIPORA1085JAMMU & KASHMIRPULWAMASHADIMERG1086JAMMU & KASHMIRPULWAMASHADIMERG1089JAMMU & KASHMIRPULWAMASHADIMERG1089JAMMU & KASHMIRPULWAMASHADIMERG1089JAMMU & KASHMIRPULWAMASHADIMERG<				
1064JAMMU & KASHMIRJAMMUVIJAYPUR1065JAMMU & KASHMIRKARGILSHARGOLE1066JAMMU & KASHMIRKATHUABARNOTI1067JAMMU & KASHMIRKATHUABHADDU1068JAMMU & KASHMIRKATHUABILLAWAR1069JAMMU & KASHMIRKATHUABILLAWAR1070JAMMU & KASHMIRKATHUAGHAGWAL1071JAMMU & KASHMIRKATHUAHIRANAGAR1072JAMMU & KASHMIRKATHUALAKHANPUR1073JAMMU & KASHMIRKATHUAMAHANPUR1074JAMMU & KASHMIRKATHUAMAHEEN1075JAMMU & KASHMIRKATHUASALLAN1076JAMMU & KASHMIRKUPWARACHAMKOOT1077JAMMU & KASHMIRKUPWARADRUGMULA1078JAMMU & KASHMIRKUPWARAHANDWARA1079JAMMU & KASHMIRKUPWARAMAWER1080JAMMU & KASHMIRKUPWARAMAER1081JAMMU & KASHMIRKUPWARAVILGAM1082JAMMU & KASHMIRLEH (LADAKH)KHALTSE1083JAMMU & KASHMIRLEH (LADAKH)LEH1084JAMMU & KASHMIRPULWAMAAWANTIPORA1085JAMMU & KASHMIRPULWAMAKEGAM1086JAMMU & KASHMIRPULWAMAKABGMI1086JAMMU & KASHMIRPULWAMAMAMSAHAIB1086JAMMU & KASHMIRPULWAMAMAMSAHAIB1086JAMMU & KASHMIRPULWAMASHADIMERG	-			
1065JAMMU & KASHMIRKARGILSHARGOLE1066JAMMU & KASHMIRKATHUABARNOTI1067JAMMU & KASHMIRKATHUABHADDU1068JAMMU & KASHMIRKATHUABILLAWAR1069JAMMU & KASHMIRKATHUAGHAGWAL1070JAMMU & KASHMIRKATHUAGHAGWAL1071JAMMU & KASHMIRKATHUAHIRANAGAR1072JAMMU & KASHMIRKATHUAKATHUA1073JAMMU & KASHMIRKATHUALAKHANPUR1074JAMMU & KASHMIRKATHUAMAHANPUR1075JAMMU & KASHMIRKATHUASALLAN1076JAMMU & KASHMIRKUPWARACHAMKOOT1077JAMMU & KASHMIRKUPWARADRUGMULA1078JAMMU & KASHMIRKUPWARAHUMRIYAL1080JAMMU & KASHMIRKUPWARAMAWER1081JAMMU & KASHMIRKUPWARAVILGAM1082JAMMU & KASHMIRLEH (LADAKH)LEH1083JAMMU & KASHMIRPULWAMAAWANTIPORA1084JAMMU & KASHMIRPULWAMAAWANTIPORA1085JAMMU & KASHMIRPULWAMAKEGGAM1086JAMMU & KASHMIRPULWAMAKADIMERG1089JAMMU & KASHMIRPULWAMAKHALTSE1084JAMMU & KASHMIRPULWAMAKABIMERG1085JAMMU & KASHMIRPULWAMAKHADIMERG1084JAMMU & KASHMIRPULWAMAKHADIMERG1084JAMMU & KASHMIRPULWAMAKHADIMERG <tr< td=""><td>-</td><td></td><td></td><td></td></tr<>	-			
1066JAMMU & KASHMIRKATHUABARNOTI1067JAMMU & KASHMIRKATHUABHADDU1068JAMMU & KASHMIRKATHUABILLAWAR1069JAMMU & KASHMIRKATHUAGHAGWAL1070JAMMU & KASHMIRKATHUAGHAGWAL1071JAMMU & KASHMIRKATHUAHIRANAGAR1072JAMMU & KASHMIRKATHUAKATHUA1073JAMMU & KASHMIRKATHUALAKHANPUR1074JAMMU & KASHMIRKATHUAMAHANPUR1075JAMMU & KASHMIRKATHUASALLAN1076JAMMU & KASHMIRKUPWARACHAMKOOT1077JAMMU & KASHMIRKUPWARADRUGMULA1078JAMMU & KASHMIRKUPWARAHANDWARA1079JAMMU & KASHMIRKUPWARAMAWER1080JAMMU & KASHMIRKUPWARAMAWER1081JAMMU & KASHMIRKUPWARAVILGAM1082JAMMU & KASHMIRLEH (LADAKH)LEH1084JAMMU & KASHMIRPULWAMAAWANTIPORA1085JAMMU & KASHMIRPULWAMAIMAMSAHAIB1086JAMMU & KASHMIRPULWAMAKEGGAM1087JAMMU & KASHMIRPULWAMASHADIMERG1089JAMMU & KASHMIRPULWAMASHADIMERG1090JAMMU & KASHMIRPULWAMAVEHIL				
1067JAMMU & KASHMIRKATHUABHADDU1068JAMMU & KASHMIRKATHUABILLAWAR1069JAMMU & KASHMIRKATHUAGHAGWAL1070JAMMU & KASHMIRKATHUAHIRANAGAR1071JAMMU & KASHMIRKATHUAHIRANAGAR1072JAMMU & KASHMIRKATHUALAKHANPUR1073JAMMU & KASHMIRKATHUALAKHANPUR1074JAMMU & KASHMIRKATHUAMAHANPUR1075JAMMU & KASHMIRKATHUAMARHEN1076JAMMU & KASHMIRKUPWARACHAMKOOT1077JAMMU & KASHMIRKUPWARADRUGMULA1078JAMMU & KASHMIRKUPWARAHANDWARA1079JAMMU & KASHMIRKUPWARAHANDWARA1080JAMMU & KASHMIRKUPWARAMAWER1081JAMMU & KASHMIRKUPWARAWILGAM1082JAMMU & KASHMIRLEH (LADAKH)LEH1083JAMMU & KASHMIRPULWAMAAWANTIPORA1084JAMMU & KASHMIRPULWAMAIMAMSAHAIB1085JAMMU & KASHMIRPULWAMAIMAMSAHAIB1086JAMMU & KASHMIRPULWAMAKEEGAM1087JAMMU & KASHMIRPULWAMAKEEGAM1088JAMMU & KASHMIRPULWAMASHADIMERG1090JAMMU & KASHMIRPULWAMATAHAB1090JAMMU & KASHMIRPULWAMAVEHIL				
1068JAMMU & KASHMIRKATHUABILLAWAR1069JAMMU & KASHMIRKATHUAGHAGWAL1070JAMMU & KASHMIRKATHUAHIRANAGAR1071JAMMU & KASHMIRKATHUAHIRANAGAR1072JAMMU & KASHMIRKATHUALAKHANPUR1073JAMMU & KASHMIRKATHUALAKHANPUR1074JAMMU & KASHMIRKATHUAMAHANPUR1075JAMMU & KASHMIRKATHUAMARHEEN1076JAMMU & KASHMIRKUPWARACHAMKOOT1077JAMMU & KASHMIRKUPWARADRUGMULA1078JAMMU & KASHMIRKUPWARADRUGMULA1079JAMMU & KASHMIRKUPWARAMAWER1080JAMMU & KASHMIRKUPWARAMAWER1081JAMMU & KASHMIRLEH (LADAKH)KHALTSE1083JAMMU & KASHMIRLEH (LADAKH)LEH1084JAMMU & KASHMIRPULWAMAAWANTIPORA1085JAMMU & KASHMIRPULWAMAIMAMSAHAIB1086JAMMU & KASHMIRPULWAMAKEEGAM1087JAMMU & KASHMIRPULWAMAKABHAIB1088JAMMU & KASHMIRPULWAMASHADIMERG1090JAMMU & KASHMIRPULWAMAYEHIL				
1069JAMMU & KASHMIRKATHUAGHAGWAL1070JAMMU & KASHMIRKATHUAHIRANAGAR1071JAMMU & KASHMIRKATHUAKATHUA1072JAMMU & KASHMIRKATHUALAKHANPUR1073JAMMU & KASHMIRKATHUAMAHANPUR1074JAMMU & KASHMIRKATHUAMARHEEN1075JAMMU & KASHMIRKATHUAMARHEEN1076JAMMU & KASHMIRKUPWARACHAMKOOT1077JAMMU & KASHMIRKUPWARADRUGMULA1078JAMMU & KASHMIRKUPWARADRUGMULA1079JAMMU & KASHMIRKUPWARAHANDWARA1079JAMMU & KASHMIRKUPWARAMAWER1080JAMMU & KASHMIRKUPWARAWILGAM1081JAMMU & KASHMIRKUPWARAVILGAM1082JAMMU & KASHMIRLEH (LADAKH)LEH1083JAMMU & KASHMIRLEH (LADAKH)LEH1084JAMMU & KASHMIRPULWAMAAWANTIPORA1085JAMMU & KASHMIRPULWAMAIMAMSAHAIB1086JAMMU & KASHMIRPULWAMAKEEGAM1087JAMMU & KASHMIRPULWAMAKAEGAM1088JAMMU & KASHMIRPULWAMASHADIMERG1090JAMMU & KASHMIRPULWAMATAHAB1090JAMMU & KASHMIRPULWAMAVEHIL				
1070JAMMU & KASHMIRKATHUAHIRANAGAR1071JAMMU & KASHMIRKATHUAKATHUA1072JAMMU & KASHMIRKATHUALAKHANPUR1073JAMMU & KASHMIRKATHUAMAHANPUR1074JAMMU & KASHMIRKATHUAMARHEEN1075JAMMU & KASHMIRKATHUAMARHEEN1076JAMMU & KASHMIRKUPWARACHAMKOOT1077JAMMU & KASHMIRKUPWARADRUGMULA1078JAMMU & KASHMIRKUPWARAHANDWARA1079JAMMU & KASHMIRKUPWARAMAWER1080JAMMU & KASHMIRKUPWARAMAWER1081JAMMU & KASHMIRKUPWARAVILGAM1082JAMMU & KASHMIRLEH (LADAKH)LEH1083JAMMU & KASHMIRPULWAMAAWANTIPORA1085JAMMU & KASHMIRPULWAMAIMAMSAHAIB1086JAMMU & KASHMIRPULWAMALOORGAM1087JAMMU & KASHMIRPULWAMASHADIMERG1088JAMMU & KASHMIRPULWAMASHADIMERG1089JAMMU & KASHMIRPULWAMAVEHIL				
1072JAMMU & KASHMIRKATHUALAKHANPUR1073JAMMU & KASHMIRKATHUAMAHANPUR1074JAMMU & KASHMIRKATHUAMARHEEN1075JAMMU & KASHMIRKATHUASALLAN1076JAMMU & KASHMIRKUPWARACHAMKOOT1077JAMMU & KASHMIRKUPWARADRUGMULA1078JAMMU & KASHMIRKUPWARAHANDWARA1079JAMMU & KASHMIRKUPWARAHANDWARA1079JAMMU & KASHMIRKUPWARAMAWER1080JAMMU & KASHMIRKUPWARAMAWER1081JAMMU & KASHMIRLEH (LADAKH)KHALTSE1082JAMMU & KASHMIRLEH (LADAKH)LEH1084JAMMU & KASHMIRPULWAMAAWANTIPORA1085JAMMU & KASHMIRPULWAMAIMAMSAHAIB1086JAMMU & KASHMIRPULWAMALOORGAM1087JAMMU & KASHMIRPULWAMASHADIMERG1088JAMMU & KASHMIRPULWAMATAHAB1090JAMMU & KASHMIRPULWAMAVEHIL	1070	JAMMU & KASHMIR	KATHUA	HIRANAGAR
1073JAMMU & KASHMIRKATHUAMAHANPUR1074JAMMU & KASHMIRKATHUAMARHEEN1075JAMMU & KASHMIRKATHUASALLAN1076JAMMU & KASHMIRKUPWARACHAMKOOT1077JAMMU & KASHMIRKUPWARADRUGMULA1078JAMMU & KASHMIRKUPWARAHANDWARA1079JAMMU & KASHMIRKUPWARAHANDWARA1079JAMMU & KASHMIRKUPWARAMAWER1080JAMMU & KASHMIRKUPWARAMAWER1081JAMMU & KASHMIRKUPWARAVILGAM1082JAMMU & KASHMIRLEH (LADAKH)KHALTSE1083JAMMU & KASHMIRLEH (LADAKH)LEH1084JAMMU & KASHMIRPULWAMAAWANTIPORA1085JAMMU & KASHMIRPULWAMAIMAMSAHAIB1086JAMMU & KASHMIRPULWAMAKEEGAM1087JAMMU & KASHMIRPULWAMALOORGAM1088JAMMU & KASHMIRPULWAMASHADIMERG1090JAMMU & KASHMIRPULWAMAVEHIL	1071	JAMMU & KASHMIR	KATHUA	KATHUA
1074JAMMU & KASHMIRKATHUAMARHEEN1075JAMMU & KASHMIRKATHUASALLAN1076JAMMU & KASHMIRKUPWARACHAMKOOT1077JAMMU & KASHMIRKUPWARADRUGMULA1078JAMMU & KASHMIRKUPWARAHANDWARA1079JAMMU & KASHMIRKUPWARAHANDWARA1080JAMMU & KASHMIRKUPWARAMAWER1081JAMMU & KASHMIRKUPWARAMAWER1082JAMMU & KASHMIRLEH (LADAKH)KHALTSE1083JAMMU & KASHMIRLEH (LADAKH)LEH1084JAMMU & KASHMIRPULWAMAAWANTIPORA1085JAMMU & KASHMIRPULWAMAIMAMSAHAIB1086JAMMU & KASHMIRPULWAMAKEEGAM1087JAMMU & KASHMIRPULWAMALOORGAM1089JAMMU & KASHMIRPULWAMASHADIMERG1090JAMMU & KASHMIRPULWAMAVEHIL	1072	JAMMU & KASHMIR	KATHUA	LAKHANPUR
1075JAMMU & KASHMIRKATHUASALLAN1076JAMMU & KASHMIRKUPWARACHAMKOOT1077JAMMU & KASHMIRKUPWARADRUGMULA1078JAMMU & KASHMIRKUPWARAHANDWARA1079JAMMU & KASHMIRKUPWARAHANDWARA1080JAMMU & KASHMIRKUPWARAMAWER1081JAMMU & KASHMIRKUPWARAMAWER1082JAMMU & KASHMIRLEH (LADAKH)KHALTSE1083JAMMU & KASHMIRLEH (LADAKH)LEH1084JAMMU & KASHMIRPULWAMAAWANTIPORA1085JAMMU & KASHMIRPULWAMAIMAMSAHAIB1086JAMMU & KASHMIRPULWAMALOORGAM1087JAMMU & KASHMIRPULWAMAKEEGAM1088JAMMU & KASHMIRPULWAMASHADIMERG1089JAMMU & KASHMIRPULWAMATAHAB1090JAMMU & KASHMIRPULWAMAVEHIL	1073	JAMMU & KASHMIR	KATHUA	MAHANPUR
1076JAMMU & KASHMIRKUPWARACHAMKOOT1077JAMMU & KASHMIRKUPWARADRUGMULA1078JAMMU & KASHMIRKUPWARAHANDWARA1079JAMMU & KASHMIRKUPWARAHANDWARA1080JAMMU & KASHMIRKUPWARAMAWER1081JAMMU & KASHMIRKUPWARAVILGAM1082JAMMU & KASHMIRLEH (LADAKH)KHALTSE1083JAMMU & KASHMIRLEH (LADAKH)LEH1084JAMMU & KASHMIRPULWAMAAWANTIPORA1085JAMMU & KASHMIRPULWAMAIMAMSAHAIB1086JAMMU & KASHMIRPULWAMALOORGAM1087JAMMU & KASHMIRPULWAMASHADIMERG1089JAMMU & KASHMIRPULWAMATAHAB1090JAMMU & KASHMIRPULWAMAVEHIL	1074	JAMMU & KASHMIR	KATHUA	MARHEEN
1077JAMMU & KASHMIRKUPWARADRUGMULA1078JAMMU & KASHMIRKUPWARAHANDWARA1079JAMMU & KASHMIRKUPWARAKHUMRIYAL1080JAMMU & KASHMIRKUPWARAMAWER1081JAMMU & KASHMIRKUPWARAVILGAM1082JAMMU & KASHMIRLEH (LADAKH)KHALTSE1083JAMMU & KASHMIRLEH (LADAKH)LEH1084JAMMU & KASHMIRPULWAMAAWANTIPORA1085JAMMU & KASHMIRPULWAMAIMAMSAHAIB1086JAMMU & KASHMIRPULWAMALOORGAM1087JAMMU & KASHMIRPULWAMASHADIMERG1089JAMMU & KASHMIRPULWAMATAHAB1090JAMMU & KASHMIRPULWAMAVEHIL	1075	JAMMU & KASHMIR	KATHUA	SALLAN
1078JAMMU & KASHMIRKUPWARAHANDWARA1079JAMMU & KASHMIRKUPWARAKHUMRIYAL1080JAMMU & KASHMIRKUPWARAMAWER1081JAMMU & KASHMIRKUPWARAVILGAM1082JAMMU & KASHMIRLEH (LADAKH)KHALTSE1083JAMMU & KASHMIRLEH (LADAKH)LEH1084JAMMU & KASHMIRPULWAMAAWANTIPORA1085JAMMU & KASHMIRPULWAMAIMAMSAHAIB1086JAMMU & KASHMIRPULWAMALOORGAM1087JAMMU & KASHMIRPULWAMASHADIMERG1089JAMMU & KASHMIRPULWAMATAHAB1090JAMMU & KASHMIRPULWAMAVEHIL	1076	JAMMU & KASHMIR	KUPWARA	СНАМКООТ
1079JAMMU & KASHMIRKUPWARAKHUMRIYAL1080JAMMU & KASHMIRKUPWARAMAWER1081JAMMU & KASHMIRKUPWARAVILGAM1082JAMMU & KASHMIRLEH (LADAKH)KHALTSE1083JAMMU & KASHMIRLEH (LADAKH)LEH1084JAMMU & KASHMIRPULWAMAAWANTIPORA1085JAMMU & KASHMIRPULWAMAIMAMSAHAIB1086JAMMU & KASHMIRPULWAMAKEEGAM1087JAMMU & KASHMIRPULWAMALOORGAM1088JAMMU & KASHMIRPULWAMASHADIMERG1089JAMMU & KASHMIRPULWAMATAHAB1090JAMMU & KASHMIRPULWAMAVEHIL	1077	JAMMU & KASHMIR	KUPWARA	DRUGMULA
1080JAMMU & KASHMIRKUPWARAMAWER1081JAMMU & KASHMIRKUPWARAVILGAM1082JAMMU & KASHMIRLEH (LADAKH)KHALTSE1083JAMMU & KASHMIRLEH (LADAKH)LEH1084JAMMU & KASHMIRPULWAMAAWANTIPORA1085JAMMU & KASHMIRPULWAMAIMAMSAHAIB1086JAMMU & KASHMIRPULWAMAKEEGAM1087JAMMU & KASHMIRPULWAMALOORGAM1088JAMMU & KASHMIRPULWAMASHADIMERG1089JAMMU & KASHMIRPULWAMATAHAB1090JAMMU & KASHMIRPULWAMAVEHIL	1078	JAMMU & KASHMIR	KUPWARA	HANDWARA
1081JAMMU & KASHMIRKUPWARAVILGAM1082JAMMU & KASHMIRLEH (LADAKH)KHALTSE1083JAMMU & KASHMIRLEH (LADAKH)LEH1084JAMMU & KASHMIRPULWAMAAWANTIPORA1085JAMMU & KASHMIRPULWAMAIMAMSAHAIB1086JAMMU & KASHMIRPULWAMAKEEGAM1087JAMMU & KASHMIRPULWAMALOORGAM1088JAMMU & KASHMIRPULWAMASHADIMERG1089JAMMU & KASHMIRPULWAMATAHAB1090JAMMU & KASHMIRPULWAMAVEHIL	1079	JAMMU & KASHMIR	KUPWARA	KHUMRIYAL
1082JAMMU & KASHMIRLEH (LADAKH)KHALTSE1083JAMMU & KASHMIRLEH (LADAKH)LEH1084JAMMU & KASHMIRPULWAMAAWANTIPORA1085JAMMU & KASHMIRPULWAMAIMAMSAHAIB1086JAMMU & KASHMIRPULWAMAKEEGAM1087JAMMU & KASHMIRPULWAMALOORGAM1088JAMMU & KASHMIRPULWAMASHADIMERG1089JAMMU & KASHMIRPULWAMATAHAB1090JAMMU & KASHMIRPULWAMAVEHIL	1080	JAMMU & KASHMIR	KUPWARA	MAWER
1083JAMMU & KASHMIRLEH (LADAKH)LEH1084JAMMU & KASHMIRPULWAMAAWANTIPORA1085JAMMU & KASHMIRPULWAMAIMAMSAHAIB1086JAMMU & KASHMIRPULWAMAKEEGAM1087JAMMU & KASHMIRPULWAMALOORGAM1088JAMMU & KASHMIRPULWAMASHADIMERG1089JAMMU & KASHMIRPULWAMATAHAB1090JAMMU & KASHMIRPULWAMAVEHIL	1081	JAMMU & KASHMIR	KUPWARA	VILGAM
1084JAMMU & KASHMIRPULWAMAAWANTIPORA1085JAMMU & KASHMIRPULWAMAIMAMSAHAIB1086JAMMU & KASHMIRPULWAMAKEEGAM1087JAMMU & KASHMIRPULWAMALOORGAM1088JAMMU & KASHMIRPULWAMASHADIMERG1089JAMMU & KASHMIRPULWAMATAHAB1090JAMMU & KASHMIRPULWAMAVEHIL	1082	JAMMU & KASHMIR	LEH (LADAKH)	KHALTSE
1085JAMMU & KASHMIRPULWAMAIMAMSAHAIB1086JAMMU & KASHMIRPULWAMAKEEGAM1087JAMMU & KASHMIRPULWAMALOORGAM1088JAMMU & KASHMIRPULWAMASHADIMERG1089JAMMU & KASHMIRPULWAMATAHAB1090JAMMU & KASHMIRPULWAMAVEHIL	1083	JAMMU & KASHMIR	LEH (LADAKH)	LEH
1086JAMMU & KASHMIRPULWAMAKEEGAM1087JAMMU & KASHMIRPULWAMALOORGAM1088JAMMU & KASHMIRPULWAMASHADIMERG1089JAMMU & KASHMIRPULWAMATAHAB1090JAMMU & KASHMIRPULWAMAVEHIL	1084	JAMMU & KASHMIR	PULWAMA	AWANTIPORA
1087JAMMU & KASHMIRPULWAMALOORGAM1088JAMMU & KASHMIRPULWAMASHADIMERG1089JAMMU & KASHMIRPULWAMATAHAB1090JAMMU & KASHMIRPULWAMAVEHIL	1085	JAMMU & KASHMIR	PULWAMA	IMAMSAHAIB
1088JAMMU & KASHMIRPULWAMASHADIMERG1089JAMMU & KASHMIRPULWAMATAHAB1090JAMMU & KASHMIRPULWAMAVEHIL	1086	JAMMU & KASHMIR	PULWAMA	KEEGAM
1089JAMMU & KASHMIRPULWAMATAHAB1090JAMMU & KASHMIRPULWAMAVEHIL	1087	JAMMU & KASHMIR	PULWAMA	LOORGAM
1090 JAMMU & KASHMIR PULWAMA VEHIL	1088	JAMMU & KASHMIR	PULWAMA	SHADIMERG
	1089	JAMMU & KASHMIR	PULWAMA	ТАНАВ
	1090	JAMMU & KASHMIR	PULWAMA	VEHIL
1091 JAMMU & KASHMIR PUNCH BAFLIAZ		JAMMU & KASHMIR	PUNCH	BAFLIAZ
1092 JAMMU & KASHMIR PUNCH HARNI	1092	JAMMU & KASHMIR	PUNCH	HARNI
1093 JAMMU & KASHMIR PUNCH KUNIYIAN	1093	JAMMU & KASHMIR	PUNCH	KUNIYIAN
1094JAMMU & KASHMIRPUNCHMANKOTEMHRDEducationrfa/06.07.20127.9			PUNCH	

1095	JAMMU & KASHMIR	PUNCH	NANGALI
1096	JAMMU & KASHMIR	PUNCH	SATHRA
1097	JAMMU & KASHMIR	RAJAURI	BALJARALLAN
1098	JAMMU & KASHMIR	RAJAURI	DANDASAR
1099	JAMMU & KASHMIR	RAJAURI	DOONGI BRAHMANA
1100	JAMMU & KASHMIR	RAJAURI	KHAWAS
1101	JAMMU & KASHMIR	RAJAURI	KOTRANKA
1102	JAMMU & KASHMIR	RAJAURI	LOWER HATHAL
1103	JAMMU & KASHMIR	RAJAURI	MOGLA
1104	JAMMU & KASHMIR	RAJAURI	NOWSHERA
1105	JAMMU & KASHMIR	RAJAURI	PEERI
1106	JAMMU & KASHMIR	RAJAURI	SUNDERBANI
1107	JAMMU & KASHMIR	RAJAURI	THANAMANDI
1108	JAMMU & KASHMIR	SRINAGAR	BATMALOO
1109	JAMMU & KASHMIR	SRINAGAR	GULAB BAGH
1110	JAMMU & KASHMIR	SRINAGAR	HARIGANWAN
1111	JAMMU & KASHMIR	SRINAGAR	HAWAL
1112	JAMMU & KASHMIR	SRINAGAR	IDDGAH
1112	JAMMU & KASHMIR	SRINAGAR	NISHAT
1114	JAMMU & KASHMIR	SRINAGAR	RAINAWARI
1114	JAMMU & KASHMIK	SRINAGAR	TULMULLA
1115	JAMMU & KASHMIK JAMMU & KASHMIR		ZALDAGAR
1117	JAMMU & KASHMIK JAMMU & KASHMIR	SRINAGAR UDHAMPUR	BABEY
1117	JAMMU & KASHMIK JAMMU & KASHMIR	UDHAMPUR	CHASSANA
1118	JAMMU & KASHMIK JAMMU & KASHMIR	UDHAMPUR	CHASSANA CHINKAH
1120	JAMMU & KASHMIR	UDHAMPUR	JIB
1121	JAMMU & KASHMIR	UDHAMPUR	KULWANTA
1122	JAMMU & KASHMIR	UDHAMPUR	MAJALTA
1123	JAMMU & KASHMIR	UDHAMPUR	POUNI
1124	JAMMU & KASHMIR	UDHAMPUR	REASI
1125	JAMMU & KASHMIR	UDHAMPUR	TIKRI
1126	JAMMU & KASHMIR	UDHAMPUR	UDHAMPUR
1127	JHARKHAND	DHANBAD	BAGHMARA
1128	JHARKHAND	DHANBAD	DHANBAD
1129	JHARKHAND	GUMLA	CHAINPUR
1130	JHARKHAND	HAZARIBAG	HAZARIBAGH
1131	JHARKHAND	KODARMA	CHANDWARA
1132	JHARKHAND	LATEHAR	MAHUADAR
1133	JHARKHAND	PALAMU	CHHATARPUR WEST
1134	JHARKHAND	RANCHI	RATU
1135	JHARKHAND	RANCHI	TORPA
1136	JHARKHAND	SARAIKELA-KHARSAWAN	GAMHARIA
1137	JHARKHAND	SARAIKELA-KHARSAWAN	RAJNAGAR
1138	KARNATAKA	BANGALORE	ANEKAL
1139	KARNATAKA	BANGALORE	EAST
1140	KARNATAKA	BANGALORE	NORTH
1141	KARNATAKA	BANGALORE	SOUTH
1142	KARNATAKA	BANGALORE RURAL	DEVANAHALLI
1143	KARNATAKA	BANGALORE RURAL	DODDABALLAPURA
1144	KARNATAKA	BANGALORE RURAL	HOSAKOTE
1145	KARNATAKA	BANGALORE RURAL	MAGADI
1146	KARNATAKA	BANGALORE RURAL	NELAMANGALA
-			

1148	KARNATAKA	BELGAUM	ATHANI
1149	KARNATAKA	BELGAUM	BAILHONGAL
1150	KARNATAKA	BELGAUM	BELGAUM CITY
1151	KARNATAKA	BELGAUM	BELGAUM RURAL.
1152	KARNATAKA	BELGAUM	CHIKODI
1153	KARNATAKA	BELGAUM	HUKKERI
1154	KARNATAKA	BELGAUM	KHANAPUR
1155	KARNATAKA	BELGAUM	NIPPANI
1156	KARNATAKA	BELLARY	HADAGALI
1157	KARNATAKA	BIDAR	BHALKI
1158	KARNATAKA	CHIKMAGALUR	CHIKMAGALUR
1159	KARNATAKA	CHIKMAGALUR	KADUR
1160	KARNATAKA	CHIKMAGALUR	КОРРА
1161	KARNATAKA	CHIKMAGALUR	MOODIGERE
1162	KARNATAKA	CHIKMAGALUR	NARASIMHARAJAPURA
1162	KARNATAKA	CHIKMAGALUR	SRINGERI
1164	KARNATAKA	CHIKMAGALUR	TARIKERE
1164	KARNATAKA	CHIKMAGALUK	
			CHITRADURGA
1166	KARNATAKA	CHITRADURGA	HIRIYUR
1167	KARNATAKA	CHITRADURGA	HOLALKERE
1168	KARNATAKA	CHITRADURGA	HOSADURGA
1169	KARNATAKA	DAKSHINA KANNADA	BANTWAL
1170	KARNATAKA	DAKSHINA KANNADA	BELTHANGADY
1171	KARNATAKA	DAKSHINA KANNADA	MANGALORE
1172	KARNATAKA	DAKSHINA KANNADA	PUTTUR
1173	KARNATAKA	DAKSHINA KANNADA	SULLIA
1174	KARNATAKA	DAVANGERE	CHANNAGIRI
1175	KARNATAKA	DAVANGERE	DAVANAGERE(N)
1176	KARNATAKA	DAVANGERE	DAVANAGERE(S)
1177	KARNATAKA	DAVANGERE	HARIHARA
1178	KARNATAKA	DAVANGERE	HONNALI
1179	KARNATAKA	DAVANGERE	JAGALOOR
1180	KARNATAKA	DHARWAD	HUBLI
1181	KARNATAKA	DHARWAD	KUNDAGOL
1182	KARNATAKA	DHARWAD	NAVALGUND
1183	KARNATAKA	GADAG	GADAG
1184	KARNATAKA	GADAG	NARAGUND
1185	KARNATAKA	GADAG	SHIRHATTI
1186	KARNATAKA	HASSAN	ALUR
1187	KARNATAKA	HASSAN	ARAKALAGUDU
1188	KARNATAKA	HASSAN	ARASIKERE
1189	KARNATAKA	HASSAN	BELUR
1190	KARNATAKA	HASSAN	CHANNARAYAPATNA
1191	KARNATAKA	HASSAN	HASSAN
1192	KARNATAKA	HASSAN	SAKALESHAPURA
1192	KARNATAKA	HASSAN HAVERI	BYADAGI
1193	KARNATAKA	HAVERI	HANAGAL
1195	KARNATAKA	HAVERI	HAVERI
1196	KARNATAKA	HAVERI	HIREKERUR
1197	KARNATAKA	HAVERI	RANNEBENNUR
1198	KARNATAKA	HAVERI	SAVANUR
1199	KARNATAKA	HAVERI	SHIGGOAN
1200	KARNATAKA	KODAGU	MADIKERI

1201	KARNATAKA	KODAGU	SOMAVARPET
1202	KARNATAKA	KODAGU	VIRAJPET
1203	KARNATAKA	KOLAR	CHIKKABALLAPUR
1204	KARNATAKA	KOLAR	KOLAR
1205	KARNATAKA	KOLAR	MALUR
1206	KARNATAKA	KOLAR	SIDLAGHATTA
1200	KARNATAKA	MANDYA	KRISHNARAJA PET
1207	KARNATAKA	MANDYA	MADDUR
1200	KARNATAKA	MANDYA	MANDYA
1210	KARNATAKA	MANDYA	NAGAMANGALA
1210	KARNATAKA	MANDYA	SRIRANGA PATNA
1211	KARNATAKA	MYSORE	MYSORE RURAL
1212	KARNATAKA	MYSORE	PERIYA PATNA
1213	KARNATAKA	SHIMOGA	BHADRAVATI
1214	KARNATAKA	SHIMOGA	HOSANAGAR
1215	KARNATAKA	SHIMOGA	SAGAR
1210	KARNATAKA	SHIMOGA	SHIKARIPUR
1217	KARNATAKA	SHIMOGA	SHIMOGA
1218	KARNATAKA	SHIMOGA	SORAB
1219	KARNATAKA		THIRTHAHALLI
1220		SHIMOGA	CHIKNAYAKANHALLI
1221	KARNATAKA KARNATAKA	TUMKUR	GUBBI
		TUMKUR	
1223	KARNATAKA	TUMKUR	KORATAGERE
1224	KARNATAKA	TUMKUR	KUNIGAL
1225	KARNATAKA	TUMKUR	MADHUGIRI
1226	KARNATAKA	TUMKUR	SIRA
1227	KARNATAKA	TUMKUR	TIPTUR
1228	KARNATAKA	TUMKUR	TUMKUR
1229	KARNATAKA	TUMKUR	TURUVEKERE
1230	KARNATAKA	UDUPI	KARKALA
1231	KARNATAKA	UDUPI	KUNDAPUR
1232	KARNATAKA		UDUPI
1233	KARNATAKA	UTTARA KANNADA	ANKOLA
1234	KARNATAKA	UTTARA KANNADA	BHATKAL
1235	KARNATAKA	UTTARA KANNADA	HALIYAL
1236	KARNATAKA	UTTARA KANNADA	HONNAVAR
1237	KARNATAKA	UTTARA KANNADA	JOIDA
1238	KARNATAKA	UTTARA KANNADA	KARWAR
1239	KARNATAKA	UTTARA KANNADA	KUMTA
1240	KARNATAKA	UTTARA KANNADA	MUNDAGOD
1241	KARNATAKA	UTTARA KANNADA	SIDDAPUR
1242	KARNATAKA	UTTARA KANNADA	SIRSI
1243	KARNATAKA	UTTARA KANNADA	YELLAPUR
1244	KERALA	ALAPPUZHA	ALAPPUZHA
1245	KERALA	ALAPPUZHA	AMBALAPPUZHA
1246	KERALA	ALAPPUZHA	CHENGANNUR
1247	KERALA	ALAPPUZHA	CHERTHALA
1248	KERALA	ALAPPUZHA	HARIPPAD
1249	KERALA	ALAPPUZHA	KAYAMKULAM
1250	KERALA	ALAPPUZHA	MAVELIKKARA
1251	KERALA	ALAPPUZHA	MONCOMPU
1252	KERALA	ALAPPUZHA	THALAVADY
1253	KERALA	ALAPPUZHA	THURAVUR 82

1254	KERALA	ALAPPUZHA	VELIYANADU
1255	KERALA	ERNAKULAM	ALUVA
1256	KERALA	ERNAKULAM	ANGAMALY
1257	KERALA	ERNAKULAM	ERNAKULAM
1258	KERALA	ERNAKULAM	KALLOORKKAD
1259	KERALA	ERNAKULAM	KOLENCHERY
1260	KERALA	ERNAKULAM	KOOTHATTUKULAM
1261	KERALA	ERNAKULAM	KOOVAPPADY
1262	KERALA	ERNAKULAM	KOTHAMANGALAM
1263	KERALA	ERNAKULAM	MATTANCHERY
1264	KERALA	ERNAKULAM	MUVATTUPUZHA
1265	KERALA	ERNAKULAM	N.PARUR
1266	KERALA	ERNAKULAM	PERUMBAVOOR
1267	KERALA	ERNAKULAM	PIRAVOM
1268	KERALA	ERNAKULAM	TRIPPUNITHURA
1269	KERALA	ERNAKULAM	VYPEEN
1270	KERALA	IDUKKI	ADIMALY
1271	KERALA	IDUKKI	ARAKKULAM
1271	KERALA	IDUKKI	KARIMANNOOR
1273	KERALA	IDUKKI	KATTAPPANA
1273	KERALA	IDUKKI	MUNNAR
1271	KERALA	IDUKKI	NEDUMKANDAM
1275	KERALA	IDUKKI	PEERMADU
1270	KERALA	IDUKKI	THODUPUZHA
1277	KERALA	KANNUR	CHOCKLI
1278	KERALA	KANNUR	IRIKKUR
1279	KERALA	KANNUR	IRITTY
1280	KERALA	KANNUR	KANNUR NORTH
1281	KERALA	KANNUR	KANNUR SOUTH
1282	KERALA	KANNUR	KOOTHUPARAMBA
1283	KERALA	KANNUR	MADAI
1284	KERALA	KANNUR	MADAI
1285	KERALA	KANNUR	PANOOR
1280	KERALA	KANNUR	PAPPINISSERI
1287	KERALA	KANNUR	PAYYANNUR
1288	KERALA	KANNUR	TALIPARAMBA NORTH
1289	KERALA	KANNUR	TALIFARAMBA NORTH
1290	KERALA	KANNUR	THALASSERY NORTH
1291	KERALA	KANNUR	THALESSERY SOUTH
1292		KASARAGOD	BEKAL
1293	KERALA		CHERUVATHUR
	KERALA	KASARAGOD	
1295 1296	KERALA KERALA	KASARAGOD KASARAGOD	CHITTARIKKAL HOSDURG
1296	KERALA KERALA	KASARAGOD	KASARAGOD
		KASARAGOD KASARAGOD	
1298 1299	KERALA KERALA		KUMBALA MANJESHWAR
1299	KERALA	KASARAGOD KOLLAM	ANCHEL
1300			CHADAYAMANGALAM
	KERALA	KOLLAM	
1302	KERALA	KOLLAM	CHATHANNOOR
1303	KERALA	KOLLAM	CHAVARA
1304	KERALA	KOLLAM	KARUNAGAPPALLY
1305	KERALA	KOLLAM	KOLLAM
1306	KERALA	KOLLAM	KOTTARAKARA 83

1307	KERALA	KOLLAM	KULAKKADA
1308	KERALA	KOLLAM	KUNDARA
1309	KERALA	KOLLAM	PUNALUR
1310	KERALA	KOLLAM	SASTHAMCOTTA
1311	KERALA	KOLLAM	VELIYAM
1312	KERALA	КОТТАУАМ	CHANGANASSERY
1312	KERALA	КОТТАУАМ	ERATTUPETTA
1313	KERALA	КОТТАУАМ	ETTUMANOOR
1315	KERALA	КОТТАУАМ	KANJIRAPPALLY
1316	KERALA	КОТТАУАМ	KARUKACHAL
1317	KERALA	КОТТАУАМ	KOTTAYAM(EAST)
1318	KERALA	КОТТАУАМ	KOTTAYAM(WEST)
1319	KERALA	КОТТАУАМ	KOZHUVANAL
1319	KERALA	КОТТАУАМ	KURAVILANGADU
1320	KERALA	КОТТАУАМ	PALA
1321	KERALA	КОТТАУАМ	PAMPADY
1322	KERALA	KOTTAYAM	RAMAPURAM
1323	KERALA	KOTTAYAM	VAIKOM
1324	KERALA	KOTTAYAM KOZHIKODE	BALUSSERI
1325	KERALA	KOZHIKODE	CHEVAYOOR
1320	KERALA	KOZHIKODE	CHOMBALA
1327	KERALA	KOZHIKODE	FEROKE
1328	KERALA	KOZHIKODE	KOYILANDY
1329	KERALA	KOZHIKODE	KOTILANDI KOZHIKODE RURAL
1330	KERALA	KOZHIKODE	KUNNAMANGALAM
1332	KERALA	KOZHIKODE	KUNNUMMAL
1333	KERALA	KOZHIKODE	MELADY
1334	KERALA	KOZHIKODE	NADAPURAM
1335 1336	KERALA	KOZHIKODE	PERAMBRA THAMARASSERY
	KERALA	KOZHIKODE	
1337	KERALA	KOZHIKODE	THODANNUR VATAKARA
1338 1339	KERALA KERALA	KOZHIKODE	
		MALAPPURAM	AREACODE
1340	KERALA	MALAPPURAM	EDAPPAL Kondotty
1341	KERALA	MALAPPURAM	
1342	KERALA	MALAPPURAM	KUTTIPURAM
1343	KERALA	MALAPPURAM	MALAPPURAM
1344	KERALA	MALAPPURAM	MANJERI
1345	KERALA	MALAPPURAM	MANKADA
1346	KERALA	MALAPPURAM	NILAMBUR
1347	KERALA	MALAPPURAM	PARAPPANANGADI DEDINITILAL MANINA
1348	KERALA	MALAPPURAM	PERINTHALMANNA
1349	KERALA	MALAPPURAM	PONNANI
1350	KERALA	MALAPPURAM	TANUR
1351	KERALA	MALAPPURAM	TIRUR
1352	KERALA	MALAPPURAM	VENGARA
1353	KERALA	MALAPPURAM	WANDOOR
1354	KERALA	PALAKKAD	AGALI
1355	KERALA	PALAKKAD	ALATHUR
1356	KERALA	PALAKKAD	CHERPULASSERY
1357	KERALA	PALAKKAD	CHITTUR
1358	KERALA	PALAKKAD	KOLLENGODE
1359	KERALA	PALAKKAD	KUZHALMANNAM 84

1360	KERALA	PALAKKAD	MANNNARKKAD
1361	KERALA	PALAKKAD	OTTAPALAM
1362	KERALA	PALAKKAD	PALAKKAD
1363	KERALA	PALAKKAD	PARLI
1364	KERALA	PALAKKAD	PATTAMBI
1365	KERALA	PALAKKAD	SHORANUR
1366	KERALA	PALAKKAD	THRITHALA
1367	KERALA	PATHANAMTHITTA	ADOOR
1368	KERALA	PATHANAMTHITTA	ARANMULA
1369	KERALA	PATHANAMTHITTA	KONNI
1309	KERALA	PATHANAMTHITTA	KOTHN
1370	KERALA	PATHANAMTHITTA	MALLAPALLY
1371	KERALA	PATHANAMTHITTA	PANDALAM
1372			
	KERALA	PATHANAMTHITTA	PULLAD
1374	KERALA	PATHANAMTHITTA	RANNI
1375	KERALA	PATHANAMTHITTA	THIRUVALLA
1376	KERALA	THIRUVANANTHAPURAM	ATTINGAL
1377	KERALA	THIRUVANANTHAPURAM	BALARAMAPURAM
1378	KERALA	THIRUVANANTHAPURAM	KANIYAPURAM
1379	KERALA	THIRUVANANTHAPURAM	KATTAKKADA
1380	KERALA	THIRUVANANTHAPURAM	KILIMANOOR
1381	KERALA	THIRUVANANTHAPURAM	NEDUMANGADU
1382	KERALA	THIRUVANANTHAPURAM	NEYYATTINKARA
1383	KERALA	THIRUVANANTHAPURAM	PALODE
1384	KERALA	THIRUVANANTHAPURAM	PARASSALA
1385	KERALA	THIRUVANANTHAPURAM	TRIVANDRUM NORTH
1386	KERALA	THIRUVANANTHAPURAM	TRIVANDRUM SOUTH
1387	KERALA	THIRUVANANTHAPURAM	VARKALA
1388	KERALA	THRISSUR	ANTHIKKAD
1389	KERALA	THRISSUR	CHAVAKKAD
1390	KERALA	THRISSUR	CHERPU
1391	KERALA	THRISSUR	CHOWANNUR
1392	KERALA	THRISSUR	ERIYAD (KODUNGALLUR)
1393	KERALA	THRISSUR	IRINJALAKUDA
1394	KERALA	THRISSUR	KODAKARA
1395	KERALA	THRISSUR	MALA
1396	KERALA	THRISSUR	MATHILAKAM
1397	KERALA	THRISSUR	MULLASSERY
1398	KERALA	THRISSUR	OLLUKKARA
1399	KERALA	THRISSUR	PAZHAYANNUR
1400	KERALA	THRISSUR	PUZHAKKAL
1401	KERALA	THRISSUR	THALIKULAM
1402	KERALA	THRISSUR	VELLANGALLUR
1403	KERALA	THRISSUR	WADAKKANCHERY
1404	KERALA	WAYANAD	MANATHAVADY
1404	KERALA	WAYANAD	SULTHAN BATTERY
1405	KERALA	WAYANAD	VYTHIRI
1400	LAKSHADWEEP	LAKSHADWEEP	AGATI
1407	LAKSHADWEEP	LAKSHADWEEP	AMINI
1408	LAKSHADWEEP	LAKSHADWEEP	ANDROTT
1409	LAKSHADWEEP	LAKSHADWEEP	CHETLATH
1410	LAKSHADWEEP		KADMAT
-		LAKSHADWEEP	
1412	LAKSHADWEEP	LAKSHADWEEP	KALPENI 85

1413	LAKSHADWEEP	LAKSHADWEEP	KAVARATTI
1414	LAKSHADWEEP	LAKSHADWEEP	KILTAN
1415	MADHYA PRADESH	BALAGHAT	BALAGHAT
1416	MADHYA PRADESH	BALAGHAT	KATANGI
1417	MADHYA PRADESH	BALAGHAT	KHAIRLANJI
1418	MADHYA PRADESH	BALAGHAT	KIRNAPUR
1419	MADHYA PRADESH	BALAGHAT	LAL BURRA
141)	MADHYA PRADESH	BALAGHAT	LANJI
1420	MADHYA PRADESH	BALAGHAT	PARASWADA
1421	MADHYA PRADESH	BALAGHAT	WARA SEONI
1423	MADHYA PRADESH	BARWANI	THIKIRI
1423	MADHYA PRADESH	BETUL	AMLA
1424	MADHYA PRADESH	BETUL	ATHNER
1425	MADHYA PRADESH	BETUL	BETUL
1420	MADHYA PRADESH	BETUL	BHAINSDEHI
1427	MADHYA PRADESH	BETUL	CHICHOLI
	MADHYA PRADESH	BETUL	
1429			MULTAI
1430	MADHYA PRADESH	BETUL	PRABHAT PATTAN
1431	MADHYA PRADESH	BHIND	ATER
1432	MADHYA PRADESH	BHIND	BHIND
1433	MADHYA PRADESH	BHIND	LAHAR
1434	MADHYA PRADESH	BHIND	MEHGAON
1435	MADHYA PRADESH	BHIND	ROUN
1436	MADHYA PRADESH	CHHINDWARA	BICHHUA
1437	MADHYA PRADESH	CHHINDWARA	CHHINDWARA
1438	MADHYA PRADESH	CHHINDWARA	CHOURAI
1439	MADHYA PRADESH	CHHINDWARA	MOHKHED
1440	MADHYA PRADESH	CHHINDWARA	PANDHURNA
1441	MADHYA PRADESH	CHHINDWARA	SAUSAR
1442	MADHYA PRADESH	DATIA	BHANDER
1443	MADHYA PRADESH	DATIA	DATIA
1444	MADHYA PRADESH	DATIA	SEONDHA
1445	MADHYA PRADESH	HARDA	HARDA
1446	MADHYA PRADESH	HARDA	TIMARNI
1447	MADHYA PRADESH	HOSHANGABAD	HOSHANGABAD
1448	MADHYA PRADESH	HOSHANGABAD	KESLA
1449	MADHYA PRADESH	HOSHANGABAD	PIPERIYA
1450	MADHYA PRADESH	HOSHANGABAD	SIVNI MAALVA
1451	MADHYA PRADESH	JABALPUR	JABALPUR (RURAL)
1452	MADHYA PRADESH	JABALPUR	MAJHOLI
1453	MADHYA PRADESH	JABALPUR	PANAGAR
1454	MADHYA PRADESH	JABALPUR	PATAN
1455	MADHYA PRADESH	JABALPUR	SIHORA
1456	MADHYA PRADESH	KHANDWA	CHHAIGAONMAKHAN
1457	MADHYA PRADESH	KHANDWA	KHANDWA
1458	MADHYA PRADESH	KHANDWA	PUNASA
1459	MADHYA PRADESH	KHARGONE	BADWAHA
1460	MADHYA PRADESH	KHARGONE	BHIKAN GOAN
1461	MADHYA PRADESH	KHARGONE	GOGAWAN
1462	MADHYA PRADESH	KHARGONE	KASRAVAD
1463	MADHYA PRADESH	KHARGONE	KHARGONE
1464	MADHYA PRADESH	KHARGONE	MAHESHWAR
1465	MADHYA PRADESH	KHARGONE	SEGAON
MUDD E	ducation rfa/06.07.2012		86

1466	MADHYA PRADESH	MANDLA	MANDLA
1467	MADHYA PRADESH	MANDLA	NAINPUR
1468	MADHYA PRADESH	MANDSAUR	MALHARGARH
1469	MADHYA PRADESH	MANDSAUR	MANDSAUR
1470	MADHYA PRADESH	MANDSAUR	SITAMAU
1471	MADHYA PRADESH	MORENA	AMBAH
1472	MADHYA PRADESH	MORENA	PORSA
1472	MADHYA PRADESH	NARSIMHAPUR	CHAWARPATHA
1474	MADHYA PRADESH	NARSIMHAPUR	CHICHLI
1475	MADHYA PRADESH	NARSIMHAPUR	GOATEGAON {SRIDHAM}
1475	MADHYA PRADESH	NARSIMHAPUR	KARELI
1477	MADHYA PRADESH	NARSIMHAPUR	NARSINGHPUR
1478	MADHYA PRADESH	NARSIMHAPUR	SAIKHEDA
1479	MADHYA PRADESH	PANNA	GUNOUR
1475	MADHYA PRADESH	PANNA	SHAHNAGAR
1480	MADHYA PRADESH	RAISEN	BADI
1481	MADHYA PRADESH	RAISEN	BEGAMGANJ
1482	MADHYA PRADESH	RAISEN	GAIRATGANJ
1483	MADHYA PRADESH	RAISEN	OBEDULLAGANJ
1484	MADHYA PRADESH MADHYA PRADESH	RAISEN	SANCHI
1486	MADHYA PRADESH	RAISEN	SILWANI
1487	MADHYA PRADESH	RAISEN	UDAIPURA
1488	MADHYA PRADESH	RATLAM	ALOT
1489	MADHYA PRADESH	RATLAM	JAORA
1490	MADHYA PRADESH	RATLAM	PIPLODA
1491	MADHYA PRADESH	RATLAM	RATLAM
1492	MADHYA PRADESH	REWA	RAIPUR(K)
1493	MADHYA PRADESH	REWA	REWA
1494	MADHYA PRADESH	REWA	SIRMOUR
1495	MADHYA PRADESH	SAGAR	DEORI
1496	MADHYA PRADESH	SAGAR	JAISINAGAR
1497	MADHYA PRADESH	SAGAR	KESLI
1498	MADHYA PRADESH	SAGAR	RAHATGARH
1499	MADHYA PRADESH	SAGAR	REHLI
1500	MADHYA PRADESH	SAGAR	SAGAR
1501	MADHYA PRADESH	SATNA	AMARPATAN
1502	MADHYA PRADESH	SATNA	NAGOD
1503	MADHYA PRADESH	SATNA	RAMPUR BAGHELAN
1504	MADHYA PRADESH	SATNA	SOHAWAL
1505	MADHYA PRADESH	SEHORE	BUDHANI
1506	MADHYA PRADESH	SEHORE	NASRULLAGANJ
1507	MADHYA PRADESH	SEONI	BARGHAT
1508	MADHYA PRADESH	SEONI	CHHAPARA
1509	MADHYA PRADESH	SEONI	DHANORA
1510	MADHYA PRADESH	SEONI	GHANSORE
1511	MADHYA PRADESH	SEONI	KEOLARI
1512	MADHYA PRADESH	SEONI	KURAI
1513	MADHYA PRADESH	SEONI	SEONI
1514	MADHYA PRADESH	SHAJAPUR	KALAPIPAL
1515	MADHYA PRADESH	SHAJAPUR	M. BARODIYA
1516	MADHYA PRADESH	SHAJAPUR	NALKHEDA
1517	MADHYA PRADESH	SHAJAPUR	SHAJAPUR
1518	MADHYA PRADESH	SHAJAPUR	SHUJALPUR
MUDD E	ducation rfa/06.07.2012		87

1519	MADHYA PRADESH	SHAJAPUR	SUSNER
1520	MADHYA PRADESH	SHIVPURI	NARVAR
1521	MADHYA PRADESH	UJJAIN	BARNAGAR
1522	MADHYA PRADESH	UJJAIN	TARANA
1523	MADHYA PRADESH	UMARIA	MANPUR
1524	MADHYA PRADESH	VIDISHA	GYARASPUR
1525	MADHYA PRADESH	VIDISHA	KURWAI
1526	MADHYA PRADESH	VIDISHA	VIDISHA
1520	MAHARASHTRA	AHMADNAGAR	AKOLE
1527	MAHARASHTRA	AHMADNAGAR	JAMKHED
1520	MAHARASHTRA	AHMADNAGAR	KARJAT
1529	MAHARASHTRA	AHMADNAGAR	KOPARGAON
1530	MAHARASHTRA	AHMADNAGAR	NAGAR
1531	MAHARASHTRA	AHMADNAGAR	NEWASA
1532	MAHARASHTRA	AHMADNAGAR	PARNER
1533	MAHARASHTRA	AHMADNAGAR	PATHARDI
1535	MAHARASHTRA	AHMADNAGAR	RAHATA
1536	MAHARASHTRA	AHMADNAGAR	RAHURI
1537	MAHARASHTRA	AHMADNAGAR	SANGAMNER
1538	MAHARASHTRA	AHMADNAGAR	SHEVGAON
1539	MAHARASHTRA	AHMADNAGAR	SHRIGONDA
1540	MAHARASHTRA	AHMADNAGAR	SHRIRAMPOOR
1541	MAHARASHTRA	AKOLA	AKOLA PS
1542	MAHARASHTRA	AKOLA	АКОТ
1543	MAHARASHTRA	AKOLA	BALAPUR
1544	MAHARASHTRA	AKOLA	BARSHITAKALI
1545	MAHARASHTRA	AKOLA	MURTIZAPUR
1546	MAHARASHTRA	AKOLA	PATUR
1547	MAHARASHTRA	AKOLA	TELHARA
1548	MAHARASHTRA	AMRAVATI	ACHALPUR
1549	MAHARASHTRA	AMRAVATI	AMRAVATI
1550	MAHARASHTRA	AMRAVATI	ANJANGAON SURJI
1551	MAHARASHTRA	AMRAVATI	BHATKULI
1552	MAHARASHTRA	AMRAVATI	CHANDUR BAZAR
1553	MAHARASHTRA	AMRAVATI	CHANDUR RAILWAY
1554	MAHARASHTRA	AMRAVATI	CHIKHALDARA
1555	MAHARASHTRA	AMRAVATI	DARYAPUR
1556	MAHARASHTRA	AMRAVATI	DHAMANGAON RLY
1557	MAHARASHTRA	AMRAVATI	DHARNI
1558	MAHARASHTRA	AMRAVATI	MORSHI
1559	MAHARASHTRA	AMRAVATI	MUNICAPAL CORP. AMRAVATI
1560	MAHARASHTRA	AMRAVATI	NANDGAON KH.
1561	MAHARASHTRA	AMRAVATI	TIOSA
1562	MAHARASHTRA	AMRAVATI	WARUD
1563	MAHARASHTRA	AURANGABAD (MAHARASHTRA)	AURANGABAD
1564	MAHARASHTRA	AURANGABAD (MAHARASHTRA)	GANGAPUR
1565	MAHARASHTRA	AURANGABAD (MAHARASHTRA)	KANNAD
1566	MAHARASHTRA	AURANGABAD (MAHARASHTRA)	KHULTABAD

1568         MAHARASHTRA         AURANGABAD (MAHARASHTRA)         PHULAMBRI           1569         MAHARASHTRA         AURANGABAD (MAHARASHTRA)         SILLOD           1570         MAHARASHTRA         AURANGABAD (MAHARASHTRA)         SOEGAON           1571         MAHARASHTRA         AURANGABAD (MAHARASHTRA)         SOEGAON           1571         MAHARASHTRA         AURANGABAD (MAHARASHTRA)         VAIJAPUR           1572         MAHARASHTRA         BHANDARA         LAKHANDUR           1573         MAHARASHTRA         BHANDARA         LAKHANUR           1574         MAHARASHTRA         BHANDARA         LAKHANUR           1575         MAHARASHTRA         BHANDARA         LAKHANUR           1576         MAHARASHTRA         BHANDARA         SAKOLI           1577         MAHARASHTRA         BHANDARA         SAKOLI           1578         MAHARASHTRA         BID         AMBAJOGAI           1579         MAHARASHTRA         BID         ABHAJOGAI           1581         MAHARASHTRA         BID         ABED           1581         MAHARASHTRA         BID         BAHAJOGAI           1581         MAHARASHTRA         BID         PATODA           1584         MAH	1567	MAHARASHTRA	AURANGABAD (MAHARASHTRA)	PAITHAN
1569         MAHARASHIRA         (MAHARASHTRA)         SILLOD           1570         MAHARASHIRA         (MAHARASHTRA)         SOEGAON           1571         MAHARASHIRA         AURANGABAD         (MAHARASHIRA)         SOEGAON           1571         MAHARASHIRA         AURANGABAD         VAIJAPUR           1572         MAHARASHIRA         BHANDARA         BHANDARA           1573         MAHARASHIRA         BHANDARA         LAKHANDUR           1574         MAHARASHIRA         BHANDARA         LAKHANDUR           1575         MAHARASHIRA         BHANDARA         MOHADI           1576         MAHARASHIRA         BHANDARA         SAKOLI           1577         MAHARASHIRA         BID         ASKOLI           1578         MAHARASHIRA         BID         ASHII           1580         MAHARASHIRA         BID         ASHII           1581         MAHARASHIRA         BID         MAJALGON           1582         MAHARASHIRA         BID         MAJALGON           1584         MAHARASHIRA         BID         PAIU           1585         MAHARASHIRA         BID         PAIU           1586         MAHARASHIRA         BID         PAIUA	1568	MAHARASHTRA		PHULAMBRI
1570     MAHARASHIRA     (MAHARASHIRA)     SOEGAON       1571     MAHARASHIRA     AURANGABAD     VAIJAPUR       1572     MAHARASHIRA     BHANDARA     BHANDARA       1573     MAHARASHIRA     BHANDARA     LAKHANI       1574     MAHARASHIRA     BHANDARA     LAKHANI       1575     MAHARASHIRA     BHANDARA     LAKHANI       1576     MAHARASHIRA     BHANDARA     MOHADI       1577     MAHARASHIRA     BHANDARA     MOHADI       1578     MAHARASHIRA     BHANDARA     SAKOLI       1579     MAHARASHIRA     BID     AMBAJOGAI       1580     MAHARASHIRA     BID     AMBAJOGAI       1581     MAHARASHIRA     BID     KAIJ       1582     MAHARASHIRA     BID     KAIJ       1583     MAHARASHIRA     BID     PATODA       1584     MAHARASHIRA     BID     PARLI       1585     MAHARASHIRA     BID     PARLI       1586     MAHARASHIRA     BID     PARLI       1587     MAHARASHIRA     BUD     PARLI       1588     MAHARASHIRA     BUD     PARLI       1589     MAHARASHIRA     BUDANA     CHIKHALI       1580     MAHARASHIRA     BUDANA     GINA	1569	MAHARASHTRA		SILLOD
1571       MAHARASHIRA       (MAHARASHIRA)       VADAPUR         1572       MAHARASHIRA       BHANDARA       BHANDARA         1573       MAHARASHIRA       BHANDARA       LAKHANDUR         1574       MAHARASHIRA       BHANDARA       LAKHANDUR         1575       MAHARASHIRA       BHANDARA       MOHADI         1576       MAHARASHIRA       BHANDARA       MOHADI         1577       MAHARASHIRA       BHANDARA       PAUNI         1577       MAHARASHIRA       BHANDARA       PAUNI         1577       MAHARASHIRA       BHANDARA       TUMSAR         1579       MAHARASHIRA       BHANDARA       TUMSAR         1579       MAHARASHIRA       BID       AMBAJOGAI         1580       MAHARASHIRA       BID       AMBAJOGAI         1581       MAHARASHIRA       BID       KAIJ         1581       MAHARASHIRA       BID       MAIARASHIRA         1582       MAHARASHIRA       BID       MAIARASHIRA         1583       MAHARASHIRA       BID       PARLI         1584       MAHARASHIRA       BUDANA       CHIKHALI         1585       MAHARASHIRA       BULDANA       DEULGANA         158	1570	MAHARASHTRA		SOEGAON
1573     MAHARASHTRA     BHANDARA     LAKHANDUR       1574     MAHARASHTRA     BHANDARA     LAKHANI       1575     MAHARASHTRA     BHANDARA     MOHADI       1576     MAHARASHTRA     BHANDARA     PAUNI       1577     MAHARASHTRA     BHANDARA     PAUNI       1577     MAHARASHTRA     BHANDARA     SAKOLI       1578     MAHARASHTRA     BID     AMBAJOGAI       1579     MAHARASHTRA     BID     AMBAJOGAI       1580     MAHARASHTRA     BID     ASHTI       1581     MAHARASHTRA     BID     MAJALGAON       1582     MAHARASHTRA     BID     PARLI       1583     MAHARASHTRA     BID     PARLI       1584     MAHARASHTRA     BID     PARLI       1585     MAHARASHTRA     BID     PARLI       1586     MAHARASHTRA     BUDANA     BULDANA       1587     MAHARASHTRA     BULDANA     BULDANA       1588     MAHARASHTRA     BULDANA     DEULGAON RAJA       1589     MAHARASHTRA     BULDANA     DEULGAON RAJA       1590     MAHARASHTRA     BULDANA     LONAR       1591     MAHARASHTRA     BULDANA     KHAMGON       1592     MAHARASHTRA     BULDANA <td>1571</td> <td>MAHARASHTRA</td> <td></td> <td>VAIJAPUR</td>	1571	MAHARASHTRA		VAIJAPUR
1574     MAHARASHTRA     BHANDARA     LAKHANI       1575     MAHARASHTRA     BHANDARA     MOHADI       1576     MAHARASHTRA     BHANDARA     PAUNI       1577     MAHARASHTRA     BHANDARA     SAKOLI       1577     MAHARASHTRA     BHANDARA     SAKOLI       1578     MAHARASHTRA     BHANDARA     TUMSAR       1579     MAHARASHTRA     BID     AMBAJOGAI       1580     MAHARASHTRA     BID     ASITI       1581     MAHARASHTRA     BID     MAJALGAON       1582     MAHARASHTRA     BID     MAJALGAON       1584     MAHARASHTRA     BID     PARLI       1585     MAHARASHTRA     BID     PARLI       1586     MAHARASHTRA     BID     SIRUR       1587     MAHARASHTRA     BUDANA     BULDANA       1588     MAHARASHTRA     BULDANA     BULDANA       1588     MAHARASHTRA     BULDANA     DEULGAON RAJA       1589     MAHARASHTRA     BULDANA     DEULGAON RAJA       1590     MAHARASHTRA     BULDANA     JALGAON       1591     MAHARASHTRA     BULDANA     JALGAON       1592     MAHARASHTRA     BULDANA     MAHARASHTRA       1593     MAHARASHTRA     B	1572	MAHARASHTRA	BHANDARA	BHANDARA
1575     MAHARASHTRA     BHANDARA     MOHADI       1576     MAHARASHTRA     BHANDARA     PAUNI       1577     MAHARASHTRA     BHANDARA     SAKOLI       1578     MAHARASHTRA     BHANDARA     TUMSAR       1579     MAHARASHTRA     BID     AMBAJOGAI       1580     MAHARASHTRA     BID     ASHTI       1581     MAHARASHTRA     BID     BEED       1582     MAHARASHTRA     BID     MAJALGAON       1584     MAHARASHTRA     BID     PARLI       1585     MAHARASHTRA     BID     PARLI       1586     MAHARASHTRA     BID     PATODA       1586     MAHARASHTRA     BID     PATODA       1586     MAHARASHTRA     BULDANA     CHIKHALI       1587     MAHARASHTRA     BULDANA     DEULGAON RAJA       1588     MAHARASHTRA     BULDANA     DEULGAON RAJA       1589     MAHARASHTRA     BULDANA     DEULGAON RAJA       1589     MAHARASHTRA     BULDANA     JALGAON       1590     MAHARASHTRA     BULDANA     MAHARASHTRA       1591     MAHARASHTRA     BULDANA     MAHARASHTRA       1592     MAHARASHTRA     BULDANA     MAHARASHTRA       1594     MAHARASHTRA	1573	MAHARASHTRA	BHANDARA	LAKHANDUR
1576       MAHARASHTRA       BHANDARA       PAUNI         1577       MAHARASHTRA       BHANDARA       SAKOLI         1578       MAHARASHTRA       BHANDARA       TUMSAR         1579       MAHARASHTRA       BID       AMBAJOGAI         1580       MAHARASHTRA       BID       ASHTI         1580       MAHARASHTRA       BID       ASHTI         1581       MAHARASHTRA       BID       KAJJ         1582       MAHARASHTRA       BID       MAJALGAON         1584       MAHARASHTRA       BID       PATODA         1584       MAHARASHTRA       BID       PATODA         1584       MAHARASHTRA       BID       PATODA         1586       MAHARASHTRA       BULDANA       BULDANA         1587       MAHARASHTRA       BULDANA       CHIKHALI         1588       MAHARASHTRA       BULDANA       DEULGAON RAJA         1590       MAHARASHTRA       BULDANA       JALGAON JAMOD         1591       MAHARASHTRA       BULDANA       JALGAON JAMOD         1592       MAHARASHTRA       BULDANA       MALAKAPUR         1593       MAHARASHTRA       BULDANA       MALAKAPUR         1594	1574	MAHARASHTRA	BHANDARA	LAKHANI
1577MAHARASHTRABHANDARASAKOLI1578MAHARASHTRABIDAMBAJOGAI1579MAHARASHTRABIDASHTI1580MAHARASHTRABIDASHTI1581MAHARASHTRABIDBED1582MAHARASHTRABIDKAIJ1583MAHARASHTRABIDMAJAGAON1584MAHARASHTRABIDPARLI1585MAHARASHTRABIDPATODA1586MAHARASHTRABIDSIRUR1587MAHARASHTRABUDANACHIKHALI1588MAHARASHTRABULDANABULDANA1589MAHARASHTRABULDANACHIKHALI1589MAHARASHTRABULDANACHIKHALI1589MAHARASHTRABULDANAJALGAON JAMOD1591MAHARASHTRABULDANALONAR1592MAHARASHTRABULDANALONAR1593MAHARASHTRABULDANAMALAKAPUR1594MAHARASHTRABULDANAMATALA1595MAHARASHTRABULDANAMANDURA1597MAHARASHTRABULDANASANGRAMPUR1598MAHARASHTRABULDANASINDKHED RAJA1600MAHARASHTRABULDANASINDKHED RAJA1601MAHARASHTRACHANDRAPURBALLARPUR1601MAHARASHTRACHANDRAPURBALLARPUR1601MAHARASHTRACHANDRAPURGONDPIPARI1602MAHARASHTRACHANDRAPURCHANDRAPUR1604MAHARASHTRACHANDR	1575	MAHARASHTRA	BHANDARA	MOHADI
1578MAHARASHTRABHANDARATUMSAR1579MAHARASHTRABIDAMBAJOGAI1580MAHARASHTRABIDASHTI1581MAHARASHTRABIDBEED1582MAHARASHTRABIDMAJJLGAON1584MAHARASHTRABIDPARLI1585MAHARASHTRABIDPARLI1585MAHARASHTRABIDPARLI1585MAHARASHTRABIDPARLI1585MAHARASHTRABIDPATODA1586MAHARASHTRABULDANABULDANA1587MAHARASHTRABULDANABULDANA1588MAHARASHTRABULDANADEULGAON RAJA1589MAHARASHTRABULDANADEULGAON RAJA1590MAHARASHTRABULDANALONAR1592MAHARASHTRABULDANALONAR1593MAHARASHTRABULDANAMALAKAPUR1594MAHARASHTRABULDANAMALAKAR1595MAHARASHTRABULDANAMOTALA1596MAHARASHTRABULDANAMATAA1597MAHARASHTRABULDANASANGRAMPUR1598MAHARASHTRABULDANASINDKHED RAJA1600MAHARASHTRABULDANASINDKHED RAJA1601MAHARASHTRABULDANASINDKHED RAJA1601MAHARASHTRACHANDRAPUR1602MAHARASHTRACHANDRAPUR1604MAHARASHTRACHANDRAPUR1605MAHARASHTRACHANDRAPUR1606MAHARASHTRA <td>1576</td> <td>MAHARASHTRA</td> <td>BHANDARA</td> <td>PAUNI</td>	1576	MAHARASHTRA	BHANDARA	PAUNI
1579MAHARASHTRABIDAMBAJOGAI1580MAHARASHTRABIDASHTI1581MAHARASHTRABIDBED1582MAHARASHTRABIDKAIJ1583MAHARASHTRABIDMAJALGAON1584MAHARASHTRABIDPARLI1585MAHARASHTRABIDPARLI1586MAHARASHTRABIDSIRUR1587MAHARASHTRABUDSIRUR1588MAHARASHTRABULDANACHIKHALI1589MAHARASHTRABULDANACHIKHALI1590MAHARASHTRABULDANAJALGAON RAJA1591MAHARASHTRABULDANALONAR1592MAHARASHTRABULDANALONAR1593MAHARASHTRABULDANAMALAKAPUR1594MAHARASHTRABULDANAMALAKAPUR1595MAHARASHTRABULDANAMALAKAPUR1596MAHARASHTRABULDANAMATAA1596MAHARASHTRABULDANASANGRAMPUR1597MAHARASHTRABULDANASANGRAMPUR1599MAHARASHTRABULDANASANGRAMATI1600MAHARASHTRABULDANASANGRAMPUR1599MAHARASHTRABULDANASANGRAMPUR1599MAHARASHTRABULDANASANGRAMPUR1599MAHARASHTRABULDANASANGRAMPUR1600MAHARASHTRACHANDRAPURBALARPUR1601MAHARASHTRACHANDRAPURBALARPUR1602MAHARASHTRACHANDRAPUR </td <td>1577</td> <td>MAHARASHTRA</td> <td>BHANDARA</td> <td>SAKOLI</td>	1577	MAHARASHTRA	BHANDARA	SAKOLI
1580MAHARASHTRABIDASHTI1581MAHARASHTRABIDBEED1582MAHARASHTRABIDKAIJ1583MAHARASHTRABIDPARLI1584MAHARASHTRABIDPARLI1585MAHARASHTRABIDPATODA1586MAHARASHTRABIDSIRUR1587MAHARASHTRABULDANABULDANA1588MAHARASHTRABULDANACHIKHALI1589MAHARASHTRABULDANACHIKHALI1589MAHARASHTRABULDANAJALGAON JAMOD1590MAHARASHTRABULDANAJALGAON JAMOD1591MAHARASHTRABULDANALONAR1592MAHARASHTRABULDANAMALAKAPUR1593MAHARASHTRABULDANAMALAKAPUR1594MAHARASHTRABULDANAMALAKAPUR1595MAHARASHTRABULDANANANDURA1596MAHARASHTRABULDANASANGRAMPUR1597MAHARASHTRABULDANASANGRAMPUR1598MAHARASHTRABULDANASANGRAMPUR1599MAHARASHTRABULDANASINDKHED RAJA1600MAHARASHTRACHANDRAPURBHADRAWATI1601MAHARASHTRACHANDRAPURBHADRAWATI1602MAHARASHTRACHANDRAPURGONDPIPARI1604MAHARASHTRACHANDRAPURGONDPIPARI1605MAHARASHTRACHANDRAPURGONDPIARI1606MAHARASHTRACHANDRAPURGONDPIARI1607 <td>1578</td> <td>MAHARASHTRA</td> <td>BHANDARA</td> <td>TUMSAR</td>	1578	MAHARASHTRA	BHANDARA	TUMSAR
1581MAHARASHTRABIDBEED1582MAHARASHTRABIDKAIJ1583MAHARASHTRABIDMAJALGAON1584MAHARASHTRABIDPARLI1585MAHARASHTRABIDPATODA1586MAHARASHTRABIDSIRUR1587MAHARASHTRABULDANABULDANA1588MAHARASHTRABULDANACHIKHALI1589MAHARASHTRABULDANADEULGAON RAJA1590MAHARASHTRABULDANADEULGAON RAJA1590MAHARASHTRABULDANALONAR1591MAHARASHTRABULDANAKHAMGAON1592MAHARASHTRABULDANALONAR1593MAHARASHTRABULDANAMALAKAPUR1594MAHARASHTRABULDANAMALAKAPUR1595MAHARASHTRABULDANAMALAKAPUR1596MAHARASHTRABULDANASANGRAMPUR1597MAHARASHTRABULDANASINDKHED RAJA1598MAHARASHTRABULDANASINDKHED RAJA1600MAHARASHTRABULDANASINDKHED RAJA1601MAHARASHTRACHANDRAPURBALARAPUR1601MAHARASHTRACHANDRAPURBALARAPUR1601MAHARASHTRACHANDRAPURBHADRAWATI1603MAHARASHTRACHANDRAPURBHADRAWATI1604MAHARASHTRACHANDRAPURGONDPIPARI1605MAHARASHTRACHANDRAPURGONDPIPARI1606MAHARASHTRACHANDRAPURGONDPIPARI<	1579	MAHARASHTRA	BID	AMBAJOGAI
1582MAHARASHTRABIDKAIJ1583MAHARASHTRABIDPARUI1584MAHARASHTRABIDPARUI1585MAHARASHTRABIDPATODA1586MAHARASHTRABIDSIRUR1587MAHARASHTRABULDANABULDANA1588MAHARASHTRABULDANACHIKHALI1589MAHARASHTRABULDANACHIKHALI1589MAHARASHTRABULDANADEULGAON RAJA1590MAHARASHTRABULDANAJALGAON JAMOD1591MAHARASHTRABULDANAKHAMGAON1592MAHARASHTRABULDANAKHAMGAON1593MAHARASHTRABULDANAMALAKAPUR1594MAHARASHTRABULDANAMOTALA1595MAHARASHTRABULDANAMOTALA1596MAHARASHTRABULDANAMOTALA1597MAHARASHTRABULDANASANGRAMPUR1598MAHARASHTRABULDANASINDKHED RAJA1600MAHARASHTRABULDANASINDKHED RAJA1601MAHARASHTRACHANDRAPURBALLARPUR1601MAHARASHTRACHANDRAPURBALLARPUR1602MAHARASHTRACHANDRAPURBALARPUR1603MAHARASHTRACHANDRAPURGONDPIPARI1604MAHARASHTRACHANDRAPURGONDPIPARI1605MAHARASHTRACHANDRAPURGONDPIPARI1606MAHARASHTRACHANDRAPURGONDPIPARI1607MAHARASHTRACHANDRAPURMUL <tr< td=""><td>1580</td><td>MAHARASHTRA</td><td>BID</td><td>ASHTI</td></tr<>	1580	MAHARASHTRA	BID	ASHTI
1583MAHARASHTRABIDMAJALGAON1584MAHARASHTRABIDPARLI1585MAHARASHTRABIDPATDDA1586MAHARASHTRABIDSIRUR1587MAHARASHTRABUDDANABULDANA1588MAHARASHTRABULDANABULDANA1589MAHARASHTRABULDANADEULGAON RAJA1590MAHARASHTRABULDANAJALGAON JAMOD1591MAHARASHTRABULDANAJALGAON JAMOD1592MAHARASHTRABULDANALONAR1593MAHARASHTRABULDANAMALAKAPUR1594MAHARASHTRABULDANAMAHARASHTRA1595MAHARASHTRABULDANAMATAKAR1596MAHARASHTRABULDANAMATAKAR1597MAHARASHTRABULDANANANDURA1598MAHARASHTRABULDANASANGRAMPUR1599MAHARASHTRABULDANASINGKAMPUR1599MAHARASHTRABULDANASINGKAMPUR1600MAHARASHTRABULDANASINGKHED RAJA1601MAHARASHTRACHANDRAPURBALLARPUR1602MAHARASHTRACHANDRAPURBALARPUR1603MAHARASHTRACHANDRAPURGONDPIPARI1604MAHARASHTRACHANDRAPURGONDPIPARI1605MAHARASHTRACHANDRAPURGONDPIPARI1606MAHARASHTRACHANDRAPURGONDPIPARI1607MAHARASHTRACHANDRAPURGONDPIPARI1608MAHARASHTRACHANDRAPUR<	1581	MAHARASHTRA	BID	BEED
1584MAHARASHTRABIDPARLI1585MAHARASHTRABIDPATODA1586MAHARASHTRABIDSIRUR1587MAHARASHTRABULDANABULDANA1588MAHARASHTRABULDANACHIKHALI1589MAHARASHTRABULDANACHIKHALI1589MAHARASHTRABULDANADEULGAON RAJA1590MAHARASHTRABULDANAJALGAON JAMOD1591MAHARASHTRABULDANALONAR1592MAHARASHTRABULDANALONAR1594MAHARASHTRABULDANAMALAKAPUR1595MAHARASHTRABULDANAMALAKAPUR1594MAHARASHTRABULDANAMALAKAPUR1595MAHARASHTRABULDANAMALAKAR1596MAHARASHTRABULDANAMANDURA1597MAHARASHTRABULDANASANGRAMPUR1598MAHARASHTRABULDANASINDKHED RAJA1599MAHARASHTRABULDANASINDKHED RAJA1600MAHARASHTRACHANDRAPURBALLARPUR1601MAHARASHTRACHANDRAPURBALARPUR1602MAHARASHTRACHANDRAPURGONDPIPARI1603MAHARASHTRACHANDRAPURGONDPIPARI1604MAHARASHTRACHANDRAPURGONDPIPARI1606MAHARASHTRACHANDRAPURMUL1607MAHARASHTRACHANDRAPURMOL1608MAHARASHTRACHANDRAPURMOL1609MAHARASHTRACHANDRAPURNAGBHID <td>1582</td> <td>MAHARASHTRA</td> <td>BID</td> <td>KAIJ</td>	1582	MAHARASHTRA	BID	KAIJ
1585MAHARASHTRABIDPATODA1586MAHARASHTRABIDSIRUR1587MAHARASHTRABULDANABULDANA1588MAHARASHTRABULDANACHIKHALI1589MAHARASHTRABULDANADEULGAON RAJA1590MAHARASHTRABULDANAJALGAON JAMOD1591MAHARASHTRABULDANAKHAMGAON1592MAHARASHTRABULDANALONAR1593MAHARASHTRABULDANAMALAKAPUR1594MAHARASHTRABULDANAMALAKAPUR1595MAHARASHTRABULDANAMALAKAR1596MAHARASHTRABULDANAMOTALA1597MAHARASHTRABULDANANANDURA1598MAHARASHTRABULDANASANGRAMPUR1599MAHARASHTRABULDANASINDKHED RAJA1600MAHARASHTRABULDANASINDKHED RAJA1601MAHARASHTRACHANDRAPURBALLARPUR1601MAHARASHTRACHANDRAPURBAMHAPURI1602MAHARASHTRACHANDRAPURCHANDRAPUR1604MAHARASHTRACHANDRAPURCHIMUR1605MAHARASHTRACHANDRAPURGONDPIPARI1606MAHARASHTRACHANDRAPURJIWATI1607MAHARASHTRACHANDRAPURMUL1608MAHARASHTRACHANDRAPURMALARASHTRA1609MAHARASHTRACHANDRAPURMAGBHID1610MAHARASHTRACHANDRAPURMAGBHID1611MAHARASHTRACHANDRAPUR<	1583	MAHARASHTRA	BID	MAJALGAON
1586MAHARASHTRABIDSIRUR1587MAHARASHTRABULDANABULDANA1588MAHARASHTRABULDANACHIKHALI1589MAHARASHTRABULDANACHIKHALI1590MAHARASHTRABULDANAJALGAON RAJA1590MAHARASHTRABULDANAJALGAON JAMOD1591MAHARASHTRABULDANAKHAMGAON1592MAHARASHTRABULDANALONAR1593MAHARASHTRABULDANAMALAKAPUR1594MAHARASHTRABULDANAMEHAKAR1595MAHARASHTRABULDANAMOTALA1596MAHARASHTRABULDANANANDURA1597MAHARASHTRABULDANASANGRAMPUR1598MAHARASHTRABULDANASANGRAMPUR1599MAHARASHTRABULDANASINDKHED RAJA1600MAHARASHTRABULDANASINDKHED RAJA1601MAHARASHTRACHANDRAPURBALLARPUR1602MAHARASHTRACHANDRAPURBHADRAWATI1603MAHARASHTRACHANDRAPURCHANDRAPUR1604MAHARASHTRACHANDRAPURCHANDRAPUR1605MAHARASHTRACHANDRAPURGONDPIPARI1606MAHARASHTRACHANDRAPURJIWATI1607MAHARASHTRACHANDRAPURMUL1608MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURNAGBHID1610MAHARASHTRACHANDRAPUR <t< td=""><td>1584</td><td>MAHARASHTRA</td><td>BID</td><td>PARLI</td></t<>	1584	MAHARASHTRA	BID	PARLI
1587MAHARASHTRABULDANABULDANA1588MAHARASHTRABULDANACHIKHALI1589MAHARASHTRABULDANADEULGAON RAJA1590MAHARASHTRABULDANAJALGAON JAMOD1591MAHARASHTRABULDANAKHAMGAON1592MAHARASHTRABULDANALONAR1593MAHARASHTRABULDANAMALAKAPUR1594MAHARASHTRABULDANAMALAKAPUR1595MAHARASHTRABULDANAMALAKAPUR1596MAHARASHTRABULDANAMOTALA1597MAHARASHTRABULDANANANDURA1598MAHARASHTRABULDANASANGRAMPUR1599MAHARASHTRABULDANASINDKHED RAJA1600MAHARASHTRABULDANASINDKHED RAJA1601MAHARASHTRACHANDRAPURBHADRAWATI1602MAHARASHTRACHANDRAPURBRAMHAPURI1603MAHARASHTRACHANDRAPURCHANDRAPUR1604MAHARASHTRACHANDRAPURGONDPIPARI1605MAHARASHTRACHANDRAPURGONDPIPARI1606MAHARASHTRACHANDRAPURJIWATI1607MAHARASHTRACHANDRAPURMUL1608MAHARASHTRACHANDRAPURMOL1609MAHARASHTRACHANDRAPURMOL1610MAHARASHTRACHANDRAPURNAGBHID1611MAHARASHTRACHANDRAPURRAJURA1612MAHARASHTRACHANDRAPURRAJURA1612MAHARASHTRACHAND	1585	MAHARASHTRA	BID	PATODA
1588MAHARASHTRABULDANACHIKHALI1589MAHARASHTRABULDANADEULGAON RAJA1590MAHARASHTRABULDANAJALGAON JAMOD1591MAHARASHTRABULDANAKHAMGAON1592MAHARASHTRABULDANALONAR1593MAHARASHTRABULDANAMALAKAPUR1594MAHARASHTRABULDANAMALAKAPUR1595MAHARASHTRABULDANAMEHAKAR1596MAHARASHTRABULDANAMOTALA1596MAHARASHTRABULDANANANDURA1597MAHARASHTRABULDANASANGRAMPUR1598MAHARASHTRABULDANASINDKHED RAJA1600MAHARASHTRABULDANASINDKHED RAJA1601MAHARASHTRABULDANASINDKHED RAJA1602MAHARASHTRACHANDRAPURBHADRAWATI1603MAHARASHTRACHANDRAPURBHADRAWATI1604MAHARASHTRACHANDRAPURGONDPIPARI1605MAHARASHTRACHANDRAPURGONDPIPARI1606MAHARASHTRACHANDRAPURJWATI1607MAHARASHTRACHANDRAPURJWATI1608MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURNAGBHID1610MAHARASHTRACHANDRAPURNAGBHID1611MAHARASHTRACHANDRAPURRAJURA1612MAHARASHTRACHANDRAPURRAJURA	1586	MAHARASHTRA	BID	SIRUR
1589MAHARASHTRABULDANADEULGAON RAJA1590MAHARASHTRABULDANAJALGAON JAMOD1591MAHARASHTRABULDANAKHAMGAON1592MAHARASHTRABULDANALONAR1593MAHARASHTRABULDANAMALAKAPUR1594MAHARASHTRABULDANAMEHAKAR1595MAHARASHTRABULDANAMEHAKAR1596MAHARASHTRABULDANAMOTALA1596MAHARASHTRABULDANANANDURA1597MAHARASHTRABULDANASANGRAMPUR1598MAHARASHTRABULDANASHEGAON1599MAHARASHTRABULDANASINDKHED RAJA1600MAHARASHTRABULDANASINDKHED RAJA1600MAHARASHTRACHANDRAPURBALLARPUR1601MAHARASHTRACHANDRAPURBALARPUR1602MAHARASHTRACHANDRAPURBRAMHAPURI1603MAHARASHTRACHANDRAPURCHANDRAPUR1604MAHARASHTRACHANDRAPURCHANDRAPUR1605MAHARASHTRACHANDRAPURGONDPIPARI1606MAHARASHTRACHANDRAPURJIWATI1607MAHARASHTRACHANDRAPURMUL1608MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURMAGBHID1610MAHARASHTRACHANDRAPURMUL1604MAHARASHTRACHANDRAPURMUL1605MAHARASHTRACHANDRAPUR1606MAHARASHTRACHANDRAPUR<	1587	MAHARASHTRA	BULDANA	BULDANA
1590MAHARASHTRABULDANAJALGAON JAMOD1591MAHARASHTRABULDANAKHAMGAON1592MAHARASHTRABULDANALONAR1593MAHARASHTRABULDANAMALAKAPUR1594MAHARASHTRABULDANAMEHAKAR1595MAHARASHTRABULDANAMOTALA1596MAHARASHTRABULDANAMOTALA1597MAHARASHTRABULDANANANDURA1598MAHARASHTRABULDANASANGRAMPUR1599MAHARASHTRABULDANASINDKHED RAJA1600MAHARASHTRABULDANASINDKHED RAJA1601MAHARASHTRACHANDRAPURBALLARPUR1602MAHARASHTRACHANDRAPURBRAMHAPURI1603MAHARASHTRACHANDRAPURCHANDRAPUR1604MAHARASHTRACHANDRAPURGONDPIPARI1605MAHARASHTRACHANDRAPURGONDPIPARI1606MAHARASHTRACHANDRAPURJIWATI1607MAHARASHTRACHANDRAPURJIWATI1608MAHARASHTRACHANDRAPURJIWATI1609MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURNAGBHID1610MAHARASHTRACHANDRAPURNAGBHID1610MAHARASHTRACHANDRAPURNAGBHID1610MAHARASHTRACHANDRAPURRAJURA1611MAHARASHTRACHANDRAPURRAJURA1612MAHARASHTRACHANDRA	1588	MAHARASHTRA	BULDANA	CHIKHALI
1591MAHARASHTRABULDANAKHAMGAON1592MAHARASHTRABULDANALONAR1593MAHARASHTRABULDANAMALAKAPUR1594MAHARASHTRABULDANAMEHAKAR1595MAHARASHTRABULDANAMOTALA1596MAHARASHTRABULDANANANDURA1597MAHARASHTRABULDANASANGRAMPUR1598MAHARASHTRABULDANASANGRAMPUR1599MAHARASHTRABULDANASINDKHED RAJA1600MAHARASHTRABULDANASINDKHED RAJA1601MAHARASHTRACHANDRAPURBALLARPUR1602MAHARASHTRACHANDRAPURBHADRAWATI1603MAHARASHTRACHANDRAPURBRAMHAPURI1604MAHARASHTRACHANDRAPURCHIMUR1605MAHARASHTRACHANDRAPURGONDPIPARI1606MAHARASHTRACHANDRAPURJIWATI1607MAHARASHTRACHANDRAPURMUL1608MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURNAGBHID1610MAHARASHTRACHANDRAPURNAGBHID1611MAHARASHTRACHANDRAPURRAJURA1612MAHARASHTRACHANDRAPURSAOLI	1589	MAHARASHTRA	BULDANA	DEULGAON RAJA
1592MAHARASHTRABULDANALONAR1593MAHARASHTRABULDANAMALAKAPUR1594MAHARASHTRABULDANAMEHAKAR1595MAHARASHTRABULDANAMOTALA1596MAHARASHTRABULDANANANDURA1597MAHARASHTRABULDANASANGRAMPUR1598MAHARASHTRABULDANASHEGAON1599MAHARASHTRABULDANASINDKHED RAJA1600MAHARASHTRABULDANASINDKHED RAJA1601MAHARASHTRACHANDRAPURBALLARPUR1601MAHARASHTRACHANDRAPURBHADRAWATI1602MAHARASHTRACHANDRAPURBRAMHAPURI1603MAHARASHTRACHANDRAPURCHANDRAPUR1604MAHARASHTRACHANDRAPURCHIMUR1605MAHARASHTRACHANDRAPURGONDPIPARI1606MAHARASHTRACHANDRAPURJIWATI1607MAHARASHTRACHANDRAPURJIWATI1608MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURNAGBHID1610MAHARASHTRACHANDRAPURNAGBHID1611MAHARASHTRACHANDRAPURRAJURA1612MAHARASHTRACHANDRAPURSAOLI	1590	MAHARASHTRA	BULDANA	JALGAON JAMOD
1593MAHARASHTRABULDANAMALAKAPUR1594MAHARASHTRABULDANAMEHAKAR1595MAHARASHTRABULDANAMOTALA1596MAHARASHTRABULDANANANDURA1597MAHARASHTRABULDANASANGRAMPUR1598MAHARASHTRABULDANASHEGAON1599MAHARASHTRABULDANASINDKHED RAJA1600MAHARASHTRABULDANASINDKHED RAJA1601MAHARASHTRACHANDRAPURBALLARPUR1602MAHARASHTRACHANDRAPURBHADRAWATI1603MAHARASHTRACHANDRAPURBRAMHAPURI1604MAHARASHTRACHANDRAPURCHANDRAPUR1605MAHARASHTRACHANDRAPURGONDPIPARI1606MAHARASHTRACHANDRAPURJIWATI1607MAHARASHTRACHANDRAPURJIWATI1608MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURNAGBHID1610MAHARASHTRACHANDRAPURNAGBHID1610MAHARASHTRACHANDRAPURRAJURA1611MAHARASHTRACHANDRAPURRAJURA1612MAHARASHTRACHANDRAPURSAOLI	1591	MAHARASHTRA	BULDANA	KHAMGAON
1594MAHARASHTRABULDANAMEHAKAR1595MAHARASHTRABULDANAMOTALA1596MAHARASHTRABULDANANANDURA1597MAHARASHTRABULDANASANGRAMPUR1598MAHARASHTRABULDANASHEGAON1599MAHARASHTRABULDANASINDKHED RAJA1600MAHARASHTRABULDANASINDKHED RAJA1601MAHARASHTRACHANDRAPURBALLARPUR1602MAHARASHTRACHANDRAPURBHADRAWATI1603MAHARASHTRACHANDRAPURBRAMHAPURI1604MAHARASHTRACHANDRAPURCHANDRAPUR1605MAHARASHTRACHANDRAPURGONDPIPARI1606MAHARASHTRACHANDRAPURJIWATI1607MAHARASHTRACHANDRAPURJIWATI1608MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURMIL1610MAHARASHTRACHANDRAPURNAGBHID1610MAHARASHTRACHANDRAPURNAGBHID1610MAHARASHTRACHANDRAPURRAJURA1611MAHARASHTRACHANDRAPURRAJURA1612MAHARASHTRACHANDRAPURSAOLI	1592	MAHARASHTRA	BULDANA	LONAR
1595MAHARASHTRABULDANAMOTALA1596MAHARASHTRABULDANANANDURA1597MAHARASHTRABULDANASANGRAMPUR1598MAHARASHTRABULDANASHEGAON1599MAHARASHTRABULDANASINDKHED RAJA1600MAHARASHTRABULDANASINDKHED RAJA1601MAHARASHTRACHANDRAPURBALLARPUR1601MAHARASHTRACHANDRAPURBHADRAWATI1602MAHARASHTRACHANDRAPURBRAMHAPURI1603MAHARASHTRACHANDRAPURCHANDRAPUR1604MAHARASHTRACHANDRAPURCHIMUR1605MAHARASHTRACHANDRAPURGONDPIPARI1606MAHARASHTRACHANDRAPURJIWATI1607MAHARASHTRACHANDRAPURJIWATI1608MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURNAGBHID1610MAHARASHTRACHANDRAPURNAGBHID1611MAHARASHTRACHANDRAPURRAJURA1612MAHARASHTRACHANDRAPURSAOLI	1593	MAHARASHTRA	BULDANA	MALAKAPUR
1596MAHARASHTRABULDANANANDURA1597MAHARASHTRABULDANASANGRAMPUR1598MAHARASHTRABULDANASHEGAON1599MAHARASHTRABULDANASINDKHED RAJA1600MAHARASHTRACHANDRAPURBALLARPUR1601MAHARASHTRACHANDRAPURBHADRAWATI1602MAHARASHTRACHANDRAPURBRAMHAPURI1603MAHARASHTRACHANDRAPURBRAMHAPURI1604MAHARASHTRACHANDRAPURCHANDRAPUR1605MAHARASHTRACHANDRAPURCHIMUR1606MAHARASHTRACHANDRAPURGONDPIPARI1607MAHARASHTRACHANDRAPURJIWATI1608MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURNAGBHID1610MAHARASHTRACHANDRAPURNAGBHID1611MAHARASHTRACHANDRAPURRAJURA1612MAHARASHTRACHANDRAPURSAOLI	1594	MAHARASHTRA	BULDANA	MEHAKAR
1597MAHARASHTRABULDANASANGRAMPUR1598MAHARASHTRABULDANASHEGAON1599MAHARASHTRABULDANASINDKHED RAJA1600MAHARASHTRACHANDRAPURBALLARPUR1601MAHARASHTRACHANDRAPURBHADRAWATI1602MAHARASHTRACHANDRAPURBRAMHAPURI1603MAHARASHTRACHANDRAPURCHANDRAPUR1604MAHARASHTRACHANDRAPURCHIMUR1605MAHARASHTRACHANDRAPURGONDPIPARI1606MAHARASHTRACHANDRAPURJIWATI1607MAHARASHTRACHANDRAPURMUL1608MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURMUL1610MAHARASHTRACHANDRAPURNAGBHID1611MAHARASHTRACHANDRAPURRAJURA1612MAHARASHTRACHANDRAPURSAOLI	1595	MAHARASHTRA	BULDANA	MOTALA
1598MAHARASHTRABULDANASHEGAON1599MAHARASHTRABULDANASINDKHED RAJA1600MAHARASHTRACHANDRAPURBALLARPUR1601MAHARASHTRACHANDRAPURBHADRAWATI1602MAHARASHTRACHANDRAPURBRAMHAPURI1603MAHARASHTRACHANDRAPURCHANDRAPUR1604MAHARASHTRACHANDRAPURCHIMUR1605MAHARASHTRACHANDRAPURGONDPIPARI1606MAHARASHTRACHANDRAPURJIWATI1607MAHARASHTRACHANDRAPURJIWATI1608MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURMUL1610MAHARASHTRACHANDRAPURNAGBHID1611MAHARASHTRACHANDRAPURRAJURA1612MAHARASHTRACHANDRAPURSAOLI	1596	MAHARASHTRA	BULDANA	NANDURA
1599MAHARASHTRABULDANASINDKHED RAJA1600MAHARASHTRACHANDRAPURBALLARPUR1601MAHARASHTRACHANDRAPURBHADRAWATI1602MAHARASHTRACHANDRAPURBRAMHAPURI1603MAHARASHTRACHANDRAPURCHANDRAPUR1604MAHARASHTRACHANDRAPURCHIMUR1605MAHARASHTRACHANDRAPURGONDPIPARI1606MAHARASHTRACHANDRAPURJIWATI1607MAHARASHTRACHANDRAPURKORPANA1608MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURNAGBHID1610MAHARASHTRACHANDRAPURNAGBHID1611MAHARASHTRACHANDRAPURRAJURA1612MAHARASHTRACHANDRAPURSAOLI	1597	MAHARASHTRA	BULDANA	SANGRAMPUR
1600MAHARASHTRACHANDRAPURBALLARPUR1601MAHARASHTRACHANDRAPURBHADRAWATI1602MAHARASHTRACHANDRAPURBRAMHAPURI1603MAHARASHTRACHANDRAPURCHANDRAPUR1604MAHARASHTRACHANDRAPURCHIMUR1605MAHARASHTRACHANDRAPURGONDPIPARI1606MAHARASHTRACHANDRAPURJIWATI1607MAHARASHTRACHANDRAPURKORPANA1608MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURMUL1610MAHARASHTRACHANDRAPURNAGBHID1611MAHARASHTRACHANDRAPURRAJURA1612MAHARASHTRACHANDRAPURSAOLI	1598	MAHARASHTRA	BULDANA	SHEGAON
1601MAHARASHTRACHANDRAPURBHADRAWATI1602MAHARASHTRACHANDRAPURBRAMHAPURI1603MAHARASHTRACHANDRAPURCHANDRAPUR1604MAHARASHTRACHANDRAPURCHIMUR1605MAHARASHTRACHANDRAPURGONDPIPARI1606MAHARASHTRACHANDRAPURJIWATI1607MAHARASHTRACHANDRAPURKORPANA1608MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURMUL1610MAHARASHTRACHANDRAPURNAGBHID1611MAHARASHTRACHANDRAPURRAJURA1612MAHARASHTRACHANDRAPURSAOLI	1599	MAHARASHTRA	BULDANA	SINDKHED RAJA
1602MAHARASHTRACHANDRAPURBRAMHAPURI1603MAHARASHTRACHANDRAPURCHANDRAPUR1604MAHARASHTRACHANDRAPURCHIMUR1605MAHARASHTRACHANDRAPURGONDPIPARI1606MAHARASHTRACHANDRAPURJIWATI1607MAHARASHTRACHANDRAPURKORPANA1608MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURNAGBHID1610MAHARASHTRACHANDRAPURNAGBHID1611MAHARASHTRACHANDRAPURRAJURA1612MAHARASHTRACHANDRAPURSAOLI	1600	MAHARASHTRA	CHANDRAPUR	BALLARPUR
1603MAHARASHTRACHANDRAPURCHANDRAPUR1604MAHARASHTRACHANDRAPURCHIMUR1605MAHARASHTRACHANDRAPURGONDPIPARI1606MAHARASHTRACHANDRAPURJIWATI1607MAHARASHTRACHANDRAPURKORPANA1608MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURNAGBHID1610MAHARASHTRACHANDRAPURNAGBHID1611MAHARASHTRACHANDRAPURPOMBHURNA1612MAHARASHTRACHANDRAPURSAOLI	1601	MAHARASHTRA	CHANDRAPUR	BHADRAWATI
1604MAHARASHTRACHANDRAPURCHIMUR1605MAHARASHTRACHANDRAPURGONDPIPARI1606MAHARASHTRACHANDRAPURJIWATI1607MAHARASHTRACHANDRAPURKORPANA1608MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURNAGBHID1610MAHARASHTRACHANDRAPURPOMBHURNA1611MAHARASHTRACHANDRAPURRAJURA1612MAHARASHTRACHANDRAPURSAOLI	1602	MAHARASHTRA	CHANDRAPUR	BRAMHAPURI
1605MAHARASHTRACHANDRAPURGONDPIPARI1606MAHARASHTRACHANDRAPURJIWATI1607MAHARASHTRACHANDRAPURKORPANA1608MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURNAGBHID1610MAHARASHTRACHANDRAPURPOMBHURNA1611MAHARASHTRACHANDRAPURRAJURA1612MAHARASHTRACHANDRAPURSAOLI	1603	MAHARASHTRA	CHANDRAPUR	CHANDRAPUR
1606MAHARASHTRACHANDRAPURJIWATI1607MAHARASHTRACHANDRAPURKORPANA1608MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURNAGBHID1610MAHARASHTRACHANDRAPURPOMBHURNA1611MAHARASHTRACHANDRAPURRAJURA1612MAHARASHTRACHANDRAPURSAOLI	1604	MAHARASHTRA	CHANDRAPUR	CHIMUR
1607MAHARASHTRACHANDRAPURKORPANA1608MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURNAGBHID1610MAHARASHTRACHANDRAPURPOMBHURNA1611MAHARASHTRACHANDRAPURRAJURA1612MAHARASHTRACHANDRAPURSAOLI	1605	MAHARASHTRA	CHANDRAPUR	GONDPIPARI
1608MAHARASHTRACHANDRAPURMUL1609MAHARASHTRACHANDRAPURNAGBHID1610MAHARASHTRACHANDRAPURPOMBHURNA1611MAHARASHTRACHANDRAPURRAJURA1612MAHARASHTRACHANDRAPURSAOLI	1606	MAHARASHTRA	CHANDRAPUR	JIWATI
1609MAHARASHTRACHANDRAPURNAGBHID1610MAHARASHTRACHANDRAPURPOMBHURNA1611MAHARASHTRACHANDRAPURRAJURA1612MAHARASHTRACHANDRAPURSAOLI	1607	MAHARASHTRA	CHANDRAPUR	KORPANA
1610MAHARASHTRACHANDRAPURPOMBHURNA1611MAHARASHTRACHANDRAPURRAJURA1612MAHARASHTRACHANDRAPURSAOLI	1608	MAHARASHTRA	CHANDRAPUR	MUL
1611MAHARASHTRACHANDRAPURRAJURA1612MAHARASHTRACHANDRAPURSAOLI	1609	MAHARASHTRA	CHANDRAPUR	NAGBHID
1612 MAHARASHTRA CHANDRAPUR SAOLI	1610	MAHARASHTRA	CHANDRAPUR	POMBHURNA
	1611	MAHARASHTRA	CHANDRAPUR	RAJURA
	1612	MAHARASHTRA	CHANDRAPUR	SAOLI
	1613	MAHARASHTRA	CHANDRAPUR	SINDEWAHI
1614 MAHARASHTRA CHANDRAPUR WARORA				

1615	MAHARASHTRA	DHULE	DHULE
1616	MAHARASHTRA	DHULE	SAKRI
1617	MAHARASHTRA	DHULE	SHINDKHEDA
1618	MAHARASHTRA	DHULE	SHIRPUR
1619	MAHARASHTRA	GADCHIROLI	ARMORI
1620	MAHARASHTRA	GADCHIROLI	CHAMORSHI
1621	MAHARASHTRA	GADCHIROLI	DESAIGANJ
1621	MAHARASHTRA	GADCHIROLI	GADCHIROLI
1622	MAHARASHTRA	GADCHIROLI	KORCHI
1623	MAHARASHTRA	GADCHIROLI	KURKHEDA
1624	MAHARASHTRA	GADCHIROLI	MULCHERA
1625	MAHARASHTRA	GONDIYA	AMGAON
1627	MAHARASHTRA	GONDIYA	ARJUNI MOR
1627	MAHARASHTRA	GONDIYA	DEORI
1628	MAHARASHTRA	GONDIYA	GONDIA
1629	MAHARASHTRA	GONDIYA	GOREGAON
			SADAK ARJUNI
1631	MAHARASHTRA	GONDIYA	
1632	MAHARASHTRA	GONDIYA	SALEKASA
1633	MAHARASHTRA	GONDIYA	TIRODA
1634	MAHARASHTRA	HINGOLI	AUNDHA
1635	MAHARASHTRA	HINGOLI	BASMATH
1636	MAHARASHTRA	HINGOLI	KALAMNURI
1637	MAHARASHTRA	HINGOLI	SENGAON
1638	MAHARASHTRA	JALGAON	AMALNER
1639	MAHARASHTRA	JALGAON	BHADGAON
1640	MAHARASHTRA	JALGAON	BHUSAWAL
1641	MAHARASHTRA	JALGAON	BODWAD
1642	MAHARASHTRA	JALGAON	CHALISGAON
1643	MAHARASHTRA	JALGAON	CHOPADA
1644	MAHARASHTRA	JALGAON	DHARANGAON
1645	MAHARASHTRA	JALGAON	ERANDOL
1646	MAHARASHTRA	JALGAON	JALGAON
1647	MAHARASHTRA	JALGAON	JAMNER
1648	MAHARASHTRA	JALGAON	MUKTAINAGAR
1649	MAHARASHTRA	JALGAON	PACHORA
1650	MAHARASHTRA	JALGAON	PAROLA
1651	MAHARASHTRA	JALGAON	RAWER
1652	MAHARASHTRA	JALGAON	YAWAL
1653	MAHARASHTRA	JALNA	JAFRABAD
1654	MAHARASHTRA	KOLHAPUR	AAJARA
1655	MAHARASHTRA	KOLHAPUR	BHUDARGAD
1656	MAHARASHTRA	KOLHAPUR	CHANDGAD
1657	MAHARASHTRA	KOLHAPUR	GADHINGLAJ
1658	MAHARASHTRA	KOLHAPUR	HATKALANGLE
1659	MAHARASHTRA	KOLHAPUR	KAGAL
1660	MAHARASHTRA	KOLHAPUR	KARVEER
1661	MAHARASHTRA	KOLHAPUR	KOLHAPUR
1662	MAHARASHTRA	KOLHAPUR	PANHALA
1663	MAHARASHTRA	KOLHAPUR	RADHANAGARI
1664	MAHARASHTRA	KOLHAPUR	SHAHUWADI
1665	MAHARASHTRA	KOLHAPUR	SHIROL
1666	MAHARASHTRA	LATUR	AHAMADPUR
1667	MAHARASHTRA	LATUR	AUSA
	ducation rfa/06.07.2012		90

1668	MAHARASHTRA	LATUR	CHAKUR
1669	MAHARASHTRA	LATUR	DEVANI
1670	MAHARASHTRA	LATUR	JALKOT
1671	MAHARASHTRA	LATUR	LATUR
1672	MAHARASHTRA	LATUR	NILANGA
1672	MAHARASHTRA	LATUR	RENAPUR
1674	MAHARASHTRA	LATUR	SHIRUR ANANTPAL
1675	MAHARASHTRA	LATUR	UDGIR
1676	MAHARASHTRA	NAGPUR	BHIWAPUR
1677	MAHARASHTRA	NAGPUR	HINGANA
1678	MAHARASHTRA	NAGPUR	KALMESHWAR
1679	MAHARASHTRA	NAGPUR	KAMPTEE
1680	MAHARASHTRA	NAGPUR	KATOL
1681	MAHARASHTRA	NAGPUR	KUHI
1682	MAHARASHTRA	NAGPUR	MOUDA
1682	MAHARASHTRA	NAGPUR	NAGPUR (GRAMIN)
			× /
1684	MAHARASHTRA	NAGPUR	NARKHED DADSHIWANI
1685	MAHARASHTRA	NAGPUR	PARSHIWANI BAMTEK
1686	MAHARASHTRA	NAGPUR	RAMTEK
1687	MAHARASHTRA	NAGPUR	SAONER
1688	MAHARASHTRA	NAGPUR	UMRED
1689	MAHARASHTRA	NANDED	ARDHAPUR
1690	MAHARASHTRA	NANDED	BHOKAR
1691	MAHARASHTRA	NANDED	DEGLOOR
1692	MAHARASHTRA	NANDED	HADGAON
1693	MAHARASHTRA	NANDED	HIMAYAT NAGAR
1694	MAHARASHTRA	NANDED	KANDHAR
1695	MAHARASHTRA	NANDED	KINWAT
1696	MAHARASHTRA	NANDED	LOHA
1697	MAHARASHTRA	NANDED	MAHOOR
1698	MAHARASHTRA	NANDED	MUKHED
1699	MAHARASHTRA	NANDED	NAIGAON
1700	MAHARASHTRA	NANDED	NANDED
1701	MAHARASHTRA	NASHIK	BAGLAN
1702	MAHARASHTRA	NASHIK	CHANDWAD
1703	MAHARASHTRA	NASHIK	DEVLA
1704	MAHARASHTRA	NASHIK	DINDORI
1705	MAHARASHTRA	NASHIK	KALWAN
1706	MAHARASHTRA	NASHIK	MALEGAON
1707	MAHARASHTRA	NASHIK	NANDGAON
1708	MAHARASHTRA	NASHIK	NASHIK
1709	MAHARASHTRA	NASHIK	NIPHAD
1710	MAHARASHTRA	NASHIK	SINNER
1711	MAHARASHTRA	NASHIK	YEOLA
1712	MAHARASHTRA	OSMANABAD	BHOOM
1713	MAHARASHTRA	OSMANABAD	KALLAMB
1714	MAHARASHTRA	OSMANABAD	LOHARA
1715	MAHARASHTRA	OSMANABAD	OMERGA
1716	MAHARASHTRA	OSMANABAD	OSMANABAD
1717	MAHARASHTRA	OSMANABAD	PARANDA
1718	MAHARASHTRA	OSMANABAD	TULJAPUR
1719	MAHARASHTRA	OSMANABAD	WASHI
1720	MAHARASHTRA	PARBHANI	PALAM
	ducation rfa/06.07.2012	•	91

1721	MAHARASHTRA	PARBHANI	SONPETH
1722	MAHARASHTRA	PUNE	AMBEGAON
1723	MAHARASHTRA	PUNE	BARAMATI
1724	MAHARASHTRA	PUNE	BHOR
1725	MAHARASHTRA	PUNE	DAUND
1726	MAHARASHTRA	PUNE	HAVELI
1727	MAHARASHTRA	PUNE	INDAPUR
1728	MAHARASHTRA	PUNE	JUNNAR
1729	MAHARASHTRA	PUNE	KHED
1730	MAHARASHTRA	PUNE	MAVAL
1731	MAHARASHTRA	PUNE	MULSHI
1732	MAHARASHTRA	PUNE	PURANDAR
1732	MAHARASHTRA	PUNE	SHIRUR
1734	MAHARASHTRA	PUNE	VELHE
		RAIGARH	
1735	MAHARASHTRA	(MAHARASHTRA)	ALIBAG
1736		RAIGARH	<b>VADIAT</b>
1/30	MAHARASHTRA	(MAHARASHTRA)	KARJAT
1737	MAHARASHTRA	RAIGARH	KHALAPUR
1151		(MAHARASHTRA)	
1738	MAHARASHTRA	RAIGARH (MAHARASHTRA)	MAHAD
		RAIGARH	
1739	MAHARASHTRA	(MAHARASHTRA)	MAHASALA
1740		RAIGARH	NANGAON
1740	MAHARASHTRA	(MAHARASHTRA)	MANGAON
1741	MAHARASHTRA	RAIGARH	MURUD
1/71		(MAHARASHTRA)	Morrob
1742	MAHARASHTRA	RAIGARH	PANVEL
		(MAHARASHTRA) RAIGARH	
1743	MAHARASHTRA	(MAHARASHTRA)	PEN
1544		RAIGARH	
1744	MAHARASHTRA	(MAHARASHTRA)	POLADPUR
1745	MAHARASHTRA	RAIGARH	ROHA
1715		(MAHARASHTRA)	
1746	MAHARASHTRA	RAIGARH	SHRIVARDHAN
		(MAHARASHTRA) RAIGARH	
1747	MAHARASHTRA	(MAHARASHTRA)	SUDHAGAD
1540		RAIGARH	
1748	MAHARASHTRA	(MAHARASHTRA)	TALA
1749	MAHARASHTRA	RAIGARH	URAN
-		(MAHARASHTRA)	
1750	MAHARASHTRA	RATNAGIRI	CHIPLUN
1751	MAHARASHTRA	RATNAGIRI	DAPOLI
1752	MAHARASHTRA	RATNAGIRI	GUHAGAR
1753	MAHARASHTRA	RATNAGIRI	KHED
1754	MAHARASHTRA	RATNAGIRI	LANJA
1755	MAHARASHTRA	RATNAGIRI	MANDANGAD
1756	MAHARASHTRA	RATNAGIRI	RAJAPUR
1757	MAHARASHTRA	RATNAGIRI	RATNAGIRI
1758	MAHARASHTRA	RATNAGIRI	SANGAMESHWAR
1759	MAHARASHTRA	SANGLI	ATAPADI
1760	MAHARASHTRA	SANGLI	JATH
1761	MAHARASHTRA	SANGLI	KADEGAON

1762	MAHARASHTRA	SANGLI	KAVATHE MAHANKAL
1762	MAHARASHTRA	SANGLI	KHANAPUR
1764	MAHARASHTRA	SANGLI	MIRAJ
1765	MAHARASHTRA	SANGLI	PALUS
1766	MAHARASHTRA	SANGLI	SHIRALA
1767	MAHARASHTRA	SANGLI	TASGAON
1768	MAHARASHTRA	SANGLI	WALWA
1769	MAHARASHTRA	SATARA	JAOLI
1770	MAHARASHTRA	SATARA	KARAD
1771	MAHARASHTRA	SATARA	KHANDALA
1772	MAHARASHTRA	SATARA	KHATAV
1773	MAHARASHTRA	SATARA	KOREGAON
1774	MAHARASHTRA	SATARA	MAHABALESHWAR
1775	MAHARASHTRA	SATARA	MAN
1776	MAHARASHTRA	SATARA	PATAN
1777	MAHARASHTRA	SATARA	PHALTAN
1778	MAHARASHTRA	SATARA	SATARA
1779	MAHARASHTRA	SATARA	WAI
1779	MAHARASHTRA	SINDHUDURG	DEOGAD
1781	MAHARASHTRA	SINDHUDURG	DODAMARG
1782	MAHARASHTRA	SINDHUDURG	KANKAVLI
1783	MAHARASHTRA	SINDHUDURG	KUDAL
1784	MAHARASHTRA	SINDHUDURG	MALVAN
1785	MAHARASHTRA	SINDHUDURG	SAWANTWADI
1786	MAHARASHTRA	SINDHUDURG	VAIBHAVWADI
1787	MAHARASHTRA	SINDHUDURG	VENGURLE
1787	MAHARASHTRA	SOLAPUR	AKKALKOT
1789	MAHARASHTRA	SOLAPUR	BARSHI
1790	MAHARASHTRA	SOLAPUR	KARMALA
1790	MAHARASHTRA	SOLAPUR	MADHA
1791	MAHARASHTRA	SOLAPUR	MALSHIRUS
1792	MAHARASHTRA	SOLAPUR	MALSHIKUS MANGALWEDHA
1793	MAHARASHTRA	SOLAPUR	MANGALWEDHA
1794	MAHARASHTRA	SOLAPUR	PANDHARPUR
1795	MAHARASHTRA	SOLAPUR	SANGOLA
1790	MAHARASHTRA	SOLAPUR	SOLAPUR NORTH
1797	MAHARASHTRA		SOLAPUR NORTH
1798	MAHARASHTRA	SOLAPUR THANE	AMBERNATH
1800	MAHARASHTRA	THANE	BHIVANDI
1800	MAHARASHTRA	THANE	KALYAN
1801	MAHARASHTRA	THANE	MURBAD
1802	MAHARASHTRA	THANE	NAVI MUMBAI
1803	MAHARASHTRA	THANE	PALGHAR
1804	MAHARASHTRA	THANE	SHAHAPUR
1805	MAHARASHTRA	THANE	THANE
1800	MAHARASHTRA	THANE	VASAI
1807	MAHARASHTRA	THANE	WADA
1808	MAHARASHTRA	WARDHA	ARVI
1809	MAHARASHTRA		
		WARDHA	ASHTI
1811	MAHARASHTRA	WARDHA	DEOLI
1812	MAHARASHTRA	WARDHA	HINGANGHAT
1813	MAHARASHTRA	WARDHA	KARANJA SAMUDDADUD
1814	MAHARASHTRA	WARDHA	SAMUDRAPUR 93

1815	MAHARASHTRA	WARDHA	SELOO
1816	MAHARASHTRA	WARDHA	WARDHA
1817	MAHARASHTRA	WASHIM	KARANJA LAD
1818	MAHARASHTRA	WASHIM	MALEGAON
1819	MAHARASHTRA	WASHIM	MANGARULPIR
1820	MAHARASHTRA	WASHIM	MANORA
1820	MAHARASHTRA	WASHIM	RISOD
1822	MAHARASHTRA	WASHIM	WASHIM
1823	MAHARASHTRA	YAVATMAL	ARNI
1824	MAHARASHTRA	YAVATMAL	BABHULGAON
1825	MAHARASHTRA	YAVATMAL	DARWHA
1826	MAHARASHTRA	YAVATMAL	DIGRAS
1820	MAHARASHTRA	YAVATMAL	GHATANJI
1827	MAHARASHTRA	YAVATMAL	KALAMB
1829	MAHARASHTRA	YAVATMAL	MAHAGAON
1829	MAHARASHTRA	YAVATMAL	MAREGAON
1830	MAHARASHTRA	YAVATMAL	NER
1831	MAHARASHTRA	YAVATMAL	PANDHARKAWADA
1832	MAHARASHTRA	YAVATMAL	PUSAD
1833	MAHARASHTRA	YAVATMAL	RALEGAON
1834	MAHARASHTRA	YAVATMAL	UMARKHED
1835	MAHARASHTRA	YAVATMAL	WANI
1830	MAHARASHTRA	YAVATMAL	YAVATMAL
1837	MAHARASHTRA		ZARI
1839	MANIPUR	YAVATMAL BISHNUPUR	BISHNUPUR
1839			
	MANIPUR	BISHNUPUR	MOIRANG
1841	MANIPUR	CHANDEL	CHANDEL
1842 1843	MANIPUR MANIPUR	CHURACHANDPUR CHURACHANDPUR	CHURACHANDPUR SAMULAMLAN
1843	MANIPUR	CHURACHANDPUR	SINGNGAT
1844	MANIPUR	CHURACHANDPUR	THANLON
1845	MANIPUR	CHURACHANDPUR	TIPAIMUKH
1840	MANIPUR	IMPHAL EAST	IRILBUNG
1847	MANIPUR	IMPHAL EAST IMPHAL EAST	JIRIBAM
1849	MANIPUR	IMPHAL EAST IMPHAL EAST	SAWOMBUNG
1849	MANIPUR	IMPHAL EAST IMPHAL WEST	HAORANG
1850			IMC
1851	MANIPUR MANIPUR	IMPHAL WEST IMPHAL WEST	WANGOI
	MANIPUR		
1853		SENAPATI	KANGPOKPI PAOMATA
1854	MANIPUR	SENAPATI	PAOMATA PURUL
1855 1856	MANIPUR	SENAPATI SENAPATI	SAIKUL
1856	MANIPUR MANIPUR	SENAPATI SENAPATI	SAIKUL
1857	MANIPUR	SENAPATI SENAPATI	TADUBI
1858	MANIPUR	TAMENGLONG	NUNGBA
1859	MANIPUR	TAMENGLONG	TAMEI
1860	MANIPUR	TAMENGLONG	TAMEI
1861		THOUBAL	KAKCHING
1862	MANIPUR MANIPUR		THOUBAL
		THOUBAL	
1864	MANIPUR	UKHRUL	CHINGAI
1865	MANIPUR	UKHRUL	KAMJONG
1866	MANIPUR	UKHRUL	KASOM
1867 MURD Fo	MANIPUR ducation rfq/06.07.2012	UKHRUL	PHUNGYAR 94

1868	MANIPUR	UKHRUL	UKHRUL
1869	MEGHALAYA	EAST GARO HILLS	KHARKUTTA
1870	MEGHALAYA	EAST GARO HILLS	RESUBELPARA
1871	MEGHALAYA	EAST GARO HILLS	RONGJENG
1872	MEGHALAYA	EAST KHASI HILLS	KHADARSHNONG - LAITKROH
1873	MEGHALAYA	EAST KHASI HILLS	MAWKYNREW
1874	MEGHALAYA	EAST KHASI HILLS	MAWPHLANG
1875	MEGHALAYA	EAST KHASI HILLS	MAWRYNGKNENG
1876	MEGHALAYA	EAST KHASI HILLS	MAWSYNRAM
1877	MEGHALAYA	EAST KHASI HILLS	MYLLIEM
1878	MEGHALAYA	EAST KHASI HILLS	PYNURSLA
1879	MEGHALAYA	EAST KHASI HILLS	SHELLA BHOLAGANJ
1880	MEGHALAYA	JAINTIA HILLS	AMLAREM
1881	MEGHALAYA	JAINTIA HILLS	KHLIEHRIAT
1882	MEGHALAYA	JAINTIA HILLS	LASKEIN
1883	MEGHALAYA	JAINTIA HILLS	SAIPUNG
1884	MEGHALAYA	JAINTIA HILLS	THADLASKEIN
1885	MEGHALAYA	RI BHOI	JIRANG
1885	MEGHALAYA	RI BHOI	UMLING
1887	MEGHALAYA	RI BHOI	UMSNING
1888	MEGHALAYA	SOUTH GARO HILLS	СНОКРОТ
		SOUTH GARO HILLS	
1889	MEGHALAYA MEGHALAYA		GASUAPARA
1890	-	SOUTH GARO HILLS	RONGARA
1891	MEGHALAYA	WEST GARO HILLS	GAMBEGRE
1892	MEGHALAYA	WEST GARO HILLS	RONGRAM
1893	MEGHALAYA	WEST KHASI HILLS	MAIRANG
1894	MEGHALAYA	WEST KHASI HILLS	MAWKYRWAT
1895	MEGHALAYA	WEST KHASI HILLS	MAWSHYNRUT
1896	MEGHALAYA	WEST KHASI HILLS	MAWTHADRAISHAN
1897	MEGHALAYA	WEST KHASI HILLS	NONGSTOIN
1898	MEGHALAYA	WEST KHASI HILLS	RANIKOR
1899	MIZORAM	AIZAWL	AIBAWK
1900	MIZORAM	AIZAWL	BAWNGKAWN
1901	MIZORAM	AIZAWL	BUNGKAWN
1902	MIZORAM	AIZAWL	CHANMARI
1903	MIZORAM	AIZAWL	DARLAWN
1904	MIZORAM	AIZAWL	ELECTRIC
1905	MIZORAM	AIZAWL	KHAWRUHLIAN
1906	MIZORAM	AIZAWL	SAITUAL
1907	MIZORAM	AIZAWL	THINGSULTHLIAH
1908	MIZORAM	AIZAWL	VENGHLUI
1909	MIZORAM	AIZAWL	ZEMABAWK
1910	MIZORAM	CHAMPHAI	CHAMPHAI
1911	MIZORAM	CHAMPHAI	KHAWZAWL
1912	MIZORAM	CHAMPHAI	NGOPA
1913	MIZORAM	СНАМРНАІ	VANZAU
1914	MIZORAM	KOLASIB	BILKHAWTHLIR
1915	MIZORAM	KOLASIB	KAWNPUI
1916	MIZORAM	KOLASIB	KOLASIB
1917	MIZORAM	LAWNGTLAI	BARAPANSURY
1918	MIZORAM	LAWNGTLAI	BUNGTLANG
1919	MIZORAM	LAWNGTLAI	KAMALANAGAR
1920	MIZORAM	LAWNGTLAI	LAWNGTLAI
	ducation rfa/06.07.2012		95

1921	MIZORAM	LAWNGTLAI	SANGAU
1922	MIZORAM	LUNGLEI	BUNGHMUN
1923	MIZORAM	LUNGLEI	EASTERN
1924	MIZORAM	LUNGLEI	STATION
1925	MIZORAM	MAMIT	MAMIT
1926	MIZORAM	MAMIT	RAWPUICHHIP
1927	MIZORAM	MAMIT	REIEK
1928	MIZORAM	MAMIT	WEST PHAILENG
1929	MIZORAM	SAIHA	SAIHA
1929	MIZORAM	SAIHA	TUIPANG
1931	MIZORAM	SERCHHIP	LUNGPHO
1932	MIZORAM	SERCHHIP	N.VANLAIPHAI
1933	MIZORAM	SERCHHIP	SERCHHIP
1934	NAGALAND	DIMAPUR	DHANSIRIPAR
1935	NAGALAND	DIMAPUR	KUSHIABILL
1936	NAGALAND	DIMAPUR	MEDZIPHEMA
1930	NAGALAND	DIMAPUR	NUILAND
1937	NAGALAND	КОНІМА	L .KHEL KOHIMA
1939	NAGALAND	KOHIMA	NERHEMA (CHIEPHOBOZOU)
1940	NAGALAND	КОНІМА	PEREN
1940	NAGALAND	КОНІМА	TENING
1942	NAGALAND	KOHIMA	TSEMINYU
1942	NAGALAND	КОНІМА	VISWEMA
1944	NAGALAND	MOKOKCHUNG	CHANGTONGYA-B
1945	NAGALAND	MOKOKCHUNG	KHENSA
1946	NAGALAND	MOKOKCHUNG	KOBULONG
1940	NAGALAND	MOKOKCHUNG	MANGKOLEMBA-II
1947	NAGALAND	MOKOKCHUNG	MOKOKCHUNG VILLAGE
1949	NAGALAND	MON	ABOI
1949	NAGALAND	MON	MON
1950	NAGALAND	MON	TIZIT
1951	NAGALAND	PHEK	CHOZUBA
1952	NAGALAND	PHEK	MELURI
1955	NAGALAND	PHEK	PFUTSERO
1955	NAGALAND	PHEK	PHEK
1956	NAGALAND	TUENSANG	KIPHERE
1957	NAGALAND	TUENSANG	LONGKHIM
1958	NAGALAND	TUENSANG	NOKSEN
1959	NAGALAND	TUENSANG	SEYONCHUNG
1960	NAGALAND	TUENSANG	TUENSANG
1961	NAGALAND	WOKHA	BHANDARI
1962	NAGALAND	WOKHA	SANIS
1962	NAGALAND	WOKHA	WOKHA
1964	NAGALAND	ZUNHEBOTO	AGHUNATO
1965	NAGALAND	ZUNHEBOTO	AKULUTO
1965	NAGALAND	ZUNHEBOTO	РИСНОВОТО
1967	NAGALAND	ZUNHEBOTO	SATAKHA
1967	NAGALAND	ZUNHEBOTO	SURUHUTO
1969	NAGALAND	ZUNHEBOTO	ZUNHEBOTO
1970	ORISSA	ANGUL	ANGUL
1970	ORISSA	ANGUL	ATHAMALLIK
1971	ORISSA	ANGUL	BANARPAL
1972	ORISSA	ANGUL	CHHENDIPADA
	ducation rfg/06.07.2012		96

1974	ORISSA	ANGUL	KANIHA
1975	ORISSA	ANGUL	KISHORENAGAR
1976	ORISSA	ANGUL	TALCHER
1977	ORISSA	BALASORE	BAHANAGA
1978	ORISSA	BALASORE	BALIAPAL
1979	ORISSA	BALASORE	BASTA
1980	ORISSA	BALASORE	BHOGRAI
1981	ORISSA	BALASORE	JALESWAR
1982	ORISSA	BALASORE	KHAIRA
1983	ORISSA	BALASORE	REMUNA
1984	ORISSA	BALASORE	SADAR
1985	ORISSA	BALASORE	SIMULIA
1986	ORISSA	BALASORE	SORO
1987	ORISSA	BARGARH	AMBABHONA
1988	ORISSA	BARGARH	ATTABIRA
1989	ORISSA	BARGARH	BARGARH
1990	ORISSA	BARGARH	BARPALI
1990	ORISSA	BARGARH	BHATALI
1991	ORISSA	BARGARH	BHEDEN
1992	ORISSA	BARGARH	BIJEPUR
1994	ORISSA	BARGARH	PADAMPUR NAC
1995	ORISSA	BARGARH	SOHELLA
1996	ORISSA	BHADRAK	BASUDEVPUR
1997	ORISSA	BHADRAK	BHADRAK
1998	ORISSA	BHADRAK	BHANDARIPOKHARI
1999	ORISSA	BHADRAK	BONTH
2000	ORISSA	BHADRAK	CHANDABALI
2001	ORISSA	BHADRAK	DHAMNAGAR
2002	ORISSA	BHADRAK	Tihidi
2003	ORISSA	BOLANGIR	AGALPUR
2004	ORISSA	CUTTACK	ATHGARH
2005	ORISSA	CUTTACK	BANKI
2006	ORISSA	CUTTACK	BARAMBA
2007	ORISSA	CUTTACK	BARANGA
2008	ORISSA	CUTTACK	CUTTACK SADAR
2009	ORISSA	CUTTACK	DOMPARA
2010	ORISSA	CUTTACK	KANTAPARA
2011	ORISSA	CUTTACK	MAHANGA
2012	ORISSA	CUTTACK	NARSINGHPUR
2013	ORISSA	CUTTACK	NIALI
2014	ORISSA	CUTTACK	NISCHINTAKOILI
2015	ORISSA	CUTTACK	SALIPUR
2016	ORISSA	CUTTACK	TANGI
2017	ORISSA	CUTTACK	TIGIRIA
2018	ORISSA	DEOGARH	BARKOTE
2019	ORISSA	DHENKANAL	BHUBAN
2020	ORISSA	DHENKANAL	DHENKANAL SADAR
2021	ORISSA	DHENKANAL	GONDIA
2022	ORISSA	DHENKANAL	HINDOL
2023	ORISSA	DHENKANAL	KAMAKHYANAGAR
2024	ORISSA	DHENKANAL	ODAPADA
2025	ORISSA	DHENKANAL	PARJANG

2026	ORISSA	GANJAM	ASKA
2027	ORISSA	GANJAM	BELLAGUNTHA
2028	ORISSA	GANJAM	BHANJANAGAR
2029	ORISSA	GANJAM	BUGUDA
2030	ORISSA	GANJAM	JAGANATHPRASAD
2031	ORISSA	JAGATSINGHPUR	BALIKUDA
2032	ORISSA	JAGATSINGHPUR	BIRIDI
2033	ORISSA	JAGATSINGHPUR	ERASAMA
2034	ORISSA	JAGATSINGHPUR	JAGATSINGHPUR
2035	ORISSA	JAGATSINGHPUR	KUJANG
2036	ORISSA	JAGATSINGHPUR	NAUGAON
2037	ORISSA	JAGATSINGHPUR	RAGHUNATHPUR
2038	ORISSA	JAGATSINGHPUR	TIRTOL
2039	ORISSA	JAJPUR	BARCHANA
2039	ORISSA	JAJPUR	BARI
2010	ORISSA	JAJPUR	BINJHARPUR
2041	ORISSA	JAJPUR	DANAGADI
2042	ORISSA	JAJPUR	DANAGADI DASARATHPUR
2043	ORISSA	JAJPUR	DHARMASALA
2044	ORISSA	JAJPUR	JAJPUR
2045	ORISSA	JAJPUR	KORAI
2040	ORISSA	JAJPUR	RASULPUR
2047	ORISSA	JHARSUGUDA	JHARSUGUDA
2048	ORISSA	JHARSUGUDA	KIRMIRA
2049	ORISSA	JHARSUGUDA	KOLABIRA
2051	ORISSA	JHARSUGUDA	LAIKERA
2052 2053	ORISSA ORISSA	JHARSUGUDA KANDHAMAL	LAKHANPUR G.UDAYAGIRI
2033	ORISSA	KANDHAMAL	AUL
2034	ORISSA	KENDRAPARA	DERABISH
2055	ORISSA	KENDRAPARA	GARADAPUR
2030	ORISSA	KENDRAPARA	KENDRAPARA
2037	ORISSA	KENDRAPARA	MAHAKALPARA
2038	ORISSA		MARSHAGHAI
2059	ORISSA	KENDRAPARA KENDRAPARA	PATTAMUNDAI
			RAJKANIKA
2061	ORISSA	KENDRAPARA	
2062	ORISSA	KENDRAPARA	RAJNAGAR
2063	ORISSA	KEONJHAR	GHASIPURA
2064	ORISSA	KEONJHAR	HATADIHI
2065	ORISSA	KEONJHAR	PATNA
2066	ORISSA	KHORDHA	BALIANTA
2067	ORISSA	KHORDHA	BALIPATNA
2068	ORISSA	KHORDHA	BANAPUR
2069	ORISSA	KHORDHA	BEGUNIA
2070	ORISSA	KHORDHA	BHUBANESWAR
2071	ORISSA	KHORDHA	BOLGARH
2072	ORISSA	KHORDHA	CHILIKA
2073	ORISSA	KHORDHA	JATNI
2074	ORISSA	KHORDHA	KHORDHA
2075	ORISSA	KHORDHA	TANGI
2076	ORISSA	NAYAGARH	BHAPUR
2077	ORISSA	NAYAGARH	DASPALLA
2078	ORISSA ducation rfa/06.07.2012	NAYAGARH	GANIA 98

2079	ORISSA	NAYAGARH	KHANDAPADA
2080	ORISSA	NAYAGARH	NAYAGARH
2081	ORISSA	NAYAGARH	NUAGAON
2082	ORISSA	NAYAGARH	ODAGAON
2083	ORISSA	NAYAGARH	RANPUR
2083	ORISSA	PURI	ASTARANG
2085	ORISSA	PURI	BRAHMAGIRI
2085	ORISSA	PURI	DELANG
2080	ORISSA	PURI	GOP
2087	ORISSA	PURI	KAKATPUR
2089	ORISSA	PURI	KANAS
2009	ORISSA	PURI	KRUSHNAPRASAD
2090	ORISSA	PURI	NIMAPARA
2091	ORISSA	PURI	PIPLI
2092	ORISSA	PURI	PURI SADAR
2093	ORISSA	PURI	SATYABADI
2095	ORISSA	SAMBALPUR SAMBAL DUD	DHANKAUDA
2096	ORISSA ORISSA	SAMBALPUR SAMBAL DUD	JAMANKIRA
2097 2098	ORISSA	SAMBALPUR SAMBAL DUD	JUJUMURA KUCHINDA
		SAMBALPUR	
2099	ORISSA	SAMBALPUR	MANESWAR
2100	ORISSA	SAMBALPUR	NAKTIDEUL
2101	ORISSA	SAMBALPUR	RAIRAKHOL
2102	ORISSA	SAMBALPUR	RENGALI
2103	ORISSA	SONEPUR	DUNGURIPALI
2104	ORISSA	SUNDERGARH	BALISANKARA
2105	ORISSA	SUNDERGARH	BARGAON
2106	ORISSA	SUNDERGARH	BONAI
2107	ORISSA	SUNDERGARH	HEMGIRI
2108	ORISSA	SUNDERGARH	KUTRA
2109	ORISSA	SUNDERGARH	LEPHRIPARA
2110	ORISSA	SUNDERGARH	SUNDARGARH
2111	ORISSA	SUNDERGARH	TANGARPALI
2112	PUDUCHERRY	KARAIKAL	KARAIKAL
2113	PUDUCHERRY	MAHE	MAHE
2114	PUDUCHERRY	YANAM	YANAM
2115	PUNJAB	AMRITSAR	AJNALA
2116	PUNJAB	AMRITSAR	CHOGAWAN
2117	PUNJAB	AMRITSAR	HARSHAN CHINNA
2118	PUNJAB	AMRITSAR	JANDIALA GURU
2119	PUNJAB	AMRITSAR	MAJITHA
2120	PUNJAB	AMRITSAR	RAYYA
2121	PUNJAB	AMRITSAR	TARSIKKA
2122	PUNJAB	AMRITSAR	VERKA
2123	PUNJAB	BATHINDA	BATHINDA
2124	PUNJAB	BATHINDA	BHAGTA BHAI KA
2125	PUNJAB	BATHINDA	NATHANA
2126	PUNJAB	BATHINDA	PHUL
2127	PUNJAB	BATHINDA	RAMPURA
2128	PUNJAB	FARIDKOT	FARIDKOT
2129	PUNJAB	FARIDKOT	KOTKAPURA
2130	PUNJAB	FATEHGARH SAHIB	AMLOH
2131	PUNJAB	FATEHGARH SAHIB	BASSI PATHANAN 99

2132	PUNJAB	FATEHGARH SAHIB	KHAMANO
2133	PUNJAB	FATEHGARH SAHIB	KHERA
2134	PUNJAB	FATEHGARH SAHIB	SIRHIND
2135	PUNJAB	FIROZPUR	GHALL KHURD
2136	PUNJAB	FIROZPUR	GURUHARSHAI
2137	PUNJAB	FIROZPUR	MAKHU
2138	PUNJAB	FIROZPUR	ZIRA
2139	PUNJAB	GURDASPUR	BAMIAL
2140	PUNJAB	GURDASPUR	BATALA
2141	PUNJAB	GURDASPUR	DERA BABA NANAK
2142	PUNJAB	GURDASPUR	DHAR KALAN
2143	PUNJAB	GURDASPUR	DHARIWAL
2144	PUNJAB	GURDASPUR	DINANAGAR
2145	PUNJAB	GURDASPUR	DORANGLA
2145	PUNJAB	GURDASPUR	FATEHGARH CHURIAN
2140	PUNJAB	GURDASPUR	GURDASPUR
2147	PUNJAB	GURDASPUR	KAHNUWAN
2148	PUNJAB	GURDASPUR	KALANAUR
2149	PUNJAB	GURDASPUR	N.J.SINGH
2150	PUNJAB	GURDASPUR	PATHANKOT
2151	PUNJAB		
		GURDASPUR	QADIAN
2153	PUNJAB	GURDASPUR	SHRI HARGOBINDPUR
2154	PUNJAB	GURDASPUR	SUJANPUR
2155	PUNJAB	HOSHIARPUR	BHUNGA
2156	PUNJAB	HOSHIARPUR	DASUYA
2157	PUNJAB	HOSHIARPUR	GARSHANKAR
2158	PUNJAB	HOSHIARPUR	HAJIPUR
2159	PUNJAB	HOSHIARPUR	HOSHIARPUR 1
2160	PUNJAB	HOSHIARPUR	HOSHIARPUR 2
2161	PUNJAB	HOSHIARPUR	MAHILPUR
2162	PUNJAB	HOSHIARPUR	MUKERIAN
2163	PUNJAB	HOSHIARPUR	TALWARA
2164	PUNJAB	HOSHIARPUR	TANDA
2165	PUNJAB	JALANDHAR	ADAMPUR
2166	PUNJAB	JALANDHAR	BHOGPUR
2167	PUNJAB	JALANDHAR	JALANDHAR EAST
2168	PUNJAB	JALANDHAR	JALANDHAR WEST
2169	PUNJAB	JALANDHAR	LOHIAN KHASS
2170	PUNJAB	JALANDHAR	NAKODAR
2171	PUNJAB	JALANDHAR	NOORMAHAL
2172	PUNJAB	JALANDHAR	PHILLOUR
2173	PUNJAB	JALANDHAR	RURKA KALAN
2174	PUNJAB	JALANDHAR	SHAHKOT
2175	PUNJAB	KAPURTHALA	DHILWAN
2176	PUNJAB	KAPURTHALA	KAPURTHALA
2177	PUNJAB	KAPURTHALA	NADALA
2178	PUNJAB	KAPURTHALA	PHAGWARA
2179	PUNJAB	KAPURTHALA	SULTANPUR
2180	PUNJAB	LUDHIANA	DEHLON
2181	PUNJAB	LUDHIANA	DORAHA
2182	PUNJAB	LUDHIANA	JAGRAON
2183	PUNJAB	LUDHIANA	KHANNA
2184	PUNJAB	LUDHIANA	LUDHIANA-1

2185	PUNJAB	LUDHIANA	LUDHIANA-2
2186	PUNJAB	LUDHIANA	MACCHIWARA
2187	PUNJAB	LUDHIANA	PAKHOWAL
2188	PUNJAB	LUDHIANA	RAIKOT
2189	PUNJAB	LUDHIANA	SAMRALA
2109	PUNJAB	LUDHIANA	SIDHWAN BET
2190	PUNJAB	LUDHIANA	SUDHAR
2191	PUNJAB	MOGA	BAGHA PURANA
2192	PUNJAB	MOGA	DHARAMKOT
2193	PUNJAB	MOGA	MOGA 2
2194	PUNJAB	MOGA	MOGA 1
2195	PUNJAB	MOGA	NIHAL SINGH WALA
2190	PUNJAB	MOGA	DERABASSI
2197	PUNJAB	MOHALI	KHARAR
2198			
	PUNJAB	MOHALI	MAJRI
2200	PUNJAB	MOHALI	MORINDA
2201	PUNJAB	MUKTSAR	GIDDERBAHA
2202	PUNJAB	MUKTSAR	MALOUT
2203	PUNJAB	NAWANSHAHR	AUR
2204	PUNJAB	NAWANSHAHR	BALACHAUR
2205	PUNJAB	NAWANSHAHR	BANGA
2206	PUNJAB	NAWANSHAHR	NAWAN SHAHR
2207	PUNJAB	NAWANSHAHR	SAROYA
2208	PUNJAB	PATIALA	BHUNERHERI
2209	PUNJAB	PATIALA	GHANAUR
2210	PUNJAB	PATIALA	NABHA
2211	PUNJAB	PATIALA	PATIALA
2212	PUNJAB	PATIALA	RAJPURA
2213	PUNJAB	PATIALA	SAMANA
2214	PUNJAB	PATIALA	SANAUR
2215	PUNJAB	ROPAR	ANANDPUR SAHIB
2216	PUNJAB	ROPAR	CHAMKAUR SAHIB
2217	PUNJAB	ROPAR	MORINDA
2218	PUNJAB	ROPAR	NURPUR BEDI
2219	PUNJAB	ROPAR	ROPAR
2220	PUNJAB	SANGRUR	AHMEDGARH
2221	PUNJAB	SANGRUR	BARNALA
2222	PUNJAB	SANGRUR	BHAWANIGARH
2223	PUNJAB	SANGRUR	DHURI
2224	PUNJAB	SANGRUR	MALERKOTLA
2225	PUNJAB	SANGRUR	MEHALKALAN
2226	PUNJAB	SANGRUR	SANGRUR
2227	PUNJAB	SANGRUR	SEHNA
2228	PUNJAB	SANGRUR	SHERPUR
2229	PUNJAB	TARAN TARAN	BHIKHIWIND
2230	PUNJAB	TARAN TARAN	CHOHLA SAHIB
2231	PUNJAB	TARAN TARAN	GANDIWIND
2232	PUNJAB	TARAN TARAN	KHADOOR SAHIB
2233	PUNJAB	TARAN TARAN	NAUSHEHRA PANNUAN
2234	PUNJAB	TARAN TARAN	PATTI
2235	PUNJAB	TARAN TARAN	TARAN TARAN
2236	RAJASTHAN	AJMER	JAWAJA
2237	RAJASTHAN	ALWAR	ALWAR
	ducation_rfa/06.07.2012	L	101

2238	RAJASTHAN	ALWAR	BEHRORE
2239	RAJASTHAN	ALWAR	KOTKASIM
2240	RAJASTHAN	ALWAR	MUNDAWAR
2241	RAJASTHAN	ALWAR	NEEMRANA
2242	RAJASTHAN	BANSWARA	TALWARA
2243	RAJASTHAN	BARMER	DHORIMANA
2243	RAJASTHAN	BARMER	SINDHARI
2244	RAJASTHAN	BHARATPUR	ROOPWAS
2245	RAJASTHAN	CHURU	CHURU
2240	RAJASTHAN	CHURU	RAJGARH
2247	RAJASTHAN	CHURU	RATANGARH
2249 2250	RAJASTHAN	CHURU	SUJANGARH TADANACAD
	RAJASTHAN	CHURU	TARANAGAR
2251	RAJASTHAN	DAUSA	BANDIKUI
2252	RAJASTHAN	GANGANAGAR	GHARSANA
2253	RAJASTHAN	GANGANAGAR	KARANPUR
2254	RAJASTHAN	GANGANAGAR	PADAMPUR
2255	RAJASTHAN	GANGANAGAR	RAISINGHNAGAR
2256	RAJASTHAN	GANGANAGAR	SADULSHAHAR
2257	RAJASTHAN	GANGANAGAR	SRI GANGANAGAR
2258	RAJASTHAN	GANGANAGAR	VIJAYNAGAR
2259	RAJASTHAN	HANUMANGARH	BHADRA
2260	RAJASTHAN	HANUMANGARH	NOHAR
2261	RAJASTHAN	HANUMANGARH	PILIBANGAN
2262	RAJASTHAN	HANUMANGARH	RAWATSAR
2263	RAJASTHAN	HANUMANGARH	SANGRIA
2264	RAJASTHAN	HANUMANGARH	TIBBI
2265	RAJASTHAN	JAIPUR	GOVINDGARH
2266	RAJASTHAN	JAIPUR	JHOTWARA
2267	RAJASTHAN	JAIPUR	KOTPUTLI
2268	RAJASTHAN	JAIPUR	SANMBHER LAKE
2269	RAJASTHAN	JAIPUR	SHAHPURA
2270	RAJASTHAN	JAIPUR	VIRATNAGAR
2271	RAJASTHAN	JHALAWAR	AKLERA
2272	RAJASTHAN	JHALAWAR	BHAWANIMANDI
2273	RAJASTHAN	JHALAWAR	JHALAWAR
2274	RAJASTHAN	JHALAWAR	KHANPUR
2275	RAJASTHAN	JHALAWAR	PIRAWA
2276	RAJASTHAN	JHALAWAR	SUNEL
2277	RAJASTHAN	JHUNJHUNUN	ALSISAR
2278	RAJASTHAN	JHUNJHUNUN	BAGAR
2279	RAJASTHAN	JHUNJHUNUN	BISAU
2280	RAJASTHAN	JHUNJHUNUN	BUHANA
2280	RAJASTHAN	JHUNJHUNUN	CHIRAWA
2281	RAJASTHAN	JHUNJHUNUN	JHUNJHUNU
2282	RAJASTHAN	JHUNJHUNUN	KHETRI
2283	RAJASTHAN	JHUNJHUNUN	NAWALGARH
2284	RAJASTHAN	JHUNJHUNUN	SURAJGARH
2285	RAJASTHAN	JHUNJHUNUN	UDAIPURWATI
2280	RAJASTHAN	KARAULI	HINDAUN
2287	RAJASTHAN	KOTA	KHERABAD
2288	RAJASTHAN	KOTA	KOTA
2290	RAJASTHAN ducation rfa/06.07.2012	КОТА	LADPURA 102

2291	RAJASTHAN	КОТА	SANGOD
2292	RAJASTHAN	КОТА	SULTANPUR
2293	RAJASTHAN	NAGAUR	KUCHAMAN
2294	RAJASTHAN	SIKAR	DANTARAMGARH
2295	RAJASTHAN	SIKAR	DHOD
2296	RAJASTHAN	SIKAR	FATEHPUR
2297	RAJASTHAN	SIKAR	KHANDELA
2298	RAJASTHAN	SIKAR	LAXMANGARH
2299	RAJASTHAN	SIKAR	NEEM KA THANA
2300	RAJASTHAN	SIKAR	PIPRALI
2301	RAJASTHAN	SIKAR	SHRI MADHOPUR
2302	RAJASTHAN	SIKAR	SIKAR SHAHAE
2302	RAJASTHAN	UDAIPUR	BADGAON
2303	SIKKIM	EAST SIKKIM	GANGTOK-A
2304	SIKKIM	EAST SIKKIM	PAKYONG
2305	SIKKIM	EAST SIKKIM	RONGLI
2307	SIKKIM	NORTH SIKKIM	CHUNGTHANG
2307	SIKKIM	NORTH SIKKIM	MANGAN
2308	SIKKIM	SOUTH SIKKIM	NAMCHI
2309	SIKKIM	SOUTH SIKKIM	RAVANGLA
2310	SIKKIM	WEST SIKKIM	GYALSHING
2311	SIKKIM	WEST SIKKIM	SORENG
2312	TAMIL NADU	CHENNAI	ADYAR
2313	TAMIL NADU	CHENNAI	EGMORE
2314	TAMIL NADU	CHENNAI	MYLAPORE
2316	TAMIL NADU	CHENNAI	PERAMBUR
2317	TAMIL NADU	CHENNAI	PERIAMEDU PURASAIWALKKAM
2318	TAMIL NADU	CHENNAI	ROYAPURAM
2319 2320	TAMIL NADU TAMIL NADU	CHENNAI CHENNAI	T.NAGAR
2320			
2321	TAMIL NADU TAMIL NADU	CHENNAI COIMBATORE	TRIPLICANE ANAIMALAI
2322	TAMIL NADU	COIMBATORE	ANAIMALAI
2323	TAMIL NADU	COIMBATORE	AVINASHI
2324	TAMIL NADU TAMIL NADU	COIMBATORE	GUDIMANGALAM
2326	TAMIL NADU	COIMBATORE	KARAMADAI
2327 2328	TAMIL NADU	COIMBATORE COIMBATORE	KINATHUKKADAVU MADATHUKKULAM
	TAMIL NADU		
2329	TAMIL NADU	COIMBATORE	MADUKKARAI P.N.PALAYAM
2330	TAMIL NADU	COIMBATORE	P.N.PALAYAM PALLADAM
2331	TAMIL NADU TAMIL NADU	COIMBATORE	
2332 2333	TAMIL NADU TAMIL NADU	COIMBATORE	PERUR POLLACHENOPTH
	TAMIL NADU TAMIL NADU	COIMBATORE COIMBATORE	POLLACHI NORTH
2334			POLLACHI SOUTH
2335	TAMIL NADU	COIMBATORE	PONGALUR S S KULAM
2336	TAMIL NADU	COIMBATORE	S.S.KULAM
2337	TAMIL NADU	COIMBATORE	SULTANPET
2338	TAMIL NADU	COIMBATORE	SULUR THONDAMUTHUR
2339	TAMIL NADU	COIMBATORE	THONDAMUTHUR
2340	TAMIL NADU	COIMBATORE	TIRUPPUR
2341	TAMIL NADU	COIMBATORE	UDUMALPET
2342	TAMIL NADU	COIMBATORE	VALPARAI
2343	TAMIL NADU	CUDDALORE	ANNAGRAMAM 103

2344	TAMIL NADU	CUDDALORE	BHUVANAGIRI
2345	TAMIL NADU	CUDDALORE	CUDDALORE
2346	TAMIL NADU	CUDDALORE	KAMMAPURAM
2347	TAMIL NADU	CUDDALORE	KATTUMANNARKOIL
2348	TAMIL NADU	CUDDALORE	KEERAPALAYAM
2349	TAMIL NADU	CUDDALORE	KUMARATCHI
2350	TAMIL NADU	CUDDALORE	KURINJIPADI
2350	TAMIL NADU	CUDDALORE	NEYVELI
2352	TAMIL NADU	CUDDALORE	PARANGIPETTAI
2352	TAMIL NADU	CUDDALORE	VIRDDHACHALAM
2353	TAMIL NADU	DHARMAPURI	DHARMAPURI
2355	TAMIL NADU	DHARMAPURI	HARUR
2355	TAMIL NADU	DHARMAPURI	MORAPPUR
2350	TAMIL NADU	DHARMAPURI	NALLAMPALLI
2358	TAMIL NADU	DHARMAPURI	PAPPIREDDIPATTY
2359	TAMIL NADU	DINDIGUL	ATHOOR
2360	TAMIL NADU	DINDIGUL	BATLAGUNDU
2361	TAMIL NADU	DINDIGUL	DINDIGUL
2361	TAMIL NADU	DINDIGUL	GUJILIAMPARAI
2362	TAMIL NADU	DINDIGUL	KODIKANAL
2363	TAMIL NADU	DINDIGUL	NATHAM
2365	TAMIL NADU	DINDIGUL	NILAKKOTTAI
2365	TAMIL NADU	DINDIGUL	ODDANCHATRAM
2367	TAMIL NADU	DINDIGUL	PALANI
2368	TAMIL NADU	DINDIGUL	REDDIARCHATRAM
2369	TAMIL NADU	DINDIGUL	SANARPATTY
2309	TAMIL NADU	DINDIGUL	THOPPAMPATTY
2370	TAMIL NADU	DINDIGUL	VADAMADURAI
2371	TAMIL NADU	DINDIGUL	VEDASANDUR
2372	TAMIL NADU	ERODE	BHAVANI
2373	TAMIL NADU	ERODE	BHAVANI BHAVANI SAGAR
2374	TAMIL NADU	ERODE	CHENNIMALAI
2375	TAMIL NADU	ERODE	DHARAPURAM
2370	TAMIL NADU	ERODE	ERODE
2378	TAMIL NADU	ERODE	GOBI
2378	TAMIL NADU	ERODE	KANGEYAM
2379	TAMIL NADU	ERODE	KODUMUDI
2380	TAMIL NADU	ERODE	MODAKURICHI
2381	TAMIL NADU	ERODE	PERUNDURAI
2382	TAMIL NADU	ERODE	T.N. PALAYAM
2383	TAMIL NADU	ERODE	UTHUKULI
2384	TAMIL NADU	ERODE	VELLOKOIL
2385	TAMIL NADU	KANCHEEPURAM	ACHARAPAKKAM
2380	TAMIL NADU	KANCHEEPURAM	CHITHAMUR
2387	TAMIL NADU	KANCHEEPURAM	KANCHIPURAM
2388	TAMIL NADU	KANCHEEPURAM	KANCHIFUKAM
2389	TAMIL NADU	KANCHEEPURAM	KATTANKOLATHUK KUNDRATHUR
2390	TAMIL NADU	KANCHEEPURAM	LATHUR
2391	TAMIL NADU	KANCHEEPURAM	MADURANTHAGAM
2392	TAMIL NADU	KANCHEEPURAM	SRIPERUMPUDUR
2393	TAMIL NADU	KANCHEEPURAM	ST.THOMAS MOUNT
2394	TAMIL NADU	KANCHEEPURAM	THIRUKALLUKUNDRAM
2395	TAMIL NADU	KANCHEEPURAM	THIRUPORUR
	ducation rfa/06.07.2012		104

2397	TAMIL NADU	KANCHEEPURAM	UTHIRAMERUR
2398	TAMIL NADU	KANCHEEPURAM	WALAJABAD
2399	TAMIL NADU	KANNIYAKUMARI	AGASTHEESWARAM
2400	TAMIL NADU	KANNIYAKUMARI	KILLIYOOR
2401	TAMIL NADU	KANNIYAKUMARI	KURUNTHANCODE
2402	TAMIL NADU	KANNIYAKUMARI	MELPURAM
2403	TAMIL NADU	KANNIYAKUMARI	MUNCHIRAI
2404	TAMIL NADU	KANNIYAKUMARI	RAJAKKAMANGALAM
2405	TAMIL NADU	KANNIYAKUMARI	THIRUVATTAR
2406	TAMIL NADU	KANNIYAKUMARI	THOVALAI
2407	TAMIL NADU	KANNIYAKUMARI	THUCKALAY
2408	TAMIL NADU	KARUR	ARAVAKURICHI
2409	TAMIL NADU	KARUR	K.PARAMATHI
2410	TAMIL NADU	KARUR	KARUR
2411	TAMIL NADU	KARUR	KRISHNARAYAPURAM
2412	TAMIL NADU	KARUR	KULITHALAI
2413	TAMIL NADU	KARUR	THANTHONI
2413	TAMIL NADU	KARUR	THOGAIMALAI
2414	TAMIL NADU	KRISHANAGIRI	BARGUR
2413	TAMIL NADU	KRISHANAGIRI	HOSUR
2417	TAMIL NADU	KRISHANAGIRI	KAVERIPATINAM
2417	TAMIL NADU	KRISHANAGIRI	KRISHNAGIRI
2418	TAMIL NADU	KRISHANAGIRI	MATHUR
2419	TAMIL NADU	KRISHANAGIRI	UTHANGARAI
2420	TAMIL NADU	MADURAI	ALANGANALLUR
2422	TAMIL NADU	MADURAI	CHELLAMPATTI
2423	TAMIL NADU	MADURAI	KALLIKUDI
2424	TAMIL NADU	MADURAI	KOTTAMPATTI
2425	TAMIL NADU	MADURAI	MADURAI EAST
2426	TAMIL NADU	MADURAI	MADURAI NORTH
2427	TAMIL NADU	MADURAI	MADURAI SOUTH
2428	TAMIL NADU	MADURAI	MADURAI WEST
2429	TAMIL NADU	MADURAI	MELUR
2430	TAMIL NADU	MADURAI	SEDAPATTI
2431	TAMIL NADU	MADURAI	T. VADIPATTI
2432	TAMIL NADU	MADURAI	T.KALLUPATTI
2433	TAMIL NADU	MADURAI	THIRUMANGALAM
2434	TAMIL NADU	MADURAI	THIRUPARAMKUNDRAM
2435	TAMIL NADU	MADURAI	USILAMPATTI
2436	TAMIL NADU	NAGAPATTINAM	KEEZHAIYUR
2437	TAMIL NADU	NAGAPATTINAM	KELVELUR
2438	TAMIL NADU	NAGAPATTINAM	KOLLIDAM
2439	TAMIL NADU	NAGAPATTINAM	KUTHALAM
2440	TAMIL NADU	NAGAPATTINAM	MAYILADUDURAI
2441	TAMIL NADU	NAGAPATTINAM	NAGAPATTINAM
2442	TAMIL NADU	NAGAPATTINAM	SEMBANARKOIL
2443	TAMIL NADU	NAGAPATTINAM	SIRKALI
2444	TAMIL NADU	NAGAPATTINAM	THALAINAYAR
2445	TAMIL NADU	NAGAPATTINAM	THIRUMARUGAL
2446	TAMIL NADU	NAGAPATTINAM	VEDARANYAM
2447	TAMIL NADU	NAMAKKAL	ELACHIPALAYAM
2448	TAMIL NADU	NAMAKKAL	ERUMAPATTI
2449	TAMIL NADU	NAMAKKAL	KABILARMALAI

2450	TAMIL NADU	NAMAKKAL	MOHANUR
2451	TAMIL NADU	NAMAKKAL	NAMAGIRIPETTAI
2452	TAMIL NADU	NAMAKKAL	NAMAKKAL
2453	TAMIL NADU	NAMAKKAL	PALLIPALAYAM
2454	TAMIL NADU	NAMAKKAL	PARAMATHY
2455	TAMIL NADU	NAMAKKAL	PUDUCHADRAM
2456	TAMIL NADU	NAMAKKAL	RASIPURAM
2450	TAMIL NADU	NAMAKKAL	SENDAMANGALAM
2457	TAMIL NADU	NAMAKKAL	TRIUCHENGODU
2459	TAMIL NADU	NAMAKKAL	VENNANDUR
2460	TAMIL NADU	PERAMBALUR	ALATHUR
2460	TAMIL NADU	PERAMBALUR	ANDIMADAM
2461	TAMIL NADU	PERAMBALUR	JAYANKONDAM
2462	TAMIL NADU	PERAMBALUR	PERAMBALUR
2463	TAMIL NADU	PERAMBALUR	SENDURAI
2464	TAMIL NADU	PERAMBALUR	T.PALUR
2466	TAMIL NADU	PERAMBALUR	THIRUMANUR
2467	TAMIL NADU	PERAMBALUR	VEPPANTHATTAI
2468	TAMIL NADU	PUDUKKOTTAI	ANNAVASAL
2469	TAMIL NADU	PUDUKKOTTAI	ARANTHANGI
2470	TAMIL NADU	PUDUKKOTTAI	ARIMALAM
2471	TAMIL NADU	PUDUKKOTTAI	AVUDAIYARKOIL
2472	TAMIL NADU	PUDUKKOTTAI	GANDARVAKKOTTAI
2473	TAMIL NADU	PUDUKKOTTAI	KARAMBAKUDI
2474	TAMIL NADU	PUDUKKOTTAI	KUNNANDARKOIL
2475	TAMIL NADU	PUDUKKOTTAI	MANAMELKUDI
2476	TAMIL NADU	PUDUKKOTTAI	PONNAMARAVATHY
2477	TAMIL NADU	PUDUKKOTTAI	PUDUKKOTTAI
2478	TAMIL NADU	PUDUKKOTTAI	THIRUMAYAM
2479	TAMIL NADU	PUDUKKOTTAI	THIRUVARANKULAM
2480	TAMIL NADU	PUDUKKOTTAI	VIRALIMALAI
2481	TAMIL NADU	RAMANATHAPURAM	BOGALUR
2482	TAMIL NADU	RAMANATHAPURAM	KADALADI
2483	TAMIL NADU	RAMANATHAPURAM	KAMUTHI
2484	TAMIL NADU	RAMANATHAPURAM	MANDAPAM
2485	TAMIL NADU	RAMANATHAPURAM	MUDUKULATHUR
2486	TAMIL NADU	RAMANATHAPURAM	NAINARKOIL
2487	TAMIL NADU	RAMANATHAPURAM	PARAMAKUDI
2488	TAMIL NADU	RAMANATHAPURAM	RAJASINGAMANGALAM
2489	TAMIL NADU	RAMANATHAPURAM	RAMANATHAPURAM
2490	TAMIL NADU	RAMANATHAPURAM	THIRUPPULLANI
2491	TAMIL NADU	RAMANATHAPURAM	THIRUVADANAI
2492	TAMIL NADU	SALEM	ATTUR
2493	TAMIL NADU	SALEM	AYOTHIYAPATTINAM
2494	TAMIL NADU	SALEM	GANGAVALLI
2495	TAMIL NADU	SALEM	MECHERI
2496	TAMIL NADU	SALEM	OMALUR
2497	TAMIL NADU	SALEM	SALEMRURAL
2498	TAMIL NADU	SALEM	THALAIVASAL
2499	TAMIL NADU	SALEM	VALAPPADI
2500	TAMIL NADU	SIVAGANGA	DEVAKOTTAI
2501	TAMIL NADU	SIVAGANGA	IIAYANKUDI
2502	TAMIL NADU	SIVAGANGA	KALAYARKOIL

2503	TAMIL NADU	SIVAGANGA	KALLAL
2504	TAMIL NADU	SIVAGANGA	KANNANGUDI
2505	TAMIL NADU	SIVAGANGA	MANAMADURAI
2506	TAMIL NADU	SIVAGANGA	SAKKOTTAI
2507	TAMIL NADU	SIVAGANGA	SINGAMPUNARI
2508	TAMIL NADU	SIVAGANGA	SIVAGANGAI
2509	TAMIL NADU	SIVAGANGA	THIRUPPATUR
2510	TAMIL NADU	SIVAGANGA	THIRUPPUVANAM
2511	TAMIL NADU	THANJAVUR	AMMAPETTAI
2512	TAMIL NADU	THANJAVUR	BUDALUR
2513	TAMIL NADU	THANJAVUR	KUMBAKONAM
2514	TAMIL NADU	THANJAVUR	MADUKKUR
2515	TAMIL NADU	THANJAVUR	ORATHANDU
2516	TAMIL NADU	THANJAVUR	PAPANASAM
2510	TAMIL NADU	THANJAVUR	ΡΑΤΤυΚΚΟΤΤΑΙ
2518	TAMIL NADU	THANJAVUR	PERAVURANI
2510	TAMIL NADU	THANJAVUR	SETHUBAVACHATRAM
2520	TAMIL NADU	THANJAVUR	THANJAVUR (RURAL)
2521	TAMIL NADU	THANJAVUR	THIRUVAIYARU
2521	TAMIL NADU	THANJAVUR	THIRUVIDAIMARUDHUR
2522	TAMIL NADU	THANJAVUR	THIRUVONAM
2523	TAMIL NADU	THANJAVUR	TIRUPANADAL
2525	TAMIL NADU	THE NILGIRIS	COONOOR
2526	TAMIL NADU	THE NILGIRIS	GUDALUR
2527	TAMIL NADU	THE NILGIRIS	KOTAGIRI
2528	TAMIL NADU	THE NILGIRIS	UDHAGAMANDALAM
2528	TAMIL NADU	THENI	AUNDIPATTY
2529	TAMIL NADU	THENI	BODINAYAKANUR
2530	TAMIL NADU	THENI	CHINNAMANUR
2532	TAMIL NADU	THENI	CUMBUM
2532	TAMIL NADU	THENI	MYLADUMPARAI
2533	TAMIL NADU	THENI	PERIYAKULAM
2534	TAMIL NADU	THENI	THENI
2535	TAMIL NADU	THENI	UTHAMAPALAYAM
2536	TAMIL NADU TAMIL NADU	THIRUVALLUR	ELLAPURAM
2538	TAMIL NADU	THIRUVALLUR	
2538	TAMIL NADU TAMIL NADU		GUMMIDIPOONDI KADAMBATHUR
2539	TAMIL NADU	THIRUVALLUR	MINJUR
	TAMIL NADU	THIRUVALLUR	
2541	TAMIL NADU	THIRUVALLUR	PALLIPAT POONAMALLEE
2542		THIRUVALLUR	POONAMALLEE
2543 2544	TAMIL NADU	THIRUVALLUR	POONDI
	TAMIL NADU	THIRUVALLUR	PUZHAL
2545	TAMIL NADU	THIRUVALLUR	R.K.PET SHOLAVARAM
2546	TAMIL NADU	THIRUVALLUR	
2547	TAMIL NADU	THIRUVALLUR	THIRUVALANGADU
2548	TAMIL NADU	THIRUVALLUR	THIRUVALLUR
2549	TAMIL NADU	THIRUVALLUR	TIRUTTANI
2550	TAMIL NADU	THIRUVALLUR	VILLIVAKKAM
2551	TAMIL NADU	THIRUVARUR	KODAVASAL
2552	TAMIL NADU	THIRUVARUR	KORADACHERY
2553	TAMIL NADU	THIRUVARUR	KOTTUR
2554	TAMIL NADU	THIRUVARUR	MANNARGUDI
2555	TAMIL NADU	THIRUVARUR	MUTHUPETTAI 107

2556	TAMIL NADU	THIRUVARUR	NANNILAM
2557	TAMIL NADU	THIRUVARUR	NEEDAMANGALAM
2558	TAMIL NADU	THIRUVARUR	THIRUTHURAIPOONDI
2559	TAMIL NADU	THIRUVARUR	TIRUVARUR
2560	TAMIL NADU	THIRUVARUR	VALANGAIMAN
2561	TAMIL NADU	THOOTHUKKUDI	ALWARTHIRUNAGARI
2562	TAMIL NADU	THOOTHUKKUDI	KARUNGULAM
2563	TAMIL NADU	ТНООТНИККИОІ	KAYATHAR
2564	TAMIL NADU	ТНООТНИККИОІ	KOVILPATTI RURAL, URBAN
2565	TAMIL NADU	ТНООТНИККИДІ	OTTAPIDARAM RURAL
2566	TAMIL NADU	ТНООТНИККИДІ	PUDUR
2567	TAMIL NADU	ТНООТНИККИДІ	SATTANKULAM
2568	TAMIL NADU	THOOTHUKKUDI	SRIVAIKUNTAM
2569	TAMIL NADU	THOOTHUKKUDI	THOOTHUKUDI RURAL
2570	TAMIL NADU	THOOTHUKKUDI	TIRUCHENDUR
2570	TAMIL NADU	THOOTHUKKUDI	TUTICORIN
2572	TAMIL NADU	THOOTHUKKUDI	UDANGUDY
2573	TAMIL NADU	THOOTHUKKUDI	VILATHIKULAM
2574	TAMIL NADU	TIRUCHIRAPPALLI	ANDHANALLUR
2575	TAMIL NADU	TIRUCHIRAPPALLI	LALGUDI
2576	TAMIL NADU	TIRUCHIRAPPALLI	MANAPPARAI
2577	TAMIL NADU	TIRUCHIRAPPALLI	MANIKANDAM
2578	TAMIL NADU	TIRUCHIRAPPALLI	MANNACHANNALLUR
2579	TAMIL NADU	TIRUCHIRAPPALLI	MARUNGAPURI
2580	TAMIL NADU	TIRUCHIRAPPALLI	MUSIRI
2581	TAMIL NADU	TIRUCHIRAPPALLI	PULLAMBADI
2582	TAMIL NADU	TIRUCHIRAPPALLI	THATHIENGARPET
2583	TAMIL NADU	TIRUCHIRAPPALLI	THIRUVERUMBUR
2584	TAMIL NADU	TIRUCHIRAPPALLI	THOTTIAM
2585	TAMIL NADU	TIRUCHIRAPPALLI	THURAIYUR
2586	TAMIL NADU	TIRUCHIRAPPALLI	TRICHY-WEST
2587	TAMIL NADU	TIRUCHIRAPPALLI	UPPILIAPURAM
2588	TAMIL NADU	TIRUCHIRAPPALLI	VAIYAMPATTY
2589	TAMIL NADU	TIRUNELVELI	ALANGULAM
2590	TAMIL NADU	TIRUNELVELI	AMBASAMUDRAM
2591	TAMIL NADU	TIRUNELVELI	CHERANMAHADEVI
2592	TAMIL NADU	TIRUNELVELI	KADAYAM
2593	TAMIL NADU	TIRUNELVELI	KADAYANALLUR
2594	TAMIL NADU	TIRUNELVELI	KALAKAD
2595	TAMIL NADU	TIRUNELVELI	KEELAPAVOOR
2596	TAMIL NADU	TIRUNELVELI	KURUVIKULAM
2597	TAMIL NADU	TIRUNELVELI	MANUR
2598	TAMIL NADU	TIRUNELVELI	MELANEELITHANALLUR
2599	TAMIL NADU	TIRUNELVELI	NANGUNERI
2600	TAMIL NADU	TIRUNELVELI	PALAYAMKOTTAI RURAL
2601	TAMIL NADU	TIRUNELVELI	PAPPAKUDI
2602	TAMIL NADU	TIRUNELVELI	RADHAPURAM
2603	TAMIL NADU	TIRUNELVELI	SANKARANKOVIL
2604	TAMIL NADU	TIRUNELVELI	SHENGOTTAI
		TIRUNELVELI	TENKASI
2605	TAMIL NADU	TIKUNELVELI	
2605 2606	TAMIL NADU TAMIL NADU		
2605 2606 2607	TAMIL NADU TAMIL NADU TAMIL NADU	TIRUNELVELI TIRUNELVELI TIRUNELVELI	VALLIYOOR VASUDEVANALLUR

2609	TAMIL NADU	TIRUVANNAMALAI	ARNI
2610	TAMIL NADU	TIRUVANNAMALAI	CHENGAM
2611	TAMIL NADU	TIRUVANNAMALAI	СНЕТРЕТ
2612	TAMIL NADU	TIRUVANNAMALAI	CHEYYAR
2612	TAMIL NADU	TIRUVANNAMALAI	KALASAPAKKAM
2613	TAMIL NADU	TIRUVANNAMALAI	KILPENNATHUR
2615	TAMIL NADU	TIRUVANNAMALAI	PERNAMALLUR
2616	TAMIL NADU	TIRUVANNAMALAI	POLUR
2617	TAMIL NADU	TIRUVANNAMALAI	PUDUPALAYAM
2618	TAMIL NADU	TIRUVANNAMALAI	THANDRAMPET
2619	TAMIL NADU	TIRUVANNAMALAI	THELLAR
2620	TAMIL NADU	TIRUVANNAMALAI	THURINJAPURAM
2621	TAMIL NADU	TIRUVANNAMALAI	TIRUVANNAMALAI
2621			
	TAMIL NADU	TIRUVANNAMALAI	VANDAVASI
2623	TAMIL NADU	TIRUVANNAMALAI	VEMBAKKAM
2624	TAMIL NADU	TIRUVANNAMALAI	WEST ARNI
2625	TAMIL NADU	VELLORE	ALANGAYAM
2626	TAMIL NADU	VELLORE	ANAICUT
2627	TAMIL NADU	VELLORE	ARAKKONAM
2628	TAMIL NADU	VELLORE	ARCOT
2629	TAMIL NADU	VELLORE	GUDIYATTAM
2630	TAMIL NADU	VELLORE	JOLARPET
2631	TAMIL NADU	VELLORE	K V KUPPAM
2632	TAMIL NADU	VELLORE	KANDHILI
2633	TAMIL NADU	VELLORE	KANIYAMBADI
2634	TAMIL NADU	VELLORE	KATPADI
2635	TAMIL NADU	VELLORE	KAVERIPAKKAM
2636	TAMIL NADU	VELLORE	MADHANUR
2637	TAMIL NADU	VELLORE	NATRAMPALLI
2638	TAMIL NADU	VELLORE	NEMILI
2639	TAMIL NADU	VELLORE	PERNAMBUT
2640	TAMIL NADU	VELLORE	SHOLINGUR
2641	TAMIL NADU	VELLORE	THIMIRI
2642	TAMIL NADU	VELLORE	THIRUPATTUR
2643	TAMIL NADU	VELLORE	VELLORE RURAL
2644	TAMIL NADU	VELLORE	WALAJAH EAST
2645	TAMIL NADU	VELLORE	WALAJAH WEST
2646	TAMIL NADU	VILUPPURAM	CHINNASALEM
2647	TAMIL NADU	VILUPPURAM	GINGEE
2648	TAMIL NADU	VILUPPURAM	KANAI
2649	TAMIL NADU	VILUPPURAM	KANDAMANGALAM
2650	TAMIL NADU	VILUPPURAM	KOLIYANUR
2651	TAMIL NADU	VILUPPURAM	MAILAM
2652	TAMIL NADU	VILUPPURAM	MARAKKANAM
2653	TAMIL NADU	VILUPPURAM	MELMALAYANUR
2654	TAMIL NADU	VILUPPURAM	MUGAIYUR
2655	TAMIL NADU	VILUPPURAM	OLAKKUR
2656	TAMIL NADU	VILUPPURAM	VALLAM
2657	TAMIL NADU	VILUPPURAM	VANUR
2658	TAMIL NADU	VILUPPURAM	VIKARAVANDI
2659	TAMIL NADU	VILUPPURAM	VILUPPURAM
2660	TAMIL NADU	VIRUDHUNAGAR	ARUPPUKOTTAI
2661	TAMIL NADU	VIRUDHUNAGAR	KARIYAPATTI
	ducation rfa/06.07.2012	•	109

2662	TAMIL NADU	VIRUDHUNAGAR	NARIKUDI
2663	TAMIL NADU	VIRUDHUNAGAR	RAJAPALAYAM
2664	TAMIL NADU	VIRUDHUNAGAR	SATTUR
2665	TAMIL NADU	VIRUDHUNAGAR	SIVAKASI
2666	TAMIL NADU	VIRUDHUNAGAR	SRIVILLIPUTHUR
2667	TAMIL NADU	VIRUDHUNAGAR	THIRUCHULI
2668	TAMIL NADU	VIRUDHUNAGAR	VEMBAKOTTAI
2669	TAMIL NADU	VIRUDHUNAGAR	VIRUDHUNAGAR
2670	TAMIL NADU	VIRUDHUNAGAR	WATRAP
2671	TRIPURA	DHALAI	MANU
2672	TRIPURA	DHALAI	SALEMA
2672	TRIPURA	NORTH TRIPURA	GOURNAGAR
2674	TRIPURA	NORTH TRIPURA	JAMPUI
2675	TRIPURA	NORTH TRIPURA	KADAMTALA
2676	TRIPURA	NORTH TRIPURA	KUMARGHAT
2677	TRIPURA	NORTH TRIPURA	PANISAGAR
2678	TRIPURA	NORTH TRIPURA	PECHARTHAL
2679	TRIPURA	SOUTH TRIPURA	AMPINAGAR
2680	TRIPURA	SOUTH TRIPURA	BAGAFA
2680	TRIPURA	SOUTH TRIPURA	HRISHYAMUKH
2682	TRIPURA	SOUTH TRIPURA	KAKRABAN
2683	TRIPURA	SOUTH TRIPURA	MATABARI
2683	TRIPURA	SOUTH TRIPURA	RAJNAGAR
2685	TRIPURA	SOUTH TRIPURA	SATCHAND
2685	TRIPURA	WEST TRIPURA	BISHALGARH
2687	TRIPURA	WEST TRIPURA	
		WEST TRIPURA	BOXANAGAR
2688 2689	TRIPURA TRIPURA	WEST TRIPURA	DUKLI HEZAMARA
2690	TRIPURA	WEST TRIPURA	JAMPOIJALA
2690	TRIPURA	WEST TRIPURA	JIRANIA
2692	TRIPURA	WEST TRIPURA	KALYANPUR
2692	TRIPURA	WEST TRIPURA	KATHALIA
2693	TRIPURA	WEST TRIPURA	KHOWAI
2694	TRIPURA	WEST TRIPURA	MANDAI
2693	TRIPURA	WEST TRIPURA	MANDAI MELAGHAR
2697	TRIPURA	WEST TRIPURA	MOHANPUR
2698	TRIPURA	WEST TRIPURA	MUNGIAKAMI
2699	TRIPURA	WEST TRIPURA	PADMABIL
2700	TRIPURA	WEST TRIPURA	TELIAMURA
2700	TRIPURA	WEST TRIPURA	TULASHIKHAR
2701	UTTAR PRADESH	AGRA	ETMADPUR
2702	UTTAR PRADESH	AGRA	JAITPUR KALAN
2703	UTTAR PRADESH	AUKA	SORAM
2704	UTTAR PRADESH	AMBEDKAR NAGAR	BASKHARI
2703	UTTAR PRADESH	AURAIYA	ACHHALDA
2708	UTTAR PRADESH	AURAIYA	AJITMAL
2707	UTTAR PRADESH	AURAIYA	AURAIYA
2708	UTTAR PRADESH	AURAIYA	BHAGYANAGAR
2709	UTTAR PRADESH	AURAIYA	BIDHUNA
2710	UTTAR PRADESH	AURAIYA	ERVA KATRA
2711 2712	UTTAR PRADESH	AURAIYA	SAHAR
2712	UTTAR PRADESH	AJAMGARH	AZAMGRAH
2713	UTTAR PRADESH	AZAMGARH	MIRZAPUR
	ducation rfa/06.07.2012		110

2715	UTTAR PRADESH	BAGHPAT	BAGPAT
2715	UTTAR PRADESH	BAGHPAT	BAROUT
2710	UTTAR PRADESH	BAGHPAT	BINOLY
2717	UTTAR PRADESH	BAGHPAT	KHEKRA
2718	UTTAR PRADESH	BAGHPAT	PILANA
2719			
2720	UTTAR PRADESH UTTAR PRADESH	BALLIA BASTI	HANUMANGANJ
			DUBAULIA
2722	UTTAR PRADESH	BIJNOR	AAKU(NEHTOR)
2723	UTTAR PRADESH	BIJNOR	AFZALGRAH
2724	UTTAR PRADESH	BIJNOR	ALLHEPUR(DHAMPUR)
2725	UTTAR PRADESH	BULANDSHAHR	AGAUTA
2726	UTTAR PRADESH	BULANDSHAHR	B.B.NAGAR
2727	UTTAR PRADESH	CHANDAULI	CHAHANIYA
2728	UTTAR PRADESH	CHITRAKOOT	CHITRAKOOT
2729	UTTAR PRADESH	CHITRAKOOT	RAMNAGAR
2730	UTTAR PRADESH	DEORIA	BHAGALPUR
2731	UTTAR PRADESH	DEORIA	SALIMPUR
2732	UTTAR PRADESH	DEORIA	TARAKULAWA
2733	UTTAR PRADESH	ETAWAH	BARHPURA
2734	UTTAR PRADESH	ETAWAH	BASREHAR
2735	UTTAR PRADESH	ETAWAH	BHARTHANA
2736	UTTAR PRADESH	ETAWAH	CHAKARNAGAR
2737	UTTAR PRADESH	ETAWAH	JASWANT NAGAR
2738	UTTAR PRADESH	ETAWAH	MAHEWA
2739	UTTAR PRADESH	ETAWAH	SAIFAIE
2740	UTTAR PRADESH	ЕТАWAH	ТАКНА
2741	UTTAR PRADESH	FAIZABAD	MASAUDHA
2742	UTTAR PRADESH	FARRUKHABAD	BARH PUR
2743	UTTAR PRADESH	FARRUKHABAD	MOHAMDABAD
2744	UTTAR PRADESH	FATEHPUR	AMAULI
2745	UTTAR PRADESH	FATEHPUR	DEOMAI
2746	UTTAR PRADESH	FATEHPUR	KHAJUHA
2747	UTTAR PRADESH	FIROZABAD	AARANW
2748	UTTAR PRADESH	FIROZABAD	FIROZABAD
2749	UTTAR PRADESH	FIROZABAD	JASRANA
2750	UTTAR PRADESH	FIROZABAD	KHAIRGARH
2751	UTTAR PRADESH	FIROZABAD	MADANPUR
2751	UTTAR PRADESH	FIROZABAD	NARKHI
2752	UTTAR PRADESH	FIROZABAD	SHIKOHABAD
2753	UTTAR PRADESH	FIROZABAD	TUNDLA
2755	UTTAR PRADESH	GAUTAM BUDDHA NAGAR	DADRI
2756	UTTAR PRADESH	GAUTAM BUDDHA NAGAR	VISRAKH
2757	UTTAR PRADESH	GHAZIABAD	BHOJPUR
2758	UTTAR PRADESH	GHAZIABAD	HAPUR
2759	UTTAR PRADESH	GHAZIABAD	LONI
2759	UTTAR PRADESH	GHAZIABAD	MURAD NAGAR
2761	UTTAR PRADESH	GHAZIABAD	SIMBHAWALI
2761	UTTAR PRADESH	GHAZIABAD GHAZIPUR	BHADAURA
-	UTTAR PRADESH		KARANDA
2763 2764	UTTAR PRADESH	GHAZIPUR HATHRAS	HATHRAS
2765	UTTAR PRADESH	HATHRAS	SASNI
2766	UTTAR PRADESH	JALAUN	JALAUN

2767	UTTAR PRADESH	JALAUN	KONCH
2768	UTTAR PRADESH	JALAUN	KUTHAUND
2769	UTTAR PRADESH	JALAUN	MADHAUGADH
2770	UTTAR PRADESH	JAUNPUR	SIKRARA
2771	UTTAR PRADESH	JAUNPUR	SIRKONI
2772	UTTAR PRADESH	KANNAUJ	CHIBRAMAU
2773	UTTAR PRADESH	KANNAUJ	HASERAN
2774	UTTAR PRADESH	KANNAUJ	SAURIKHA
2775	UTTAR PRADESH	KANNAUJ	UMARDA
2776	UTTAR PRADESH	KANPUR DEHAT	AKBARPUR
2777	UTTAR PRADESH	KANPUR DEHAT	AMARAUDHA
2778	UTTAR PRADESH	KANPUR DEHAT	DERAPUR
2778	UTTAR PRADESH	KANPUR DEHAT	JHIJHAK
2779	UTTAR PRADESH	KANPUR DEHAT	KAKWAN
2780	UTTAR PRADESH	KANPUR DEHAT	MAITHA
2781			MALASA
	UTTAR PRADESH	KANPUR DEHAT	
2783	UTTAR PRADESH	KANPUR DEHAT	RAJPUR
2784	UTTAR PRADESH	KANPUR DEHAT	RASULABAD
2785	UTTAR PRADESH	KANPUR DEHAT	SANDALPUR
2786	UTTAR PRADESH	KANPUR DEHAT	SARVANKHERA
2787	UTTAR PRADESH	KANPUR NAGAR	BHITERGAON
2788	UTTAR PRADESH	KANPUR NAGAR	BIDHNU
2789	UTTAR PRADESH	KANPUR NAGAR	BILHAUR
2790	UTTAR PRADESH	KANPUR NAGAR	CHAUBEPUR
2791	UTTAR PRADESH	KANPUR NAGAR	GHATAMPUR
2792	UTTAR PRADESH	KANPUR NAGAR	KAKWAN
2793	UTTAR PRADESH	KANPUR NAGAR	KALYANPUR
2794	UTTAR PRADESH	KANPUR NAGAR	PATARA
2795	UTTAR PRADESH	KANPUR NAGAR	SARSAUL
2796	UTTAR PRADESH	KANPUR NAGAR	SHIVRAJPUR
2797	UTTAR PRADESH	KHERI	GOLA
2798	UTTAR PRADESH	KHERI	LAKHIMPUR
2799	UTTAR PRADESH	KHERI	MOM
2800	UTTAR PRADESH	KHERI	PAL
2801	UTTAR PRADESH	LALITPUR	LALITPUR
2802	UTTAR PRADESH	MAHARAJGANJ	ANAND NAGAR
2803	UTTAR PRADESH	MAINPURI	BARNAHAL
2804	UTTAR PRADESH	MAINPURI	BEWAR
2805	UTTAR PRADESH	MAINPURI	JAGIR
2806	UTTAR PRADESH	MAINPURI	KISHANI
2807	UTTAR PRADESH	MAINPURI	KURAWALI
2808	UTTAR PRADESH	MAINPURI	MAINPURI
2809	UTTAR PRADESH	MAINPURI	SULTANGANJ
2810	UTTAR PRADESH	MAU	DOHRIGHAT
2811	UTTAR PRADESH	MAU	FATHEPUR MANDAON
2812	UTTAR PRADESH	MAU	GHOSI
2813	UTTAR PRADESH	MAU	PARDAHA
2814	UTTAR PRADESH	MEERUT	DAURALA
2815	UTTAR PRADESH	MEERUT	HASTINAPUR
2816	UTTAR PRADESH	MEERUT	JANI
2817	UTTAR PRADESH	MEERUT	MACHRA
2818	UTTAR PRADESH	MEERUT	MAWANA
2819	UTTAR PRADESH	MEERUT	MEERUT
	ducation rfa/06.07.2012	-	112

2820	UTTAR PRADESH	MEERUT	RAJPURA
2821	UTTAR PRADESH	MEERUT	ROHTA
2822	UTTAR PRADESH	MEERUT	SARDHANA
2823	UTTAR PRADESH	MIRZAPUR	NERAYEN PUR
2824	UTTAR PRADESH	MIRZAPUR	SEEKHAR
2825	UTTAR PRADESH	MORADABAD	KUNDERKI
2826	UTTAR PRADESH	MUZAFFARNAGAR	BAGHRA
2827	UTTAR PRADESH	MUZAFFARNAGAR	KHATAULI
2828	UTTAR PRADESH	MUZAFFARNAGAR	SADAR
2829	UTTAR PRADESH	MUZAFFARNAGAR	SHAMLI
2830	UTTAR PRADESH	MUZAFFARNAGAR	THANABHAWAN
2830	UTTAR PRADESH	PRATAPGARH	BABA BHELKHARNATH
2832	UTTAR PRADESH	PRATAPGARH	SADAR
2832	UTTAR PRADESH	RAE BARELI	LALGANJ
2833	UTTAR PRADESH	RAE BARELI RAE BARELI	SARENI
2834	UTTAR PRADESH	SAHARANPUR	BALIA KHEDI
		SAHARANPUR	DEOBAND
2836	UTTAR PRADESH		
2837	UTTAR PRADESH	SAHARANPUR	NAGAL
2838	UTTAR PRADESH	SAHARANPUR	NAKUR
2839	UTTAR PRADESH	SAHARANPUR	NANAUTA
2840	UTTAR PRADESH	SAHARANPUR	RAMPUR MANIHARAN
2841	UTTAR PRADESH	SANT KABIR NAGAR	BELHAR KALA
2842	UTTAR PRADESH	SANT KABIR NAGAR	PAULI
2843	UTTAR PRADESH	SHAHJAHANPUR	MADNAPUR
2844	UTTAR PRADESH	SIDDHARTHNAGAR	SATHA
2845	UTTAR PRADESH	SULTANPUR	MOTIGARPUR
2846	UTTAR PRADESH	UNNAO	BIGHAPUR
2847	UTTAR PRADESH	UNNAO	SIKANDERPUR KARN
2848	UTTAR PRADESH	UNNAO	SUMERPUR
2849	UTTAR PRADESH	VARANASI	BARAGAON
2850	UTTAR PRADESH	VARANASI	HARHUA
2851	UTTAR PRADESH	VARANASI	KASHIVIDYAPEETH
2852	UTTARAKHAND	ALMORA	BHAISIYACHHANA
2853	UTTARAKHAND	ALMORA	BHIKIYASAIN
2854	UTTARAKHAND	ALMORA	CHAUKHUTIYA
2855	UTTARAKHAND	ALMORA	DWARAHAT
2856	UTTARAKHAND	ALMORA	HAWALBAG
2857	UTTARAKHAND	ALMORA	LAMGARAH
2858	UTTARAKHAND	ALMORA	SULT
2859	UTTARAKHAND	ALMORA	SYALDEY
2860	UTTARAKHAND	ALMORA	TAKULA
2861	UTTARAKHAND	ALMORA	TARIKHET
2862	UTTARAKHAND	BAGESHWAR	BAGESHWAR
2863	UTTARAKHAND	BAGESHWAR	GARUR
2864	UTTARAKHAND	BAGESHWAR	КАРКОТЕ
2865	UTTARAKHAND	CHAMOLI	DASHOLI
2866	UTTARAKHAND	CHAMOLI	DEWAL
2867	UTTARAKHAND	CHAMOLI	GAIRSAIN
2868	UTTARAKHAND	CHAMOLI	GHAT
2869	UTTARAKHAND	CHAMOLI	JOSHIMATH
2870	UTTARAKHAND	CHAMOLI	KARANPRAYAG
2871	UTTARAKHAND	CHAMOLI	NARAYANBAGAR
2872	UTTARAKHAND	CHAMOLI	POKHARI
	ducation rfa/06.07.2012		113

2873	UTTARAKHAND	CHAMOLI	THARALI
2874	UTTARAKHAND	CHAMPAWAT	BARAKOT
2875	UTTARAKHAND	CHAMPAWAT	СНАМРАЖАТ
2876	UTTARAKHAND	CHAMPAWAT	LOHAGHAT
2877	UTTARAKHAND	СНАМРАЖАТ	PATI
2878	UTTARAKHAND	DEHRADUN	DOIWALA
2879	UTTARAKHAND	DEHRADUN	DOIWALA RISHIKESH
2880	UTTARAKHAND	DEHRADUN	RAIPUR
2881	UTTARAKHAND	DEHRADUN	SAHASPUR
2882	UTTARAKHAND	DEHRADUN	VIKASNAGAR
2883	UTTARAKHAND	NAINITAL	BETAALGHAT
2884	UTTARAKHAND	NAINITAL	BHEEMTAL
2885	UTTARAKHAND	NAINITAL	DHARI
2885	UTTARAKHAND	NAINITAL	HALDWANI
2880	UTTARAKHAND	NAINITAL	KOTABAGH
	UTTARAKHAND		
2888		NAINITAL	OKHALKANDA
2889	UTTARAKHAND	NAINITAL	RAMGARH
2890	UTTARAKHAND	NAINITAL	RAMNAGAR
2891	UTTARAKHAND	PAURI GARHWAL	BIRONKHAL
2892	UTTARAKHAND	PAURI GARHWAL	DUGADDA
2893	UTTARAKHAND	PAURI GARHWAL	DWARIKAHL
2894	UTTARAKHAND	PAURI GARHWAL	EKESHWAR
2895	UTTARAKHAND	PAURI GARHWAL	JAYHARIKHAL
2896	UTTARAKHAND	PAURI GARHWAL	KALJIKHAL
2897	UTTARAKHAND	PAURI GARHWAL	KHIRSU
2898	UTTARAKHAND	PAURI GARHWAL	КОТ
2899	UTTARAKHAND	PAURI GARHWAL	NAINIDANDA
2900	UTTARAKHAND	PAURI GARHWAL	PABAU
2901	UTTARAKHAND	PAURI GARHWAL	PAURI
2902	UTTARAKHAND	PAURI GARHWAL	POKHRA
2903	UTTARAKHAND	PAURI GARHWAL	RIKHANIKHAL
2904	UTTARAKHAND	PAURI GARHWAL	THALISAIN
2905	UTTARAKHAND	PAURI GARHWAL	YAMKESHWAR
2906	UTTARAKHAND	PITHORAGARH	BERINAG
2907	UTTARAKHAND	PITHORAGARH	BIN
2908	UTTARAKHAND	PITHORAGARH	DHARCHULA
2909	UTTARAKHAND	PITHORAGARH	DIDIHAT
2910	UTTARAKHAND	PITHORAGARH	GANGOLIHAT
2911	UTTARAKHAND	PITHORAGARH	KANALICHHEENA
2912	UTTARAKHAND	PITHORAGARH	MOONAKOT
2913	UTTARAKHAND	PITHORAGARH	MUNSYARI
2914	UTTARAKHAND	RUDRAPRAYAG	AUGUSTYAMUNI
2915	UTTARAKHAND	RUDRAPRAYAG	JAKHOLI
2916	UTTARAKHAND	RUDRAPRAYAG	UKHIMATH
2917	UTTARAKHAND	TEHRI GARHWAL	СНАМВА
2918	UTTARAKHAND	TEHRI GARHWAL	DEVPRAYAG
2919	UTTARAKHAND	TEHRI GARHWAL	JAKHNIDHAR
2920	UTTARAKHAND	TEHRI GARHWAL	KEERTINAGAR
2921	UTTARAKHAND	UDHAM SINGH NAGAR	GADARPUR
2922	UTTARAKHAND	UDHAM SINGH NAGAR	JASPUR
2923	UTTARAKHAND	UDHAM SINGH NAGAR	KASHIPUR
2923	UTTARAKHAND	UDHAM SINGH NAGAR	KHATIMA
2925	UTTARAKHAND	UDHAM SINGH NAGAR	RUDRAPUR
	ducation rfa/06.07.2012		114

2926	UTTARAKHAND	UDHAM SINGH NAGAR	SITARGANJ
2927	UTTARAKHAND	UTTARKASHI	BHATWARI
2928	UTTARAKHAND	UTTARKASHI	DUNDA
2929	WEST BENGAL	BANKURA	BANKURA-I
2930	WEST BENGAL	BANKURA	BANKURA-II
2931	WEST BENGAL	BANKURA	BARJORA
2932	WEST BENGAL	BANKURA	INDAS
2933	WEST BENGAL	BANKURA	JOYPUR
2934	WEST BENGAL	BANKURA	KHATRA-I
2935	WEST BENGAL	BANKURA	KOTULPUR
2936	WEST BENGAL	BANKURA	RAIPUR-I
2937	WEST BENGAL	BANKURA	SARENGA [RAIPUR-II]
2938	WEST BENGAL	BANKURA	SIMLAPAL
2939	WEST BENGAL	BANKURA	TALDANGRA
2940	WEST BENGAL	BARDDHAMAN	ANDAL
2941	WEST BENGAL	BARDDHAMAN	AUSGRAM-I
2942	WEST BENGAL	BARDDHAMAN	AUSGRAM-II
2943	WEST BENGAL	BARDDHAMAN	BARABANI
2944	WEST BENGAL	BARDDHAMAN	BHATAR
2945	WEST BENGAL	BARDDHAMAN	BURDWAN-I
2946	WEST BENGAL	BARDDHAMAN	BURDWAN-II
2947	WEST BENGAL	BARDDHAMAN	DURGAPUR-FARIDPUR
2948	WEST BENGAL	BARDDHAMAN	GALSI-I
2948	WEST BENGAL	BARDDHAMAN	GALSI-II GALSI-II
2949	WEST BENGAL	BARDDHAMAN	JAMALPUR
2950	WEST BENGAL	BARDDHAMAN	JAMURIA
2931	WEST BENGAL	BARDDHAMAN	KALNA-I
2932	WEST BENGAL	BARDDHAMAN	KALNA-I KALNA-II
2933	WEST BENGAL	BARDDHAMAN	KANKSA
2934	WEST BENGAL	BARDDHAMAN	KATWA-I
2956	WEST BENGAL	BARDDHAMAN	KATWA-II
2957	WEST BENGAL WEST BENGAL	BARDDHAMAN	KETUGRAM-I
2958		BARDDHAMAN	KETUGRAM-II
2959	WEST BENGAL	BARDDHAMAN	KHANDOGHOSH
2960	WEST BENGAL	BARDDHAMAN	MEMARI-I
2961	WEST BENGAL	BARDDHAMAN	MEMARI-II
2962	WEST BENGAL	BARDDHAMAN	MONGALKOTE
2963	WEST BENGAL	BARDDHAMAN	MONTESWAR
2964	WEST BENGAL	BARDDHAMAN	PANDABESWAR
2965	WEST BENGAL	BARDDHAMAN	PURBASTHALI-I
2966	WEST BENGAL	BARDDHAMAN	PURBASTHALI-II
2967	WEST BENGAL	BARDDHAMAN	RAINA-I
2968	WEST BENGAL	BARDDHAMAN	RAINA-II RANKGANK
2969	WEST BENGAL	BARDDHAMAN	RANIGANJ
2970	WEST BENGAL	BARDDHAMAN	SALANPUR
2971	WEST BENGAL	BIRBHUM	BOLPUR SRINIKETAN
2972	WEST BENGAL	BIRBHUM	ILLAMBAZAR
2973	WEST BENGAL	BIRBHUM	KHOYRASOLE
2974	WEST BENGAL	BIRBHUM	LABPUR
2975	WEST BENGAL	BIRBHUM	MAYURESWAR-I
2976	WEST BENGAL	BIRBHUM	MAYURESWAR-II
2977	WEST BENGAL	BIRBHUM	NALHATI-I
2978	WEST BENGAL	BIRBHUM	NALHATI-II

2979	WEST BENGAL	BIRBHUM	NANOOR
2980	WEST BENGAL	BIRBHUM	RAMPURHAT-I
2981	WEST BENGAL	BIRBHUM	RAMPURHAT-II
2982	WEST BENGAL	BIRBHUM	SAINTHIA
2983	WEST BENGAL	BIRBHUM	SURI-I
2984	WEST BENGAL	BIRBHUM	SURI-II
2985	WEST BENGAL	DAKSHIN DINAJPUR	BALURGHAT
2986	WEST BENGAL	DAKSHIN DINAJPUR	BANSHIHARI
2987	WEST BENGAL	DAKSHIN DINAJPUR	GANGARAMPUR
2988	WEST BENGAL	DAKSHIN DINAJPUR	HILI
2989	WEST BENGAL	DAKSHIN DINAJPUR	KUMARGANJ
2990	WEST BENGAL	DAKSHIN DINAJPUR	TAPAN
2990	WEST BENGAL	DARJILING	ALGARAH - MUNSONG
2991	WEST BENGAL	DARJILING	BONG - DUNGRA
2992	WEST BENGAL	DARJILING	CHONGTONG - RISHIHAT
2993	WEST BENGAL	DARJILING	GHOOM - JOREBUNGLOW
			GIDDAYPAHAR - PANKHABARI
2995	WEST BENGAL	DARJILING	
2996	WEST BENGAL	DARJILING	GITDABLING - SINJI
2997	WEST BENGAL	DARJILING	GORUBATHAN
2998	WEST BENGAL	DARJILING	KALIMPONG KHAS
2999	WEST BENGAL	DARJILING	KALIMPONG TOWN - DR
			GRAHAM'S HOMES
3000	WEST BENGAL	DARJILING	KURSEONG TOWN
3001	WEST BENGAL	DARJILING	LAVA - PEDONG
3002	WEST BENGAL	DARJILING	LEBONG VALLEY
3003	WEST BENGAL	DARJILING	LOPCHU - PESHOK
3004	WEST BENGAL	DARJILING	MIRIK
3005	WEST BENGAL	DARJILING	MUNGPOO - LATPANCHAR
3006	WEST BENGAL	DARJILING	POKHRIABONG - NAGRI
3007	WEST BENGAL	DARJILING	PULBAZAR - BIJANBARI
3008	WEST BENGAL	DARJILING	RIMBIK - LODHOMA
3009	WEST BENGAL	DARJILING	RUNGBULL - DOOTERIA
3010	WEST BENGAL	DARJILING	SADAR I
3011	WEST BENGAL	DARJILING	SADAR II
3012	WEST BENGAL	DARJILING	SINGAMARI - TUKVAR
3013	WEST BENGAL	DARJILING	SONADA - TUNG
3014	WEST BENGAL	DARJILING	SOURENI - PANIGHATTA
3015	WEST BENGAL	DARJILING	SUKHIAPOKHRI - MANEYBHANJANG
3016	WEST BENGAL	DARJILING	TAKDAH - TEESTA VALLEY
3017	WEST BENGAL	DARJILING	TINDHARIA - SUKNA
3018	WEST BENGAL	DARJILING	TODEY - TANGTA - JALDHAKA
3019	WEST BENGAL	HAORA	AMTA-I
3020	WEST BENGAL	HAORA	AMTA-II
3021	WEST BENGAL	HAORA	BAGNAN-I
3022	WEST BENGAL	HAORA	BAGNAN-II
3023	WEST BENGAL	HAORA	BALLY- JAGACHA
3024	WEST BENGAL	HAORA	DOMJUR
3025	WEST BENGAL	HAORA	JAGATBALLAVPUR
3026	WEST BENGAL	HAORA	PANCHLA
3027	WEST BENGAL	HAORA	SANKRAIL
3027	WEST BENGAL	HAORA	SHYAMPUR-I
3028	WEST BENGAL	HAORA	SHYAMPUR-II
	ducation rfa/06.07.2012		116

3030	WEST BENGAL	HAORA	UDAYNARAYANPUR
3031	WEST BENGAL	HAORA	ULUBERIA-I
3032	WEST BENGAL	HAORA	ULUBERIA-II
3033	WEST BENGAL	HUGLI	ARAMBAGH
3034	WEST BENGAL	HUGLI	BALAGARH
3035	WEST BENGAL	HUGLI	CHANDITALA-I
3036	WEST BENGAL	HUGLI	CHANDITALA-II
3037	WEST BENGAL	HUGLI	CHINSURAH-MOGRA
3038	WEST BENGAL	HUGLI	DHANIAKHALI
3039	WEST BENGAL	HUGLI	GOGHAT-I
3040	WEST BENGAL	HUGLI	GOGHAT-II
3041	WEST BENGAL	HUGLI	HARIPAL
3042	WEST BENGAL	HUGLI	JANGIPARA
3043	WEST BENGAL	HUGLI	KHANAKUL-I
3044	WEST BENGAL	HUGLI	KHANAKUL-II
3045	WEST BENGAL	HUGLI	PANDUA
3046	WEST BENGAL	HUGLI	POLBA-DADPUR
3047	WEST BENGAL	HUGLI	PURSURAH
3048	WEST BENGAL	HUGLI	SERAMPUR-UTTARPARA
3049	WEST BENGAL	HUGLI	SINGUR
3050	WEST BENGAL	HUGLI	TARAKASHWAR
3051	WEST BENGAL	HUGLI	UTTARPARA KOTRANG
3052	WEST BENGAL	JALPAIGURI	ALIPURDUAR-I
3052	WEST BENGAL	JALPAIGURI	ALIPURDUAR-II
3054	WEST BENGAL	JALPAIGURI	FALAKATA
3055	WEST BENGAL WEST BENGAL	JALPAIGURI	JALANATA JALPAIGURI SADAR
3055	WEST BENGAL WEST BENGAL	JALPAIGURI	KUMARGRAMDUAR
3057	WEST BENGAL WEST BENGAL	JALPAIGURI	MAYNAGURI
3058	WEST BENGAL WEST BENGAL	JALPAIGURI	RAJGANJ
3058	WEST BENGAL WEST BENGAL	KOCH BIHAR	COOCH BEHAR-I
3060	WEST BENGAL	KOCH BIHAR	COOCH BEHAR-II
3061	WEST BENGAL	KOCH BIHAR	DINHATA-I
3062	WEST BENGAL	KOCH BIHAR	DINHATA-II
3063	WEST BENGAL	KOCH BIHAR	HALDIBARI
3064	WEST BENGAL	KOCH BIHAR	MATHABHANGA-I
3065	WEST BENGAL	KOCH BIHAR	MATHABHANGA-II
3066	WEST BENGAL	KOCH BIHAR	MEKHLIGANJ
3067	WEST BENGAL	KOCH BIHAR	TUFANGANJ-I
3068	WEST BENGAL	KOCH BIHAR	TUFANGANJ-II
3069	WEST BENGAL	MALDAH	CHANCHAL-1
3070	WEST BENGAL	MALDAH	KALIACHAK-1
3071	WEST BENGAL	MURSHIDABAD	BARWAN
3072	WEST BENGAL	MURSHIDABAD	BELDANGA-I
3073	WEST BENGAL	MURSHIDABAD	BELDANGA-II
3074	WEST BENGAL	MURSHIDABAD	BERHAMPORE
3075	WEST BENGAL	MURSHIDABAD	BHAGAWANGOLA-I
3076	WEST BENGAL	MURSHIDABAD	DOMKAL
3077	WEST BENGAL	MURSHIDABAD	HARIHARPARA
3078	WEST BENGAL	MURSHIDABAD	JALANGI
3079	WEST BENGAL	MURSHIDABAD	LALGOLA
3080	WEST BENGAL	MURSHIDABAD	MURSHIDABAD-JIAGANJ (M.J)
3081	WEST BENGAL	MURSHIDABAD	NABAGRAM
3082	WEST BENGAL	MURSHIDABAD	NOWDA

3083	WEST BENGAL	MURSHIDABAD	RANINAGAR-I
3084	WEST BENGAL	MURSHIDABAD	RANINAGAR-II
3085	WEST BENGAL	MURSHIDABAD	SAGARDIGHI
3086	WEST BENGAL	NADIA	СНАКДАН
3087	WEST BENGAL	NADIA	CHAPRA
3088	WEST BENGAL	NADIA	HANSKHALI
3089	WEST BENGAL	NADIA	HARINGHATA
3090	WEST BENGAL	NADIA	KALIGANJ
3091	WEST BENGAL	NADIA	KARIMPUR - I
3092	WEST BENGAL	NADIA	KARIMPUR - II
3092	WEST BENGAL	NADIA	KRISHNAGANJ
3094	WEST BENGAL	NADIA	KRISHNAGAR - I
3095	WEST BENGAL	NADIA	KRISHNAGAR - II
3096	WEST BENGAL	NADIA	NABADWIP
3097	WEST BENGAL	NADIA	RANAGHAT - I
3098	WEST BENGAL	NADIA	RANAGHAT - II
3099	WEST BENGAL	NADIA	SANTIPUR
3100	WEST BENGAL	NADIA	ТЕНАТТА - І
3100	WEST BENGAL	NADIA	TEHATTA - II
5101	WEST DENOAL	NORTH TWENTY FOUR	
3102	WEST BENGAL	PARGANA	AMDANGA
3103	WEST BENGAL	NORTH TWENTY FOUR PARGANA	BADURIA
3104	WEST BENGAL	NORTH TWENTY FOUR PARGANA	BAGDAH
3105	WEST BENGAL	NORTH TWENTY FOUR PARGANA	BARASAT - I
3106	WEST BENGAL	NORTH TWENTY FOUR PARGANA	BARASAT - II
3107	WEST BENGAL	NORTH TWENTY FOUR PARGANA	BARRACKPORE - I
3108	WEST BENGAL	NORTH TWENTY FOUR PARGANA	BARRACKPORE - II
3109	WEST BENGAL	NORTH TWENTY FOUR PARGANA	BASIRHAT - I
3110	WEST BENGAL	NORTH TWENTY FOUR PARGANA	BASIRHAT - II
3111	WEST BENGAL	NORTH TWENTY FOUR PARGANA	BONGAON
3112	WEST BENGAL	NORTH TWENTY FOUR PARGANA	DEGANGA
3113	WEST BENGAL	NORTH TWENTY FOUR PARGANA	GAIGHATA
3114	WEST BENGAL	NORTH TWENTY FOUR PARGANA	HABRA - I
3115	WEST BENGAL	NORTH TWENTY FOUR PARGANA	HABRA - II
3116	WEST BENGAL	NORTH TWENTY FOUR PARGANA	HAROA
3117	WEST BENGAL	NORTH TWENTY FOUR PARGANA	HASNABAD
3118	WEST BENGAL	NORTH TWENTY FOUR PARGANA	HINGALGUNJ

3119	WEST BENGAL	NORTH TWENTY FOUR PARGANA	MINAKHAN
3120	WEST BENGAL	NORTH TWENTY FOUR PARGANA	RAJARHAT
3121	WEST BENGAL	NORTH TWENTY FOUR PARGANA	SANDESHKHALI - I
3122	WEST BENGAL	NORTH TWENTY FOUR PARGANA	SANDESHKHALI - II
3123	WEST BENGAL	NORTH TWENTY FOUR PARGANA	SWARUPNAGAR
3124	WEST BENGAL	PASCHIM MEDINIPUR	BINPUR-1
3125	WEST BENGAL	PASCHIM MEDINIPUR	BINPUR-2
3126	WEST BENGAL	PASCHIM MEDINIPUR	CHANDRAKONA-1
3127	WEST BENGAL	PASCHIM MEDINIPUR	CHANDRAKONA-2
3128	WEST BENGAL	PASCHIM MEDINIPUR	DANTAN-1
3129	WEST BENGAL	PASCHIM MEDINIPUR	DANTAN-2
3130	WEST BENGAL	PASCHIM MEDINIPUR	DASPUR-1
3131	WEST BENGAL	PASCHIM MEDINIPUR	DASPUR-2
3132	WEST BENGAL	PASCHIM MEDINIPUR	DEBRA
3133	WEST BENGAL	PASCHIM MEDINIPUR	GARBETA-1
3134	WEST BENGAL	PASCHIM MEDINIPUR	GARBETA-2
3135	WEST BENGAL	PASCHIM MEDINIPUR	GARBETA-3
3136	WEST BENGAL	PASCHIM MEDINIPUR	GHATAL
3137	WEST BENGAL	PASCHIM MEDINIPUR	GOPIBALLAVPUR-2
3138	WEST BENGAL	PASCHIM MEDINIPUR	JAMBONI
3139	WEST BENGAL	PASCHIM MEDINIPUR	JHARGRAM
3140	WEST BENGAL	PASCHIM MEDINIPUR	KESHIARY
3141	WEST BENGAL	PASCHIM MEDINIPUR	KESHPUR
3142	WEST BENGAL	PASCHIM MEDINIPUR	KGP-1
3142	WEST BENGAL	PASCHIM MEDINIPUR	KGP-2
3143	WEST BENGAL	PASCHIM MEDINIPUR	MIDNAPORE SADAR
3144	WEST BENGAL	PASCHIM MEDINIFUR	MOHANPUR
3145	WEST BENGAL	PASCHIM MEDINIFUR	NARAYANGARH
3140	WEST BENGAL	PASCHIM MEDINIFUR	PINGLA
3147	WEST BENGAL	PASCHIM MEDINIFUR	SABONG
3148	WEST BENGAL	PASCHIM MEDINIPUR	SALBONI
3149	WEST BENGAL	PASCHIM MEDINIPUR	SANKRAIL
3150	WEST BENGAL	PURBA MEDINIPUR	BHAGWANPUR -I
3152	WEST BENGAL	PURBA MEDINIPUR	BHAGWANPUR -II
3152	WEST BENGAL	PURBA MEDINIPUR	CONTAI -I
3153	WEST BENGAL	PURBA MEDINIPUR	CONTAI -II
3154	WEST BENGAL WEST BENGAL	PURBA MEDINIPUR	CONTAI -III
3155	WEST BENGAL WEST BENGAL	PURBA MEDINIPUR	EGRA -I
3156			EGRA -I EGRA -II
	WEST BENGAL	PURBA MEDINIPUR PURBA MEDINIPUR	
3158	WEST BENGAL		HALDIA KHEILIDI I
3159	WEST BENGAL	PURBA MEDINIPUR	KHEJURI -I
3160 3161	WEST BENGAL	PURBA MEDINIPUR	KHEJURI -II MAHISADAL
	WEST BENGAL	PURBA MEDINIPUR PURBA MEDINIPUR	MAHISADAL MOYNA
3162	WEST BENGAL		
3163	WEST BENGAL	PURBA MEDINIPUR	NANDAKUMAR
3164	WEST BENGAL	PURBA MEDINIPUR	NANDIGRAM -I
3165	WEST BENGAL	PURBA MEDINIPUR	NANDIGRAM -II
3166	WEST BENGAL	PURBA MEDINIPUR	NANDIGRAM -III

3167	WEST BENGAL	PURBA MEDINIPUR	PANSKURA -I
3168	WEST BENGAL	PURBA MEDINIPUR	PANSKURA -II
3169	WEST BENGAL	PURBA MEDINIPUR	PATASHPUR -I
3170	WEST BENGAL	PURBA MEDINIPUR	PATASHPUR -II
3171	WEST BENGAL	PURBA MEDINIPUR	RAMNAGAR -I
3172	WEST BENGAL	PURBA MEDINIPUR	RAMNAGAR -II
3173	WEST BENGAL	PURBA MEDINIPUR	SAHID MATANGANI
3174	WEST BENGAL	PURBA MEDINIPUR	SUTAHATA
3175	WEST BENGAL	PURBA MEDINIPUR	TAMLUK
3176	WEST BENGAL	SILIGURI	MATIGARA
3177	WEST BENGAL	SILIGURI	NAXALBARI
3178	WEST BENGAL	SOUTH TWENTY FOUR PARGAN	BARUIPUR
3179	WEST BENGAL	SOUTH TWENTY FOUR PARGAN	BHANGORE - 1
3180	WEST BENGAL	SOUTH TWENTY FOUR PARGAN	BHANGORE - 2
3181	WEST BENGAL	SOUTH TWENTY FOUR PARGAN	BISHNUPUR - 1
3182	WEST BENGAL	SOUTH TWENTY FOUR PARGAN	BISHNUPUR - 2
3183	WEST BENGAL	SOUTH TWENTY FOUR PARGAN	BUDGE BUDGE - 1
3184	WEST BENGAL	SOUTH TWENTY FOUR PARGAN	BUDGE BUDGE - 2
3185	WEST BENGAL	SOUTH TWENTY FOUR PARGAN	CANNING - 1
3186	WEST BENGAL	SOUTH TWENTY FOUR PARGAN	DIAMOND HARBOUR - 1
3187	WEST BENGAL	SOUTH TWENTY FOUR PARGAN	DIAMOND HARBOUR - 2
3188	WEST BENGAL	SOUTH TWENTY FOUR PARGAN	FALTA
3189	WEST BENGAL	SOUTH TWENTY FOUR PARGAN	GOSABA
3190	WEST BENGAL	SOUTH TWENTY FOUR PARGAN	JOYNAGAR - 1
3191	WEST BENGAL	SOUTH TWENTY FOUR PARGAN	KAKDWIP
3192	WEST BENGAL	SOUTH TWENTY FOUR PARGAN	KULPI
3193	WEST BENGAL	SOUTH TWENTY FOUR PARGAN	MAGRAHAT - 1
3194	WEST BENGAL	SOUTH TWENTY FOUR PARGAN	MAGRAHAT - 2
3195	WEST BENGAL	SOUTH TWENTY FOUR PARGAN	MANDIRBAZAR
3196	WEST BENGAL	SOUTH TWENTY FOUR PARGAN	MATHURAPUR - 1
3197	WEST BENGAL	SOUTH TWENTY FOUR PARGAN	MATHURAPUR - 2
3198	WEST BENGAL	SOUTH TWENTY FOUR PARGAN	NAMKHANA

3199	WEST BENGAL	SOUTH TWENTY FOUR PARGAN	PATHARPRATIMA
3200	WEST BENGAL	SOUTH TWENTY FOUR PARGAN	SAGAR
3201	WEST BENGAL	SOUTH TWENTY FOUR PARGAN	SONARPUR
3202	WEST BENGAL	SOUTH TWENTY FOUR PARGAN	THAKURPUKUR MAHESHTALA
3203	WEST BENGAL	UTTAR DINAJPUR	HEMTABAD